Mitigating Human-Elephant Conflict through an Engineering Approach

Chinthaka M. Dissanayake ORCID:0000-0001-5925-6156

Submitted in total fulfilment of the requirements of the degree of

Doctor of Philosophy

Melbourne School of Engineering The University of Melbourne

October 2018

Copyright © 2018 Chinthaka M. Dissanayake

All rights reserved. No part of the publication may be reproduced in any form by print, photoprint, microfilm or any other means without written permission from the author.

Abstract

ELEPHANTS and humans existed with a long tradition of interdependence up until contemporary times. However, today human-elephant conflict (HEC) has become one of the most significant environmental and socio-economic crises in some parts of the world. The conflict primarily is a consequence of frequent attacks from crop-raiding elephants to the rural agricultural communities. The primary objective of this research is to reduce the occurrences of human-elephant conflict by developing a solution in a way that rural agricultural communities gain the ability to identify elephant migration towards the village in advance. The proposed solution is basically to incorporate a sensor network to detect, localize and track elephants before they move into the village area.

In this thesis first, we studied the social structure and behavioural patterns of elephants and identified significant characteristics that can be utilized in the proposed solution. We have recognized the seasonal characteristics which indicate the probability of elephants migrating towards the villages and more importantly their behavioural pattern of utilizing the same travel routes. We have further investigated their communication and call patterns and identified that low-frequency elephant calls called rumbles can be utilized to detect the presence of elephants from a faraway distance.

Next, we developed a framework which was utilized to investigate the feasibility of using acoustic sensor network for elephant localization. Then we analysed the effects on the accuracy of elephant localization and detection based on the characteristics of sound propagation over the air. We investigated the effect of various meteorological parameters in elephant vocalization propagation medium and their consequential effects on elephant positioning accuracy. In addition, we evaluated the effect of signal-to-noise ratio (SNR) on the detection of vocalization signal which subsequently affects the localization accuracy. Extensive simulation results revealed that the temperature variations between sensors and deviation in wind velocity have a significant effect on positioning accuracy of the system. In addition, localization accuracy decreased enormously once the SNR level fell under certain marginal level.

We then introduced a novel technique with sound generating probes to estimate the average speed of sound which is affected by environmental parameters. The technique was then integrated with the elephant localization system framework which utilizes acoustic sensor network. Performance of the system was evaluated under real-world ecological scenarios against a sensor network which is equipped with wind and temperature sensors. Analysis revealed that the sensor network with the proposed probe technique achieved the best overall accuracy of localization, despite the limited information it gathers from the environment.

Finally, we built a hardware system to test and validate our approach. Initial experiments were designed and performed by replaying recorded elephant sounds under different environmental conditions. The results overall illustrated that the system can provide significant localization accuracy under a range of wind and temperature conditions. An identical hardware set up was then used to localize wild elephants in Sri Lanka. Our approach enabled localization of wild elephants at a distance of over 500 m from the sensors to within 30 m accuracy, providing adequate time for the villages to take appropriate safety measures.

Declaration

This is to certify that

- 1. the thesis comprises only my original work towards the PhD,
- 2. due acknowledgement has been made in the text to all other material used,
- 3. the thesis is less than 100,000 words in length, exclusive of tables, maps, bibliographies and appendices.

Chinthaka M. Dissanayake, October 2018

Acknowledgements

I would like to express my heartfelt thanks to my principal supervisor, Dr. Malka Nishanthi Halgamuge for providing me with the opportunity to pursue my higher studies at the University of Melbourne. I am indebted to her patience, encouragement and support from the beginning to the end of my PhD candidature. Prof. Rao Kotagiri deserves the utmost respect for mentoring me at every stage of the research project. Without his constant guidance and feedback, completion of the thesis may have never been possible. I also convey my sincere thanks to Prof. Bill Moran for his amazingly generous support throughout my PhD journey.

I am grateful to Prof. Andrew Western for chairing my advisory committee and for the insight given in the progress review meetings from time to time.

I am especially indebted to The University of Melbourne for supporting me financially through Melbourne International Research Scholarship (MIRS) and Melbourne International Fee Remission Scholarship (MIFRS) during my studies. I would also express my gratitude to The Institute for a Broadband-Enabled Society (IBES) for providing me with IBES top-up scholarship and the Melbourne Sustainable Society Institute (MSSI) at The University of Melbourne for additional financial support.

I would also like to thank all my friends and colleagues at The University of Melbourne during my candidature for their motivation and making my stay at the university enjoyable and memorable.

I would especially like to thank the supporting crew and the villagers who helped me during the data collection procedures at Sri Lanka.

I am extremely grateful to the love of my life Umanga for her patience and encouragement during the completion stage of my PhD and the understanding all the difficulties I encounter. I also would express my love to my son Nethum for being my stress reliever at the times I got frustrated and bringing a smile to my face. Without the unconditional support from my family, it would have never been possible to accomplish this thesis.

Last but not least, my heartiest thanking goes to my mother Podi Menike and my father Kumarawansa for all their true love, constant encouragement and generous support in all my pursuits. All the achievements and success I have gained in my life today is a reflection of absolute commitment and sacrifices they made for the family.

Preface

The content related to this thesis has been partially published in an international conference and in a scientific journal. The publications has beed carried out during the period of PhD candidature of the author of the thesis.

- Chinthaka M. Dissanayake, K. Ramamohanarao, M. N. Halgamuge, B. Moran, and P. Farrell "Propagation Constraints in Elephant Localization Using an Acoustic Sensor Network", IEEE 6th International Conference on Information and Automation for Sustainability (ICIAfS'12), pp 101-105, Beijing, China, 27-29 September 2012. [Chapter 5 partially contains meterial from this publication. Appendix A presents the published material in the international conference.]
- Chinthaka M. Dissanayake, K. Ramamohanarao, M. N. Halgamuge, and B. Moran, "Improving Accuracy of Elephant Localization using Sound Probes", Applied Acoustics, Volume 129, pp 92–103, 2018.

[Chapter 6 and Chapter 7 partially contains meterial from this publication. Appendix B presents the published material in the international journal.]

To my parents & family...!

Contents

1	Intr	oduction	1
	1.1	Motivation	2
	1.2	Key Challenges	3
	1.3	Scope and Limitations	3
	1.4	Contributions of the Thesis	4
	1.5	Structure of the Thesis	5
	1.6	Publications by the Author Related to the Thesis	8
2	Lite	rature Review	9
	2.1	Sensor Networks for Habitat Monitoring	9
	2.2	Current Strategies Deployed in Mitigating HEC 1	.0
		2.2.1 Combative Methods	0
		2.2.2 Preventive Measures 1	.2
3	Soci	al Behaviour and Vocalizations of Elephants 1	17
	3.1	Introduction	7
	3.2	Social Structure of Elephants	.8
	3.3	Social Behaviour of Elephants	20
	3.4	Elephant Communication and Call Patterns 2	23
		3.4.1 Elephant Rumbles	24
		3.4.2 Frequency Comparison with Other Animal Calls 2	27
	3.5	Chapter Summary	28
4	Eler	bhant Localization System Framework 2	29
	4.1	Introduction	29
	4.2	Timing Observations and Localization 3	30
		4.2.1 TOA Measurements	30
		4.2.2 TDOA Measurements	30
	4.3	Mathematical Models for Elephant Localisation System	30
		4.3.1 Best Linear Unbiased Estimator (BLUE) Model	31
		4.3.2 Performance Evaluation on Loclaization using BLUE Model 3	35
		4.3.3 Maximum Likelihood Estimator (MLE) Model	38
	4.4	Chapter Summary	3

5	Pro	pagation Constraints in Elephant Localization	45
	5.1	Introduction	45
	5.2	Sound Propagation in Air	46
		5.2.1 Sound Propagation Model	46
		5.2.2 Temperature Effect	47
		5.2.3 Wind Effect	48
	5.3	Localization Algorithm	49
	5.4	Simulation Framework	52
		5.4.1 Vocalization Detection	52
		5.4.2 Sensor Placement and Source Localization	53
	5.5	Results and Discussions	54
	5.6	Chapter Summary	58
6	Loc	alization Accuracy Enhancement Technique	59
	6.1	Introduction	59
	6.2	Localization Algorithm	60
	6.3	Average Sound Speed Estimation Methodology	62
	6.4	Simulation Framework	66
	6.5	Performance Evaluation on Probe Technique	69
	6.6	Chapter Summary	72
	6.7	Supplementary Material	73
		6.7.1 Environmental Parameter Variation Maps	73
		6.7.2 Source Localization under Sound Speed Variation	73
		6.7.3 Algorithm for Source Localization with Probes	77
		6.7.4 Performance of the Probe Technique under Different Environmen-	
		tal Conditions	77
7	Sen	sor Hardware Implementation, Data Collection and Analysis	85
	7.1	Introduction	85
	7.2	Standalone Signal Recording System	86
	7.3	Central Processing Wireless Sensor Network	88
	7.4	Experimental Setup	91
		7.4.1 Open Field Experiment	92
		7.4.2 Real Forest Experiment	93
	7.5	Results and Discussions	97
		7.5.1 Open Field Experiment Results	97
		7.5.2 Real Forest Experiment Results	99
	7.6	Chapter Summary	106
8	Cor	nclusion	107
	8.1	Summary of Contributions	107
	8.2	Future Research	110

A	Propagation Constraints in Elephant Localization Using an Acoustic Sensor Net-		
	work	113	
	A.1 Introduction	113	
	A.2 Published manuscript	114	
B	Improving Accuracy of Elephant Localization using Sound Probes	119	
	B.1 Introduction	119	
	B.2 Published manuscript	120	

List of Figures

3.1	A spectrogram of a recorded elephant rumble call. Eyebrow-shape struc- ture indicates the harmonics present in the elephant in the signal.	24
3.2	A spectrogram of an elephant rumble call while signal is corrupted by noise (AWGN with SNR = 10 dB)	25
3.3	A spectrogram of an elephant rumble call while signal is corrupted by noise (AWGN with SNR = -8 dB)	25
3.4	Time domain representation of a recorded elephant rumble call. Note that the recorded signal contains minimal background noise. However, detec- tion will be much more challenging in the presence of environmental noise	26
3.5	Time domain representation of elephant rumble call while the signal is corrupted by noise (AWGN with SNR = 10 dB)	26
3.6	Time domain representation of elephant rumble call while signal is corrupted by noise (AWGN with SNR = -8 dB).	27
3.7	The frequency ranges of the emphasized frequencies of vocalization in a large range of land—dwelling animals, plotted as a function of the mass of the animal [115]	27
4.1	Source localization using <i>N</i> number of sensors. The nominal source position (x_n, y_n) assumed to be known. It is necessary to determine actual source position (x_s, y_s) .	31
4.2	Source position estimation using algorithm described in Section 4.3.1. TDOA measurements from five sensors are utilized. The estimated source position in each iteration is depicted	36
4.3	Sensors were deployed in a single line. Number of sensors were changed from 3 to 40. For each trial locations of the sensors were changed 100 times randomly. Note that the successive estimations of source always converge before 10 iterations.	37
4.4	Sensors were deployed in two lines below the source. Number of sensors were changed from 3 to 40. For each trial locations of the sensors were changed 100 times randomly. Note that the successive estimations of	
4.5	source always converge before 10 iterations	37
	were changed 100 times randomly	38

4.6 4.7	The effect of number of sensors over the probability of success within three iterations. (The result was obtained from 10000 executions) Confidence assessment on localisation estimation. Uncertainty in estimation	39
	tion for arbitrary source location (450, 800) when $\sigma = 0.2$ is illustrated by the Covariance error Ellipse.	42
5.1	Wind effect on sound speed in the direction of propagation. Effective component of V_2 along the elephant—sensor direction should be incorporated.	49
5.2	<i>N</i> number of sensors are deployed at the vicinity of the expected elephant location. Here, the distance difference between source and the p^{th} and q^{th} sensor is defined as g_{nq} .	50
5.3	The source localization error variation at different source positions within the observation area. Four sensors are deployed in curvature with equal arc length separations as discussed in Section 6.4. The $T = 22 \ ^{\circ}C$ uniform throughout the area $V = 0 \ ms^{-1} \ \theta = 270^{\circ} \ \text{SNR} = 10 \ \text{dR}$	55
5.4	The source localization error variation at different source positions within the observation area. Four sensors are deployed in curvature with equal arc length separations as discussed in Section 6.4. The $T = 30$ ⁰ C uniform	55
	throughout the area. $V_2 = 0 \text{ ms}^{-1}$, $\theta_2 = 270^{\circ}$, SNR = 10 dB	55
5.5	The source localization error variation at different source positions within the observation area. The sensors are deployed in curvature with equal arc length separations as discussed in Section 6.4. The Temperature varies between each sensor path. $T_{S1} = 25 \ {}^{0}C$, $T_{S2} = 23 \ {}^{0}C$, $T_{S3} = 22 \ T_{S4} = 24 \ {}^{0}C$, $V_{2} = 0 \ {\rm ms}^{-1}$, $\theta_{2} = 270^{0}$, SNR = 10 dB	56
5.6	The source localization error variation at different source positions within the observation area for wind velocities $V_2 = 5 \text{ ms}^{-1}$, $V_2 = 10 \text{ ms}^{-1}$, $V_2 = 15 \text{ ms}^{-1}$, $V_2 = 20 \text{ ms}^{-1}$. Four sensors are deployed in curvature with equal arc length separations as discussed in Section 6.4. The $T = 22 \ ^0C$ uniform throughout the area, $\theta_2 = 270^0$, SNR = 10 dB.	56
5.7	The source localization error variation at different source positions within the observation area. Four sensors are deployed in curvature with equal arc length separations as discussed in Section 6.4. The T = 22 $^{\circ}C$ uniform throughout the area. $V_2 = 0 \text{ ms}^{-1}$, $\theta_2 = 270^{\circ}$, SNR = -15 dB.	57
6.1	<i>N</i> number of sensors are deployed at the vicinity of the expected elephant location. The distance difference between source and the p^{th} and q^{th} sensor is defined as q_{th} .	60
6.2	The <i>i</i> th sensor (<i>S_i</i>) receiving the signal from the probes <i>P_l</i> , <i>P_m</i> , <i>P_n</i> . The wind flows with speed of <i>V_w</i> with an angle of θ_w with respect to the horizontal axis. The temperature affected speed of sound between <i>P_z</i> is <i>V_{P_zS_i}</i> (<i>T</i>) where $z = l, m, n$.	63
6.3	Example observation area with variable sound speeds. Each grid can be considered as a microclimate region with a particular wind velocity W_l and temperature T_l where $l = 1, 2, \dots, 9$. The effective sound speed varies according to the environmental parameters at each grid	68

70

71

- 6.5 Normalized source localization error variation at different source positions within the observation area. At every sensor and probe position, a **wind** and **temperature** sensors were incorporated as in Comparison Model 2 in Section 6.3. Gaussian function was utilized to model the temperature and MRF was used to model the wind velocities over the observation field. The sound speed is then determined considering the above environmental model values. (i.e. Not through probe technique). Ten probes and 12 sensors are deployed as discussed in Section 6.4. The temperature and wind velocities are non-uniform over the area. (Simulation parameters were chosen as follows: $T \in \{22 32\}^{\circ}C$. Initial mean values for wind are $V_w = 20 \text{ ms}^{-1}$, $\theta_w = 120^0$. SNR = 10 dB.)

- S8 Wind speed variation map over the simulation area. The wind direction and the speed variation over the area are represented assuming Markov random field property. Simulation parameters were chosen as follows: Initial mean values for wind speed is $V_w = 20 \text{ ms}^{-1}$ and $\sigma_{wSpeed} = 0.5. \ldots$ 75
- S9 Wind direction variation map over the simulation area. The wind direction and the speed variation over the area are represented assuming Markov random field property. Simulation parameters were chosen as follows: Initial mean values for wind direction is $\theta_w = 120^0$ and $\sigma_{wDirection} = 0.1.$. . . 77

S10	Sound source localization under variable sound speeds in observation area. Each grid consists of constant wind velocity and temperature. However, temperature and wind velocity varies between each grid. Therefore, sound signal experiences different sound speeds in diverse regions it is travelling in. For instance, the effective sound speed in the path of <i>E</i> and S ₁ at grid 2 is given by $C_{S_1E2} = C(T_2) + V_{w2} \cos \xi_{S_1E2}$. Similarly, the sound propagation speed between <i>E</i> to Sensor S ₁ will be affected by temperatures T ₂ , T ₄ , T ₅ , T ₇ and wind velocities V _{W2} , V _{W4} , V _{W5} , V _{W7} in between the propagation path.	78
S11	Summarized algorithm used in simulations explained in Section 6.4 of this chapter. The above algorithm is used to estimate unknown source (Elephant) location incorporating probe technique.	79
S12	Performance analysis of three different sensor/ probe models under the given temperature and wind velocity variation. Sensor/ probe setup models are described in Section 6.3. In this scenario temperature is kept constant over the area. Ten probes and 12 sensors are deployed as explained in Section 6.3-Model B of this chapter.	80
S13	Performance analysis of three different sensor/ probe models under the given temperature and wind velocity variation. Sensor/ probe setup models are described in Section 6.3. In this scenario wind velocity is kept constant over the area. Ten probes and 12 sensors are deployed as explained in Section 6.3-Model B of this chapter.	81
S14	Performance analysis of three different sensor/ probe models under the given temperature and wind velocity variation. Sensor/ probe setup models are described in Section 6.3. In this scenario performance under low variability of temperature and wind velocity is depicted. Ten probes and 12 sensors are deployed as explained in Section 6.3-Model B of the this chapter.	82
S15	Performance analysis of three different sensor/ probe models under the given temperature and wind velocity variation. Sensor/ probe setup models are described in Section 6.3. In this scenario performance under highly varying of temperature and wind velocities is depicted. Ten probes and 12 sensors are deployed as explained in Section 6.3-Model B of the this chapter.	83
7.1	Microcontroller interface overview. RA0 PIN of the PIC is configured as ADC input channel. RC2 – RC5 PINs are interfaced with MMC with simple resister-based voltage converter. PortD is interfaced to 16×2 LCD.	87
7.2	Circuit diagram of the signal pre-amplifier. The amplifier was designed with Proteus 7.7 and simulated before implementation. The amplifier performs well in the frequency range of $10 - 4000 \text{ Hz}$ range	88
7.3	Completed standalone recorder hardware system. The system consists of mic interface, power supply unit, integrated display interface and memory (MMC) interface.	89

7.4	Snap of complete sensor unit deployed in the field. The microphone and the transmitter units are mounted on a tripod and units are protected with weather shields. Four sets of similar sensor units are deployed for the	
	evention of the transmitter units were nevered by 12 V DC bettery	00
75	Span of Droho unit doployed in the field. The unit consists of sound shire	90
7.5	Shap of Probe unit deployed in the field. The unit consists of sound chirp	
	generator and a controller unit which determines the timing of chirp gen-	01
-	eration. The unit was powered by a 12 v DC battery.	91
7.6	Simplified block diagram of the system hardware implementation. Each	
	sensor consists of microphone and a FM transmitter. Vocalization data	
	and the probe signals acquired through each microphone is transmitted	
	through a dedicated FM channels to the centralized receiver. Probes are	
	placed in the forest area to get a better representation of the real sound	
	speeds of the observation area.	94
7.7	Snap of a Probe unit deployed in the forest implementation. The unit was	
	deployed in a predetermined location in the forest area to obtain a better	
	representation of the sound characteristics through the woodland	95
7.8	Snap of a Sensor unit deployed in the forest implementation. The unit was	
	deployed in a predetermined location close to the village boundary	96
7.9	A snap shot of recording information on a location where elephants were	
	present in a day that we conducted the experiment. GPS coordinates and	
	information on the observation were recorded manually for further refer-	
	ence	97
7.10	Picture of footprint embossed on the ground of an elephant that visited the	
	area on Day 1 when we conducted the experiment. Observation locations	
	was recorded for further reference.	98
7.11	Snap of fresh pile of elephant dung on the ground in the elephant track.	
	This observation was made on Day 1 of the forest experiment. Observation	
	location was recorded for further reference.	99
7.12	Snap of earth marks made due to elephat bashings on the ground by an	
	elephant on Day 3 of the foreset experiment. Observation location was	
	recorded for further reference.	99
7.13	Snap of marks made on a tree by an elephnat on Day 3 of the foreset ex-	
	periment. Observation location was recorded for further reference	100
7.14	Localization of Surrogate Elephant; The analysis results from the Data Set	
	1 acquired from the real experiment.	101
7.15	Localization of Surrogate Elephant; The analysis results from the Data Set	
	2 acquired from the real experiment.	102
7.16	Localization of Surrogate Elephant; The analysis results from the Data Set	
	3 acquired from the real experiment	103
7.17	Analysis results of the real forest experiment. Observations were associ-	
	ated to different tracks in Day 1 and Day 3. The estimated elephant local-	
	ization positions closely follow the elephant tracks. Evidently, the probe	
	technique has significantly improved the localization accuracy despite the	
	complex terrain.	104

Chapter 1 Introduction

E LEPHANTS and humans existed with a long tradition of interdependence up until contemporary times. Elephants figured prominently in work, warfare, culture, religion and pageantry throughout history [1]. Changes in social and ecological conditions precipitated by human needs are resulting in serious depletion in certain animal populations. Human-elephant conflict (HEC) is one such significant socio-ecological problem in some parts of the world. The conflict primarily is a consequence of frequent attacks from crop-raiding elephants on the rural agricultural communities. Humans are clearing large areas of land for food production, increasing pressure on traditional elephant habitats. This results in migration of elephants to villages to fulfill their food requirements and ensuing HEC.

In Sri Lanka where the field trials for this research was primarily conducted, it is estimated that there are approximately 4,200–5,000 wild elephants. Typically, one elephant needs at least five square kilometres of optimum habitat to inhabit [2]. In consequence, the population require in total, minimum of 21,000 to 25,000 square kilometres or 32–38% of the overall land area of the country [2]. The national parks only cover 12.5% of the total requirement and almost 70% of the elephants live outside the national parks, sharing the land with rural people. As a result, human-elephant conflict is common in eight of the nine administrative provinces in Sri Lanka affecting over three million people.

The consequences of HEC are the loss of human lives or permanent disability, damage to properties and cultivation while creating negative attitudes towards the elephants [3]. From 2012 to 2016, in Sri Lanka alone 316 people were killed by elephants and 1,171 elephants were killed by people. During the same period, an overall total of 6,415 residences in the country were damaged by elephants. Until 2007 the average number of elephants killed per year in Sri Lanka was around 150. Since 2012 this number has increased to 250 elephants per annum and in 2016 the elephant deaths reached its recorded highest of 279 [4]. Over 90% of the elephant fatalities were resulted from retaliatory killing by farmers for destroying their crops, homes and lives [2]. The gravity of the HEC is serious considering the fact that the Asian Elephants (*Elephas maximus*) categorized as *Endangered* in the International Union for Conservation of Nature (IUCN) Red List [5].

1.1 Motivation

Various strategies have been implemented to mitigate HEC. Traditional methods such as chasing elephants away by shouting, building fires and lighting fire crackers are still used by rural farmers and villages. Approaches such as implementing electric fencing systems and using satellite imaging and radio and Global Positioning Systems (GPS) collars to detect the presence of elephants [6] are also employed with the support of government authorities and conservation support groups. However, these strategies suffer from inherent drawbacks ranging from lack of effectiveness to inadequacy of resources [7].

Within the current context of human-elephant conflict, it is important to take every possible action to manage the problem at tolerable levels. Accurately detecting the presence of elephants plays a significant role in this process [8–10].

Recently, interest has grown in the detection of elephants through their low-frequency calls, commonly referred to as "rumbles" [7]. This technique is a safe, practical and non-intrusive methodology to detect these highly social animals that use rumbles for long distance communication. Once the elephants are detected at a sufficient distance from the village boundaries countermeasures and warnings to the local population can be deployed. The proposed solution in this thesis is basically to deploy an acoustic sensor network to detect and localize elephants before they move into the village area.

1.2 Key Challenges

In this thesis, the research question addresses a multidisciplinary real practical problem. The following key challenges are identified as the research problems.

- General and favourable elephant behaviour patterns need to be identified for effective implementation of a theoretical and practical elephant detection framework.
- Sensors should be able to detect the presence of elephants. This will, in practice, be achieved by positively identifying low-frequency elephant calls originating from a location where is a long distance away from the area of interest.
- Finding the location of elephants in the given environment is a challenging task. Mathematical modelling need to be performed to determine the effectiveness of acoustic localization algorithms.
- These infrasound calls travel through the air and the effects of changes in the medium need to be determined. For instance, the environmental factors such as temperature and wind velocity change the speed of sound in air. Quantitative analysis needs to be performed to estimate the effect on such variations in localization accuracy.
- A practical hardware system needs to be designed and implemented to gather data for analysis purposes. The designed hardware should be capable of implementation in a real-world system.
- The number of actual implemented wireless sensor network systems is extremely low compared to the theoretical publications presented in the wireless sensor network area. To fill the gap, sensor network hardware systems need to be designed and implemented to test and validate proposed theoretical framework and algorithms.

1.3 Scope and Limitations

This thesis investigates the feasibility of utilizing acoustic sensor network to detect the presence of elephant and accurately estimate the elephant location. Challenges in lo-

calization estimation are extensively discussed and naval methodology to improve the localization accuracy is proposed and implemented in this thesis.

Real-world implementation in this thesis focuses on proof of the concept, and due to time and resources constraints, data collection is limited to above aspect. Challenges in the long-term operation of the system framework proposed are out of the scope of this thesis. The thesis focuses on detecting the presence of the elephant(s). Feature extraction or similar techniques has not been utilized in this research to classify specific elephant or estimate the number of elephants in the vicinity of the network.

1.4 Contributions of the Thesis

Major contributions are listed in the order they appear in the thesis.

- 1. Evaluation of the key elephant social behaviours and characteristics to identify favourable behavioural patterns for an elephant detection and localization system.
- Development of the mathematical model to estimate source (elephant) locations with a passive acoustic sensor network and analysis of the network scale on localisation accuracy.
- 3. Evaluation of probabilistic bounds and error validation of the elephant localization model, identifying probabilistic estimation error regions.
- Proposal for a simulation framework to study the performance of the designed elephant localization algorithms considering real-world model inputs.
- 5. Investigation of the impact of environmental constraints and signal to noise ratio on an elephant localization sensor network. Identification of the variables that significantly compromise the estimation accuracy with respect to the application.
- 6. Introduction of a novel "probe" technique to estimate the accurate sound speed in the air which is affected by the environmental parameters of the medium.

- 7. Comparative study of the proposed novel probe technique under different environmental scenarios and evaluation of position estimation error.
- Design and implementation of a hardware system framework addressing our application requirements and including the novel probe technique and deployment. Collection of data in real-world and forest environment using developed hardware system.
- 9. Practical validation of proposed methodologies system consists of both hardware and software in Sri Lanka.

1.5 Structure of the Thesis

After this introductory chapter, the thesis is organized as follows:

Chapter 2: Literature Review

This chapter carries out a literature survey related to the thesis. In our approach, we utilize a sensor network to detect and localize elephants. A brief review of implemented wireless sensor network applications for habitat monitoring has been discussed in the early part of the chapter. Even though there are extensive theoretical publications related to wireless sensor networks, practical implementations are limited. The later parts of the chapter present the current strategies utilized in mitigating HEC. There are wide limitations in current approaches in terms of effectiveness and resource availability and these concerns are elaborated in Chapter 2.

Chapter 3: Social Behaviour and Vocalizations of Elephants

Understanding animal characteristics and behaviour under different socio-ecological conditions is important in any conservation project. This chapter presents the behaviour patterns and the social structure of elephant society. Important behavioural characteristics of elephants utilized in this thesis are highlighted and discussed. Elephants are highly social animals and they make long distance rumble calls for their communication. The frequency and time domain responses of these elephant calls are presented in this chapter whilst evaluating the effects of noise levels.

Chapter 4: Elephant Localization System Framework

This chapter describes the underlying mathematical representation of the elephant localization system framework. Initially, we explain the passive acoustic localization approaches in terms of Time of arrival (TOA) and time difference of arrival (TDOA) techniques and then mathematically model the elephant localization problem using timing observations. Elephant location estimation is obtained using two estimation models namely the Best Linear Unbiased Estimator (BLUE) and the Maximum Likelihood Estimator (MLE). Then the performances of the estimator models are evaluated in a simulation framework and results are presented. Finally, statistical bounds on the elephant localization error estimation is derived and illustrated in this chapter.

Chapter 5: Propagation Constraints in Elephant Localization

The passive acoustic sensor network described in this thesis utilizes elephant rumble calls that travel through the air as the signal. Sound propagation in air is affected by various environmental parameters, especially the temperature. In addition, the wind carries the medium that the sound propagates and wind velocity affects the speed of sound in air. Moreover, the intensity of the signal is inversely proportional to the distance it travels and therefore, signal to noise ratio (SNR) of the received signal depends on the source-sensor distance. This chapter quantitatively analyses the effects of meteorological parameters on the speed of sound in air and evaluates the significance of the phenomenon on elephant localization accuracy. Additionally, the chapter investigates the SNR dependence of the elephant localization accuracy. Analyses are performed on a simulation testbed defined with respect to the application and results are presented.

Chapter 6: Localization Accuracy Enhancement Technique

The extensive analysis in Chapter 5 emphasises the importance of incorporating the wind and temperature information into the elephant localisation system to attain acceptable positioning accuracy with respect to the application. In this chapter, we propose a novel technique to improve the localization accuracy of the elephant localization framework. The novel mechanism basically incorporates sound generating probes into the sensor network providing a means to estimate the accurate speed of sound in the observation area. The chapter presents an extensive simulation analysis on this methodology as well as interesting findings concerning the effectiveness of using sound probes in elephant localization system framework.

Chapter 7: Sensor Hardware Implementation, Data Collection and Analysis

Human-elephant conflict is a real-world problem and countermeasure require practical implementations. This chapter provides a detailed explanation of the hardware implementation of our approach. We have conducted experiments using the sensor network that we built in an outdoor environment to understand the effectiveness of the system under different ecological conditions. Elephant calls were replicated by replaying the rumble calls from a surrogate elephant system and data was collected and analysed. An identical sensor network system was then deployed to localize wild elephants in Sri Lanka. This chapter presents the results of out-door experiments performed and experiments carried out in a forest environment using our sensor-probe framework.

Chapter 8: Conclusion

This chapter summarizes the contribution of the thesis and provides some insight for future research.

1.6 Publications by the Author Related to the Thesis

Journal Article

 Chinthaka M. Dissanayake, K. Ramamohanarao, M. N. Halgamuge, and B. Moran, "Improving Accuracy of Elephant Localization using Sound Probes", Applied Acoustics, Volume 129, pp 92–103, 2018

International Conference Article

 Chinthaka M. Dissanayake, K. Ramamohanarao, M. N. Halgamuge, B. Moran, and P. Farrell "Propagation Constraints in Elephant Localization Using an Acoustic Sensor Network", IEEE 6th International Conference on Information and Automation for Sustainability (ICIAfS'12), pp 101-105, Beijing, China, 27-29 September 2012.

Chapter 2 Literature Review

Our proposed solution to mitigate human-elephant conflict incorporates a wireless sensor network to detect and localize elephants. Therefore, the first part of this chapter we present a brief literature review related to wireless sensor networks for habitat monitoring. Despite the extensive theoretical publications in the area, there are quite limited practical implementations conducted on the wireless sensor network on monitoring animals in their natural habitats. In the latter part of the chapter, we mainly discuss the currently implemented strategies to mitigate HEC. We further explain the practical limitations and drawbacks in terms of effectiveness and resource constraints with related to these methodologies.

2.1 Sensor Networks for Habitat Monitoring

PREVIOUS wildlife monitoring employed for large mammals has almost entirely been based on reasonably low-technology VHF transceivers that regularly send out a ping signal [11]. Consequent advancements have included global positioning system (GPS) based trackers, which have been used for tracking numerous animals including birds [12] and sea turtles [13], but these depend on high-power transmitters that transmit data up to a satellite, and they function from a non-recharged battery supply.

One of the earliest researches carried out using sensor network for habitat monitoring is the monitoring of seabird nesting environment at Great Duck Island [14]. There, researchers deployed the sensors statically in a grid-like fashion across two wildlife habitats. These sensors observe the animals and report the observation back to the base station through a sensor network. ZebraNet [15] was another wireless sensor network (WSN) which was designed for monitoring and tracking zebras at the Sweetwaters reserve, Kenya. The system includes custom tracking collars (nodes) carried by animals under study and the collars operate as a peer-to-peer network to deliver logged data back to researchers. The collars include a wireless computing device which consists of Flash memory, GPS module, wireless transceivers, and a small central processing unit (CPU). The architecture had been designed for an always mobile, multi-hop wireless network.

WSNs for habitat monitoring have been studied by several other research groups as well. Multi-tiered architecture for habitat monitoring was proposed by Cerpa *et al.* [16]. A system was developed based on PC104 hardware platform and it was introduced as an experimental test bed for habitat monitoring. Wen Hu *et al.* [17] investigate a wireless, acoustic sensor network application for monitoring Cane Toads populations in the monsoonal woodlands of northern Australia. They have used a hybrid mixture of resource-rich Stargate devices and resource-poor Mica2 hardware devices. Mica2s were used to collect acoustic samples and expand the sensor coverage while Stargates were used for resource-intensive tasks such as fast Fourier transforms (FFT) and machine learning. Recently, it has been presented a vision-based monitoring system for biological phenomenon [18]. This system has enabled the automated analysis of thousands of images in a study of avian behaviour during nesting cycles.

The amount of implemented wireless sensor network systems is extremely low compared to the theoretical publications presented in the area. There is certainly little data about the extended behaviour of the sensor networks, due to much less wireless networks utilized for habitat monitoring.

2.2 Current Strategies Deployed in Mitigating HEC

Methods currently adapted in mitigating HEC can be divided into two main categories [19]; combative methods and preventive measures.

2.2.1 Combative Methods

Combative methods are mainly used in dealing with the elephants after they elephants entered to the protected area.

Use of Spotlights

Use of powerful spotlights have been tried in dealing with ccrop-raiding elephants in India and it has been found that some elephants tend to move away from the beam of light [20]. The light needs to be powerful and strong enough to be effective and if used torchlight, in fact it attracts curious elephants towards the source. The spotlight fitted in vehicles such as tractors and jeeps were also experimented, however, vehicles need to be close enough to the elephants for this method to be effective. However, once the vehicle is moved away, the elephants are very smart and tend to come back [21].

Use of Domesticated Elephants

Domesticated elephants (called koonkies) are utilized to drive away the wild elephants from the crop fields in India, mainly over the daytime [20]. In 1980, a herd of around 60 wild elephants was chased away successfully by using domesticated elephants in West Bengal in northeast India. Capturing and domesticating elephants is a time consuming, social-ecologically unfriendly and bureaucratically unrealistic approach.

Use of Firecrackers

Use of firecrackers is one of the most generally used methods to drive the wild elephants away from cultivated areas. Elephants are intelligent creatures and soon identifies such psychological bluffs. The use of rockets that ends with a bang was appeared to be more successful, especially with a group of elephants. Farmers used bamboo gun rockets effectively to chase wild elephant away from of cultivation in India [22]. In the past, wildlife department authorities used to supply thunder flashes to the farmers. These devices were successful in mitigating elephant problem in short-term, but not a sustainable solution to eradicate elephant raids [23].

Use of Loudspeakers

An experiment has been done in southern India, with playback of a tape contains a jumble of noises via loudspeakers and found effective in short-term, moving raiding bull elephants from coconut farms [21]. However, a controlled experiment carried out using recorded tiger calls for deterring elephants was inconclusive [20]. In [24], it has been suggested that high-frequency sound beep may have a possibility of repelling elephants, but would not be a standalone solution.

Use of Firearms

Shooting the guns or firearms may be effective in driving away elephants back to the forest. However, elephants being one of the endangered species, shoot to kill is not authorised in any of the elephants residing countries. Therefore, the farmers and wildlife officers must use the noise of shooting to chase away the elephant. However, some aggressive bull elephants may disregard such noises and move into farming areas [19].

2.2.2 Preventive Measures

Preventive strategies are deployed to avoid elephants from migrating into protected areas.

Use of Repellents

One of the main sensory channels of elephants is the smell. Considering this fact, chemical repellents have been used as barriers for elephants without much success [20]. This is mainly due to the difficulty of keeping chemicals in the effective conditions under wet weather. However, Dutch planters in Sumatra tried to hang rags soaked in human urine as a repellent to deter wild elephants from oil palm plantation. In the recent past, experiments were conducted on free-ranging elephants in Zimbabwe to test the effectiveness of Capsicum-based aerosol as a repellent. The results have illustrated that it has a potential of being a short-term repellent up to 100 m [25]. Use of chilli briquettes found to be effective as a short-term deterrent for crop raiding elephants. The chilli briquettes were made of ground dried chillies combined with fresh elephant or cow dung and dried the mixture to make briquettes [26, 27]. Burning these briquettes create a noxious smoke which act as the repellent for elephants. However, the effectiveness of this method significantly depends on the wind direction [26]. If the prevailing wind drifts towards the crop field, the chilli-dung fires are of no-use as the smoke do not warn the elephants in the forest area.

Trenches

Trenches to be effective, it needs to be at least 2 *m* deep and 2 *m* accords at the top and 1.8 *m* wide at the base [19]. In addition to the higher cost of building them, because of wet weather and erosion, the effective depth of trenches often gets affected reducing the effectiveness of the barrier [28]. The potential of the trench can be enhanced by either establishing the walls with concrete (which is expensive) or by planting thorny vegetation along the inner-trench edge. Shallow trenches 1.2 m deep and 1.2 m wide along with a cover of bamboo matting has been used as a barrier in India. The bamboo mat provides a psychological rather than a physical barrier as the elephants have an instinctive sense of what surface could withstand their weight [20]. The cost of construction and maintenance of the trenches usually outweigh the extent to which it is effective in reducing the damage that elephants do to the villages or farming areas.

Log Fences and Stone Walls

In Malaysia, it has been used log fences (15 *cm* to 25 *cm* in diameter) wrapped in barbed wire in highly affected areas with HEC [19]. Constructing these fences are highly expensive and it is difficult to extract appropriate timber from the forest. Similarly, constructing stone walls also a tedious and expensive matter and as a result not a viable option for many areas.

Electric Fence

Electric fences got a higher attraction out of the existing solutions. Many people in the affected areas tend to plea electric fences, as they see it as the permanent solution for HEC. However, the high capital cost and the lack of routine maintenance have always resulted in the effectiveness of this approach to put well below expectations [29,30].

The electrical fences consist of two strands of steel wires which are connected to a pulse generator. The power supply often operated either by battery or solar panels. The pulse generator passes an electrical pulse of 5,000 volts in every second through strand wires, which is non-lethal as of its very short duration (3/ 10,000 th of a second). The system generally works as a mental barrier for elephants rather than a physical obstacle. However, it is difficult to outsmart the elephants and they have used several tricks to get over these fences. The ivory of the tuskers is a non-conductive and they use it to dislodge the fence or insulators. In some cases, it is observed that elephants use fresh trees which they bring from the forest to trip-off the power supply by short-circuiting the wires. In other incidents, elephants have used tree branches to break the posts and wires [31].

Demand for meticulous maintenance is another factor that causes electrical fences to fall below the anticipated success in the long run. Keeping the fence line clear from the growing vegetation, especially in the wet season is a recurrent problem that needs to address in the management of electric fences. The effort to establish Collective maintenance of an electric fence through a rural community has not often been successful due to the involvement of a long chain of responsibility, that simply fails at the weakest link. Even in countries where wildlife is authorities operate in a local level the success of electric fencing projects is quite unsatisfactory due to the maintenance deficiencies [29, 32].

Beehive Fence

It has been identified through playback experiments that elephants tend to avoid sounds of disturbed honey bees [33]. The stories from local people suggested that elephants fear bees as a result of stung behind the thinner skin of the ears, around the eyes and up the trunk. This must be a painful enough experience for elephants which creates nega-
tive memories leading them to avoid future encounters with bees. Recently, researches have been conducted to monitor the effect of the beehive fences on deterring crop-raiding elephants [34–36]. A beehive fence consists of beehives that are installed into the wires connected to the posts and usually use to surround agricultural area which need to be protected.

Effectiveness of these fences highly depend on the meticulous maintenance which is labour intensive. In addition, it could take a longer period; in some cases, months for all the beehives to be occupied with bees which minimize the effectiveness of the whole fence during the interim. Further, less aggressive honey bees may have less effect when swarming out and deterring elephants. Bees are less active in the cooler conditions and as a result the effectiveness of the fence is deteriorated during the night where most cropraiding occurs [35].

Translocation

Identify and relocate the problematic elephants who are constantly responsible for the destruction of plantation or impairment of villagers' livelihood has been a practice followed by some wildlife authorities [19, 31]. These elephants are either captured and moved by a vehicle (capture transport) or driven to a new area (elephant drives). However, in most cases elephants have returned to its usual habitats within a brief period after the translocation. It has been reported cases where elephants returned to the area of capture from a distance as far as 100 km [37].

Tagging Elephants with GPS Collars

During the past few years, feasibility to use Global Positioning System (GPS) collars to track elephants in the wild has gained attraction of the conservationists' and other stake-holders who are addressing the issue with HEC [6]. Generally, the collar consists of a GPS receiver, a radio-modem for data communication, a medium for data storage typically a non-volatile memory, and an independent VHF or mobile of satellite transmitter or and a battery. Once the elephants were collared, the location data is retrieved typically in 4-6

hour intervals. Subsequently, the elephant moving speed can be estimated by dividing the linear distance between successive locations by the time interval between them.

Tagging elephants with GPS collars is an expensive, dangerous to humans and elephants and tedious task which require complex procedures to be followed. First, the elephant needs to be captured and tranquilized. In general, gun-propelled syringes containing Etorphine is used to dart the elephants and this task need to be carried out by an expert personnel. The process involves the risk of approaching the elephant up to the close proximity of the propelled-gun range. Typically, the elephants were approached on foot or ground vehicle, but there were cases reported where it used helicopters in the process of darting [6]. While the elephant is anesthetized, heart rate, vital signs, respiration rate, blood pressure and body temperature need to be monitored continuously. Once the collaring is completed a Diprenorphine need to be injected as the reversal drug and need to make sure that elephant is act natural after recovery which again involves the risk of being near the treated animal.

Elephant collars need to be built to sustain in harsh environmental conditions and need to be water resistant. Collaring being a complex process, it needs to be robust enough to last for a longer period once it is worn. Therefore, the cost of elephant collars is considerably high (USD 3000-12000) [38] and the command unit for the collar is also expensive (USD 6500). In addition, there is a data retrieval cost involve accessing the GPS satellite data. For instance, a project tracking around 10 elephants over a period of 2 years can approximately cost at around of \$200,000 onwards making the solution unaffordable for wildlife managing government authorities in most elephant residing countries [38]. Because of the risk involve in collaring elephants, the need for expert technical personnel and higher capital and operational cost, GPS-telemetry is not a sustainable method for monitoring elephants. In addition, due to the infrequent data retrieval intervals, the approach is not a viable solution for HEC.

Considering the drawbacks and constraints in the current strategies deployed in mitigating HEC, it required to investigate a possible alternative to address the problem. In this thesis, we propose to use passive acoustic sensor network to detect and localize elephants which is an effective methodology in mitigating the occurrences of HEC.

Chapter 3 Social Behaviour and Vocalizations of Elephants

This chapter presents the social behaviour and structure of elephant society. The useful behavioural patterns of elephants for localization and tracking methodology used in this thesis are highlighted and discussed. In addition, general elephant vocalizations are described, and their frequency spectrum is examined. Characteristics of elephant rumbles in spectral and temporal domains are analysed and the favourable characteristics in regard to elephant localization are highlighted.

3.1 Introduction

THERE are three main elephant species in the world; African savannah elephants (*Loxodontaafricana*), African forest elephants (*Loxodonta cyclotis* or *Loxodonta africana cyclotis*) and Asian elephants (*Elephas maximus*) [39, 40]. However, less than 50,000 wild Asian elephants presently extant on earth and fragmented populations dispersed across 13 countries [40–42]. Elephants are found in Southeast Asia and the Indian sub-continent and with half of the inhabitants live in India [43]. Almost across the all residing territories, the main threat to Asian elephant survival is the habitat loss and fragmentation [42, 44]. Contrary to African elephants, ivory poaching is of minor concern for Asian elephants' survival as only about 5% of males carry tusks [45].

Understanding of social structure and behaviour patterns of elephants is a crucial element in an elephant conservation project. The social structure of Asian and African elephants is briefly described in Section 3.2. General elephant behaviour is discussed in Section 3.3. Elephant communication and call patterns are summarised and analysed in Section 3.4.

3.2 Social Structure of Elephants

Elephants play a crucial role in balancing ecosystem functions and biodiversity in forests [46, 47]. In [47] they were named as ecosystem engineers and mega gardeners of the forests considering their part in the transformation of landscapes, influence on vegetation regeneration, and effect on the dissemination and abundance of wildlife [48, 49]. Elephants being on the edge of extinction may cause to entire forest ecosystem to be at risk. For instance, in the tropical forest of Congo, some tree species entirely solely reliant on elephants for seed dispersal [50].

Asian elephants often inhabit densely vegetated areas [44]. Consequently, only three comprehensive studies based upon individual identifications have been conducted for wild Asian elephants [51–53].

Asian elephants occupy a variety of habitats such as scrub forests, grasslands, dry deciduous forests, swamps, moist evergreen forests and mangroves [54]. The span of their prior and current ranges demonstrates that Asian elephants are capable of adapting to a broad variety of ecological conditions. Perhaps their behavioural plasticity may have contributed this adaptability which subsequently evolved in moving into the human populated areas increasing HEC.

The social composition of wild Asian elephants can be categorised into a fissionfusion social structure [51,52]. Fission-fusion reflects an adaptive group behaviour which demonstrates the group size based on costs and benefits of association. If the benefits of association outweigh the costs, sub-groups will fuse into larger groups. On the other hand, the cost of getting together exceeds the benefits fission occurs; groups divide into sub-groups. In general, fission-fusion behaviour depends upon a number of factors such as availability of food and water resources, access to mates and predation pressure [55–57].

Group size in Asian elephant's fission-fusion social system can vary extensively from few animals to a herd of hundreds of individuals [51,52,54,58]. Asian elephants live in a range of habitats which represents diverse ecological conditions based on the availability of water and vegetation resources [42, 43, 59–61]. Wild elephants in Sri Lanka have no threat of natural predators [51], and therefore the group size mostly reflects adaptability into available local resources.

Elephants are giant herbivores [62] animals who need to consume a massive amount of food and water for their survival. In general adult elephant daily consumes about 250 Kg of forage and 180 L of water. Elephants eat most of the vegetation but, like any other large mammal (*ungulates*) [63], they specifically consume plant species and plant parts that contain less fibre and more digestible carbohydrates and proteins [64,65]. Elephants prefer young grasses over the woody vegetation as their primary food source because it contains soluble carbohydrates and lower lignin and secondary compounds [65, 66]. Thus, grasses are more digestible compared to woody plants and demand little handling time [67]. Therefore, Grass patches offer uniform concentrations of high-quality food resources for elephants [68]. Despite its higher protein content, woody vegetation such as scrub or evergreen forests is not a desired food source for elephants [63,65]. The chemical composition of woody plants varies significantly between plant parts and plant growth stages and as a result, they require longer handling time in elephant's digestive system compared to grass [67].

Similar to the most mammals, elephants depend on water for their survival; not only for drinking but for bathing and social interactions. However, they commit less than two to three hours of the day for drinking and bathing [69]. This is significantly less than the time that they spend on foraging. Depending on the age and reproductive status, elephants spend 12 to 18 hours searching for food or provisions [43, 44, 69]. Water is not just an essential resource, but it additionally provides a means for social interactions by facilitating play behaviour [70].

Water and preferable food sources are often may not be close to each other. Therefore, elephants sometimes need to travel long distances to fulfil there drinking and feeding needs [43]. The swap between drinking and feeding activities need to be synchronised among all the members of the social unit to avoid the cost of separation from the main family unit [71]. Water is less likely to be a resource of competition when it compared to food and therefore, it is possible to see elephants next to each other when they harvest water. However, when water is limited to a few hot spots, substantial aggregations of animals could be anticipated near water [72] and these may correspond to short-term

gatherings of multiple social units. Similar to African forest elephants gather by hundreds near clearings [73,74] in some habitats, Asian elephant can be seen in large groups near water sources in open grass areas or at the edges of forest and grasslands [53].

In general, animal group size increases with predation pressure and openness of habitat [63, 75] and human interference is perceived as predator threat [76, 77]. Hence, the presence of tourists in some habitats may have affected the elephant group size and persuaded the formation of larger groups composed of several social units. For instance, average elephant group size in Minneria national park in Sri Lanka is significantly higher than other local habitats and can be considered as a direct impact of human disturbance.

3.3 Social Behaviour of Elephants

Though Asian and African elephant species were separated by at least six million years [78, 79] it is possible to find a similar social organization in all species to some extent. For instance, both Savannah and Asian elephant family units are led by the eldest adult females (matriarchs) and they manage the extended association between other family units. Savannah elephants who has fission–fusion society with hierarchical 'tiers' [56, 80, 81] prefer to have companionships throughout their home ranges provided that resources permits. However, controversial behavioural observations have been made on Asian elephants [53].

A study carried out utilising population genetic data, radio telemetry and behavioural observations in Yala National Park, Sri Lanka [51] reports Asian elephant family units with adult and sub-adult females and juveniles. It has been observed low levels of association and no intergroup transfer of females in this study. On the other hand, research carried out in Udawalawa National Park, Sri Lanka depending upon behavioural observations in [52] described that associations of adult females not limited to unique family units but they formed larger social groups than in previous reports. In addition, these family units were observed to be stable across years and female elephants occasionally switched between family units [52].

Asian elephants have a lifespan of 60 years in wild and are long long-lived animals

[44,82–84]. Female elephants give birth to calves after a 20 to 22-month gestation period and calves are lactational dependent for the first 3 to 4 years of their life [44,82]. At the age of eight to nine, female elephants become sexually mature [44,85,86] and typically give birth to their first calf during 14 to 20 years of age [44,87]. Even though male become sexually matured at age of 17, they in usually will not succeed in mate until the age of 20 to 25 years. This is due to limited access to receptive female elephants and male elephants generally need to outcompete other males to connect with a female [88,89]. Even females are commonly found in groups [51,52,59,90], adult males are frequently observed alone [44,59,90].

Adult females are only receptive for a few days in their prolonged 16-week estrous cycle and they have four to five-year inter-birth period [91]. Therefore, a receptive female is an unpredictable and limited resource and male elephant competition to connect with a female is highly likely. As a result, male elephants are not benefited by permanently remain with a group of females or defending a territory [53].

The home range of individual male overlaps partly or entirely with the home range of several females. In, [92] it was investigated the home range of elephants in Yala National park, Sri Lanka using VHF tracking and have estimated that home range of male elephants could be twice as of a female elephant. In [59] it was observed that male elephants reside in an area of less than 1 km² for several days inactively before they migrate to a different region where they follow the identical pattern.

In African elephant society, sexually inactive adult males move to a location called the bull area where they interact with other male elephants in a relaxed manner. On the other hand, sexually active males or musth elephants leave bull areas and in generally observed alone or in the presence of female elephants. Elephants in musth are more often encounters aggressively with other sexually active males [93]. In African elephants, musth bulls are desired by females in estrous and sire most offspring [88,94]. However, wild Asian elephant males in and out of musth have been observed mating females [53, 95].

In general, male group size decreases with age [96] and Juvenile males can be observed in larger groups than adult or sub-adult males. Early studies on African elephants suggested when males become sexually mature they were forced out of their natal family units [80,82]. In contrast, more recent studies imply that male independence is a gradual process which could take up to eight years and not enforced by other members of the family unit [84]. Males with no peers in the family unit tend to complete the transition more quickly and become independent [89].

The means of knowledge transfer between animals is a fact affecting to their social structure [97]. In [98] it claims that there is a positive correlation between animal bonding strength and information transfer. Strong ties between old and young females could affect the reproductive success through knowledge sharing and supporting each other in caring offspring. In African elephants, family units led by elderly matriarch has demonstrated greater per capita reproductive success as these females are capable of distinguishing known from unknown male elephants and assessing risk accurately [99]. Thus, a strong association between elderly and younger females is proven to be beneficial in conflictive situations. Alloparenting can be observed in societies of elephants and other mammals [83, 100, 101]. Alloparental care denotes the care and attention given towards young by the individuals apart from their parents. Being allomother is a beneficial factor for successfully raising the first offspring of a female and it is observed short inter-birth intervals if the alloparental care is available [102]. Allomothering is a vital fact that defines the infant survival in African elephants [80, 103]. After giving birth to the offspring, adult females need to spend 12-18 hours per day to search and consume food to obtain sufficient food for lactation. Therefore, strong bonds between mothers, infants and allomothers are highly beneficial for infant survival and reproductive success [44,69]. Similar allomother caring observation has been made in African and Asian elephant societies [53].

Elephants march through the same travel routes even the environmental conditions change. Elephant marching through a hotel lobby is reported from Zambia after it was built on their migration trail [104]. The traditional travel routes may be communicated from generations to generation and elephants seem dedicated on defending their traditional elephant routes. Recently Sri Lankan government have recognised the significance of this characteristic and imposed rules on freeing elephant corridors [105]. Elephant's behaviour of using the same itinerary is one of the key behavioural features that we utilised in our study.

3.4 Elephant Communication and Call Patterns

Elephants use acoustic signals in different contexts such as mate search [106], male–male competition [107], and maintenance of social bonds etc. [106]. They are able to detect low-frequency calls over a range of several kilometers [108–111]. Further, they use alternative sensory channels such as chemical signals (e.g. smell), and visual and tactile displays all involved in short-distance interactions. It is further suggested seismic energy transmission may be viable channel for elephant communication [112, 113].

The elephant calls could be grouped into four distinctive categories, namely, trumpets, chirps, roars, and rumbles [114]. Juveniles produce three of the four call types, including trumpets, roars, and rumbles, in the context of play and distress. Adults make trumpets and roars in the context of aggression, disturbance, and play. Chirps were generally produced in circumstances of confusion and alarm. Elephants produce rumbles when coordinating family and larger group behaviours, when attracting mates and announcing reproduction and, when competing for resources and/or dominance [106].

Strength of elephant calls varies extensively from very soft calls made between mothers and their nearby infants to the extremely loud calls produced by females announcing their availability. Playback experiments indicated that savannah elephants responded to each others' loud vocalizations over a range of at least two kilometers. For the reason that playbacks were only broadcasted at half the amplitude of the strongest elephant calls in their sample, the researchers concluded that actual range should be at least four kilometres [108]. The advantage that elephant calls can travel several kilometres enables elephant societies to coordinate migrations over large areas.

Females vocalize more often than males; the frequency of vocalization in both males and females is extremely variable and dependent on social circumstances. Researches carried out in Namibia and in the Central African Republic have shown that rate of calling increases predictably with the number of elephants present [111].

3.4.1 Elephant Rumbles

Typically, elephant rumbles are rich in infrasound; however, most of these consist of frequencies high enough to be audible to humans. Infrasound is sound below the level of human hearing. Basically, the infrasonic frequency range is considered to be between 1 Hz to 20 Hz. The lower the frequency, longer the sound wave is and therefore they can travel farther without being absorbed or reflected by the environment. The structure of elephant rumbles is quite varied, but readily recognizable. The elephant calls have roughly eyebrow-shaped time-frequency response as shown in Fig. 3.1. Typically, the rumbles are in the frequencies between 10-250 Hz and last between 2-10 seconds. The other infrasonic noises present in the environment such as the broadband wind or thunder often could obscure the elephant's rumble calls. Figures 3.2, 3.3 depict spectrograms of same elephant rumble call while signal is corrupted by the noise. Additive white Gaussian noise (AWGN) is used for the illustration. Time domain representation of elephant calls are shown in Figs. 3.4-3.6.



Figure 3.1: A spectrogram of a recorded elephant rumble call. Eyebrow-shape structure indicates the harmonics present in the elephant in the signal.

From the above analysis, it is evident that the pattern of elephant rumbles can be used as a tool to recognize the presence of an elephant.



Figure 3.2: A spectrogram of an elephant rumble call while signal is corrupted by noise (AWGN with SNR = 10 dB).



Figure 3.3: A spectrogram of an elephant rumble call while signal is corrupted by noise (AWGN with SNR = -8 dB).



Figure 3.4: Time domain representation of a recorded elephant rumble call. Note that the recorded signal contains minimal background noise. However, detection will be much more challenging in the presence of environmental noise



Figure 3.5: Time domain representation of elephant rumble call while the signal is corrupted by noise (AWGN with SNR = 10 dB).



Figure 3.6: Time domain representation of elephant rumble call while signal is corrupted by noise (AWGN with SNR = -8 dB).

3.4.2 Frequency Comparison with Other Animal Calls



Figure 3.7: The frequency ranges of the emphasized frequencies of vocalization in a large range of land–dwelling animals, plotted as a function of the mass of the animal [115].

The frequency of an animal call is determined simply by the physical properties of the sound-producing mechanism. In [115] it is proposed a model for this phenomenon and suggested that vocalization frequency f should be inversely proportional to the linear size of the animal or, equivalently, that $f \propto M^{-1/3}$, where M is the mass of the animal. This hypothesis is quite agreed with the observations as depicted in Fig. 3.7. As shown in the figure, elephant infrasonic vocalizations are less susceptible to interfere with other animal calls.

Being the largest animal on the land, elephants are having significantly higher mass than the other animals, hence having a natural ability to make low-frequency calls. Combine the above fact with their distinctive frequency structure, elephant rumbles can be uniquely detected in a forest environment even under the presence of other animal calls.

3.5 Chapter Summary

Ascertaining the animal behaviour is important in any conservation project. In this chapter we described the social behaviour and structure of the elephant society. We highlighted the factors effecting to elephant group size, there marching patterns and other behaviours which are favourable in mitigating HEC thorough our approach. Then we summarised the call patterns of these highly social animals and analysed the low frequency elephant call named "rumbles" in time and frequency domain. Finally, the chapter provides frequency comparison of elephant calls with other animal calls and discuss the interference on rumble signals from other animal sounds.

Chapter 4 Elephant Localization System Framework

The approach we propose to mitigate HEC require localization of elephants through a passive acoustic sensor network. In this chapter we brief the underlying mathematical representation of the elephant localization system framework. First, we summarise the passive acoustic localization approaches in terms of Time of arrival (TOA) and time difference of arrival (TDOA) techniques and mathematically model elephant localization problem. Finally, statistical bounds on elephant localization error estimate are derived and presented.

4.1 Introduction

I N cases where a known or easily distinguishable signature is embedded in the signal, the sensor can correlate the signal with the signature to accurately estimate time of arrival. The timing estimation accuracy relies on the signature as well as the sensor capability. This is meaningful only if the sensor is synchronized either with the emitting node or another receiving node. In the former case, which computes time of arrival (TOA) measurements, an absolute distance can be computed. In the latter case where it measures the time difference of arrival (TDOA) a relative distance follows. Geometrically, these two cases constrain the position to be on a circle or on a hyperbolic function, respectively [116]. In both the cases, it is assumed that the sensor nodes are stationary with known positions.

The approach proposed in this thesis to mitigate HEC is by locating the elephants incorporating a sensor network. In ideal conditions, if the synchronized sensors identify the rumbles signals with a timing stamp, trilateration or triangulation methodology can be utilizing to estimate the location of the elephant. Section 4.2 explains the different timing observations utilize in localization strategies and comparable measurement model for our application is introduced. In Section 4.3 two different estimation models for Localisation System are derived and presented.

4.2 Timing Observations and Localization

4.2.1 TOA Measurements

Each receiver measures the arrival time of a transmitted signal from an unknown position, using accurate and synchronized clocks. If the transmitter is also synchronized, the signal propagation time can be computed, which leads to a TOA measurement. Propagation time corresponds to a distance, which leads to the distance circles around each receiver and the intersection point gives the target position.

4.2.2 TDOA Measurements

If the transmitter is unsynchronized, each pair of receivers can compute a time-difference of arrival TDOA. Each receiver pair computes separate hyperbolic function for the distance in terms of TDOA and again the intersection point results the target position.

Elephant localization is exactly identical to the hypothesis described here Section 4.2.2. The vocalization made by the elephant can be considered as distinguishable signature and the elephant call can be treated as unsynchronized transmitter. Therefore, by using TDOA measurements it is possible to localize the elephants.

4.3 Mathematical Models for Elephant Localisation System

The approach of elephant localisation by TDOA measurements can be represented mathematically utilizing estimation models [117, 118]. Here we present two different estimation models to represent elephant localisation problem.

4.3.1 Best Linear Unbiased Estimator (BLUE) Model

Let the localization sensor network has *N* sensors that each receives the signal of interest. It is assumed that the sensors are deployed at known locations and each has identical capabilities. Each sensor is Omni-directional and can detect and receive the interested signals. Further, it is assumed that all the sensors are synchronized perfectly and have accurate timing.



Figure 4.1: Source localization using *N* number of sensors. The nominal source position (x_n, y_n) assumed to be known. It is necessary to determine actual source position (x_s, y_s) .

Let the time of arrival measurements for each sensor to be t_i for i = 0, 1, ..., N - 1. It is necessary to locate the source position (x_s, y_s) . The arrival time measurements can be modelled by

$$t_i = T_0 + \frac{R_i}{c} + e_i$$
 $i = 0, 1, ..., N - 1$ (4.1)

where T_0 is the signal emitted time by the source, *c* is the sound propagation speed, e_i is the uncorrelated measurement noise with zero mean and σ^2 variance. Let the known position of *i*th sensors to be (x_i, y_i) . The range difference between the source and the *i*th sensor is given by

$$R_i = \sqrt{(x_s - x_i)^2 + (y_s - y_i)^2}$$
(4.2)

Substituting (5.3) in (5.2), the model will become nonlinear with respect to the unknown parameters x_s and y_s , and the problem will then become difficult to solve. Therefore, following the standard approach, it is assumed that nominal source position (x_n, y_n) is available where nominal range is denoted by R_{n_i} . In fact, this nominal position which is close to the actual source position could have been obtained from previous measurement, especially in a case of target is being tracked. Now, if it is estimated $\theta = [(x_s - x_n)(y_s - y_n)]^T = [\delta x_s \delta y_s]^T$, then the new source position can be determined. By using a first-order Taylor series expansion of R_i in (5.2) about the nominal position $x_s = x_n$, $y_s = y_n$, the range can be written as

$$R_i \approx R_{n_i} + \frac{x_n - x_i}{R_{n_i}} \delta x_s + \frac{y_n - y_i}{R_{n_i}} \delta y_s$$
(4.3)

By substituting (4.3) in (5.2)

$$t_i = T_0 + \frac{R_{n_i}}{c} + \frac{x_n - x_i}{R_{n_i}c}\delta x_s + \frac{y_n - y_i}{R_{n_i}c}\delta y_s + e_i$$

we obtain linear model for the unknown parameters δx_s and δy_s . For convenience, as in the the Fig. 4.1 we represent,

$$\frac{x_n - x_i}{R_{n_i}} = \cos \alpha_i, \tag{4.4}$$

$$\frac{y_n - y_i}{R_{n_i}} = \sin\alpha_i,\tag{4.5}$$

and the model simplifies to

$$t_i = T_0 + \frac{R_{n_i}}{c} + \frac{\cos\alpha_i}{c}\delta x_s + \frac{\sin\alpha_i}{c}\delta y_s + e_i$$

By incorporating the term R_{n_i}/c into the measurement by letting

$$\tau_i = t_i - \frac{R_{n_i}}{c}$$

the linear model finally becomes to

$$\tau_i = T_0 + \frac{\cos\alpha_i}{c}\delta x_s + \frac{\sin\alpha_i}{c}\delta y_s + e_i$$
(4.6)

However, the signal emitted time T_0 is not known in the case of our application. Therefore, it is incorporating TDOA measurements for the model. The TDOAs are generated as

$$\begin{array}{rcl} \xi_{1} & = & \tau_{1} - \tau_{0} \\ \\ \xi_{2} & = & \tau_{2} - \tau_{1} \\ \\ \vdots \\ \\ \\ \xi_{N-1} & = & \tau_{N-1} - \tau_{N-2} \end{array}$$

Then from (4.6) the final linear model can be denoted by

$$\xi_i = \frac{1}{c}(\cos\alpha_i - \cos\alpha_{i-1})\delta x_s + \frac{1}{c}(\sin\alpha_i - \sin\alpha_{i-1})\delta y_s + e_i - e_{i-1}$$

for i = 0, 1, ..., N - 1. Therefore our model now is in the form of

$$x = H\theta + w \tag{4.7}$$

where

$$\boldsymbol{\theta} = \begin{bmatrix} \delta x_s & \delta y_s \end{bmatrix}^T$$

$$\mathbf{H} = \frac{1}{c} \begin{bmatrix} \cos\alpha_1 - \cos\alpha_0 & \sin\alpha_1 - \sin\alpha_0 \\ \cos\alpha_2 - \cos\alpha_1 & \sin\alpha_2 - \sin\alpha_1 \\ \vdots & \vdots \\ \cos\alpha_{N-1} - \cos\alpha_{N-2} & \sin\alpha_{N-1} - \sin\alpha_{N-2} \end{bmatrix}$$

$$\mathbf{w} = \begin{bmatrix} e_1 - e_0 \\ e_2 - e_1 \\ \vdots \\ e_{N-1} - e_{N-2} \end{bmatrix}$$

Further, the noise vector ${\bf w}$ can be represented as

 $\mathbf{w} = \mathbf{A}\mathbf{e}$

where

and

$$\mathbf{e} = \begin{bmatrix} e_0 \\ e_1 \\ \vdots \\ e_{N-1} \end{bmatrix}.$$

Since the covariance matrix of **e** is given by $\sigma^2 \mathbf{I}$ the covariance matrix **C** of the noise vector **w** is given by

$$\mathbf{C} = E[\mathbf{A}\mathbf{e}\mathbf{e}^T\mathbf{A}^T] = \sigma^2\mathbf{A}\mathbf{A}^T$$

Now by applying Gauss-Makov theorem for the above linear model the best linear unbiased estimator (BLUE) for the source position can be given by

$$\hat{\boldsymbol{\theta}} = (\mathbf{H}^T \mathbf{C}^{-1} \mathbf{H})^{-1} \mathbf{H}^T \mathbf{C}^{-1} \boldsymbol{\xi}$$
(4.8)

$$= \left[\mathbf{H}^{T}(\mathbf{A}\mathbf{A}^{T})^{-1}\mathbf{H}\right]^{-1}\mathbf{H}^{T}(\mathbf{A}\mathbf{A}^{T})^{-1}\boldsymbol{\xi}$$
(4.9)

The minimum variance is given by

$$var(\hat{\boldsymbol{\theta}}_{i}) = \sigma^{2} \left[\left\{ \mathbf{H}^{T} (\mathbf{A}\mathbf{A}^{T})^{-1} \mathbf{H} \right\}^{-1} \right]_{ii}$$
(4.10)

In addition, the covariance matrix of above estimator $\hat{\theta}$ is given by

$$\boldsymbol{C}_{\hat{\boldsymbol{\theta}}} = \sigma^2 \left[\mathbf{H}^T (\mathbf{A} \mathbf{A}^T)^{-1} \mathbf{H} \right]^{-1}$$

4.3.2 Performance Evaluation on Loclaization using BLUE Model

The localization methodology discussed in Section 4.3.1 is simulated using Matlab 7.9.0 (R2009b). Figure 4.2 shows an example sensor and source configuration and the true and estimated positions of the source. Initially, sensors are deployed randomly in an area of $5000 \times 5000 m^2$ and the actual source also located inside the same area. It is assumed that each sensor has line of sight propagation path between the source and the sensors. The algorithm needs an initial source position guess to estimate the true source position from TDOA measurements. However, the system discussed above use passive localization and therefore the initial position of the source is unknown. Therefore, the arithmetic mean (\bar{x}, \bar{y}) between sensors is used as the initial position guess $[x_{es_0} y_{es_0}]^T$. Then, the distance from *i*th sensor to derived source location is calculated by

$$R_{n_{ik}} = \sqrt{(x_{es_k} - x_i)^2 + (y_{es_k} - y_i)^2}$$

where $[x_{es_k} y_{es_k}]^T$ is the estimated source location at *k*th iteration. This value is utilized in calculating $cos\alpha_i$ and $sin\alpha_i$ terms in (4.4) and (4.5) to construct the matrix H in (4.7). Variance of the uncorrelated measurement noise (σ^2) is chosen as 0.1, assuming significant noise is present at the receiver measurements. The δx_s and δy_s are calculated as per (4.9) and the new source position for next iteration is then determined as

$$[x_{es_{k+1}} y_{es_{k+1}}]^T = [(x_{es_k} + \delta x_s) (y_{es_k} + \delta y_s)]^T.$$

This estimated location is utilized as the source location for the next iteration. Likewise, the algorithm iteratively converges into the actual source location. By comparing minimum variance given by (4.10), it can be determined whether the algorithm has converged to a solution or more iterations of the estimator are needed. Figure 4.2 depicts how the location estimator derived above converged into a solution within three iterations.



Figure 4.2: Source position estimation using algorithm described in Section 4.3.1. TDOA measurements from five sensors are utilized. The estimated source position in each iteration is depicted.

The effect of sensor positioning for the overall performance is further analyzed through simulations. Figures 4.3 - 4.5 depicts the performance of the algorithm for different sen-



Figure 4.3: Sensors were deployed in a single line. Number of sensors were changed from 3 to 40. For each trial locations of the sensors were changed 100 times randomly. Note that the successive estimations of source always converge before 10 iterations.



separated by 300 m below the source

(b) Performance profile

Figure 4.4: Sensors were deployed in two lines below the source. Number of sensors were changed from 3 to 40. For each trial locations of the sensors were changed 100 times randomly. Note that the successive estimations of source always converge before 10 iterations.

sor configurations. Note that if the number of iterations exceed 100, the algorithm exits from estimating source position, deciding that it could not locate the source position for that particular sensor configuration. In typical WSN TDOA localization system if the sensors are very close to each other, then one of the sensors is virtually redundant and it provides little additional information. The result could be significant positioning error. The effect of the redundant sensors is significant while the number of sensors is low. This fact is represented in the algorithm by unsuccessful localizations.

However, it should be noted that the positioning of the sensors play a major role on the performance of the localization algorithm. For instance, the success rate is much



Figure 4.5: Sensors were deployed in two lines above and below the source. Number of sensors were changed from 3 to 40. For each trial locations of the sensors were changed 100 times randomly

significant when the sensors are deployed around the source position rather than positioning them in a single side from the source. This phenomenon can be identified by examining the success rates of Fig. 4.5b with respect to Fig. 4.3b and 4.4b.

The Fig. 4.6 depicts the effect of increasing the number of sensors with respect to the successive convergence of the algorithm. Note that after the number of sensors increased to a certain value, there is no significant improvement in the probability of success in estimating the source position in lower iterations. In this particular case, this number of sensors can roughly be identified as seven. In some situations, having a higher number of sensors could be a disadvantage in the presence of noise, because each sensor can contribute for some amount of positioning error. Therefore, the optimal number of sensors needed to cover a particular area has to be estimated while maximizing the range of detection and minimizing the localization error.

4.3.3 Maximum Likelihood Estimator (MLE) Model

As mentioned in Section 4.3.1 Arrival time measurements can be modelled as

$$au_i = T_0 + rac{\parallel Z_S - Z_i \parallel}{c} + w_i$$
 $i = 0, 1, ..., N - 1$



Figure 4.6: The effect of number of sensors over the probability of success within three iterations. (The result was obtained from 10000 executions)

where $w_i \sim N(0, \sigma^2)$.

The TDOA meausrement between i^{th} and j^{th} sensor is given by

$$\tau_{ij} = \tau_i - \tau_j = \frac{1}{c} \left[\| Z_S - Z_i \| \right] - \frac{1}{c} \left[\| Z_S - Z_j \| \right] + w_i - w_j$$

where $|| Z_S - Z_i ||$ is range difference between source and the *i*th sensor. Therefore, with *N* number of sensors we get *N* – 1 independent TDOA measurements.

We consider N - 1 independent measurements with respect to 1^{st} sensor. Then the system of measurements can be written as

$$\underbrace{ \begin{bmatrix} \tau_{12} \\ \tau_{13} \\ \vdots \\ \tau_{1N} \end{bmatrix} }_{\Delta \tau} = \underbrace{ \begin{bmatrix} 1 & -1 & 0 & \cdots & 0 \\ 1 & 0 & -1 & \cdots & 0 \\ \vdots & \vdots & \ddots & \cdots & \vdots \\ 1 & 0 & \cdots & -1 \end{bmatrix} \begin{bmatrix} \| Z_S - Z_1 \| \\ \| Z_S - Z_2 \| \\ \vdots \\ \| Z_S - Z_N \| \end{bmatrix} }_{\zeta_{Zs}} + \underbrace{ \begin{bmatrix} 1 & -1 & 0 & \cdots & 0 \\ 1 & 0 & -1 & \cdots & 0 \\ \vdots & \vdots & \ddots & \cdots & \vdots \\ 1 & 0 & \cdots & -1 \end{bmatrix} \begin{bmatrix} w_{12} \\ w_{13} \\ \vdots \\ w_{1N} \end{bmatrix} }_{\mathbf{W}_{1N}}$$

where $Z_S = (x_S, y_S)$ is the source location and the $Z_i = (x_i, y_i)$; i = 1, 2, ..., N are the sensor locations.

Therefore the probability distribution of MLE can be written as

$$\Delta \tau \sim N(\mathbf{A}\zeta_{Zs}, \sigma^2 \mathbf{A} \mathbf{A}^T)$$

where explicitly given by

$$p\left(\frac{\Delta\tau}{Z_{S}}\right) = \frac{1}{\sqrt{2\pi} \mid \sigma^{2}\mathbf{A}\mathbf{A}^{T} \mid^{\frac{1}{2}}} e^{-\frac{1}{2}(\Delta\boldsymbol{\theta} - \boldsymbol{A}\boldsymbol{\zeta}_{\mathbf{Z}s})^{T} \left(\sigma^{2}\mathbf{A}\mathbf{A}^{T}\right)^{-1}(\Delta\tau - \boldsymbol{A}\boldsymbol{\zeta}_{\mathbf{Z}s})}$$

From [118], for any unbiased estimator $\hat{\theta}$ of the parameter θ the FIM elements are given by

$$[J(\theta)]_{k,l} = E\left\{\frac{\partial logf(y)}{\partial \theta_k}\frac{\partial logf(y)}{\partial \theta_l}\right\}$$

where $1 \le k, l \le M$.

For a Gaussian time series, the FIM is given by

$$[J(\theta)]_{k,l} = \frac{1}{2} tr \left\{ R^{-1}(\theta) \frac{\partial \mathbf{R}(\theta)}{\partial \theta_k} R^{-1}(\theta) \frac{\partial \mathbf{R}(\theta)}{\partial \theta_l} \right\} + \left[\frac{\partial \mathbf{m}(\theta)}{\partial \theta_k} \right]^T R^{-1}(\theta) \left[\frac{\partial \mathbf{m}(\theta)}{\partial \theta_l} \right]$$

In the localization problem that we are considering, the covariance matrix does not depend on Z_s . Therefore,

$$[J(Z_S)]_{k,l} = \left[\frac{\partial A\zeta_{ZS}}{\partial Z_S}\right]^T \left(\sigma^2 A A^T\right)^{-1} \left[\frac{\partial A\zeta_{ZS}}{\partial Z_S}\right]$$

Note that Z_s is bivariate with $x_{S,}y_{S}$. Therefore, let us compute the elements of FIM separately. Consider partial derivative of single element of ζ_{Zs} with respect to x_S

$$\frac{\partial \parallel Z_S - Z_i \parallel}{\partial x_S} = \frac{\partial \sqrt{(x_s - x_i)^2 + (y_s - y)^2}}{\partial x_S} = \frac{(x_s - x_i)}{\parallel Z_S - Z_i \parallel}.$$

Similarly

$$\frac{\partial \parallel Z_S - Z_i \parallel}{\partial y_S} = \frac{\partial \sqrt{(x_s - x_i)^2 + (y_s - y)^2}}{\partial y_S} = \frac{(y_s - y_i)}{\parallel Z_S - Z_i \parallel}$$

Therefore, elements of FIM can be derived as

$$[J(Z_S)]_{x,x} = \begin{bmatrix} \frac{(x_s - x_1)}{\|Z_S - Z_1\|} \\ \frac{(x_s - x_2)}{\|Z_S - Z_2\|} \\ \vdots \\ \frac{(x_s - x_N)}{\|Z_S - Z_N\|} \end{bmatrix}^T \mathbf{A}^T \left(\sigma^2 \mathbf{A} \mathbf{A}^T\right)^{-1} \mathbf{A} \begin{bmatrix} \frac{(x_s - x_1)}{\|Z_S - Z_1\|} \\ \frac{(x_s - x_2)}{\|Z_S - Z_2\|} \\ \vdots \\ \frac{(x_s - x_N)}{\|Z_S - Z_N\|} \end{bmatrix}$$

$$[J(Z_S)]_{y,y} = \begin{bmatrix} \frac{(y_s - y_1)}{\|Z_S - Z_1\|} \\ \frac{(y_s - y_2)}{\|Z_S - Z_2\|} \\ \vdots \\ \frac{(y_s - y_N)}{\|Z_S - Z_N\|} \end{bmatrix}^T \mathbf{A}^T \left(\sigma^2 \mathbf{A} \mathbf{A}^T\right)^{-1} \mathbf{A} \begin{bmatrix} \frac{(y_s - y_1)}{\|Z_S - Z_1\|} \\ \frac{(y_s - y_2)}{\|Z_S - Z_2\|} \\ \vdots \\ \frac{(y_s - y_N)}{\|Z_S - Z_N\|} \end{bmatrix}$$

$$[J(Z_S)]_{x,y} = [J(Z_S)]_{y,x} = \begin{bmatrix} \frac{(x_s - x_1)}{\|Z_S - Z_1\|} \\ \frac{(x_s - x_2)}{\|Z_S - Z_2\|} \\ \vdots \\ \frac{(x_s - x_N)}{\|Z_S - Z_N\|} \end{bmatrix}^T \mathbf{A}^T \left(\sigma^2 \mathbf{A} \mathbf{A}^T\right)^{-1} \mathbf{A} \begin{bmatrix} \frac{(y_s - y_1)}{\|Z_S - Z_1\|} \\ \frac{(y_s - y_2)}{\|Z_S - Z_2\|} \\ \vdots \\ \frac{(y_s - y_N)}{\|Z_S - Z_N\|} \end{bmatrix}$$

Further, it can be derived that

$$A^{T} \left(\sigma^{2} A A^{T}\right)^{-1} A = \frac{1}{N\sigma^{2}} \begin{bmatrix} N-1 & -1 & -1 & \cdots & -1 \\ -1 & N-1 & -1 & \cdots & -1 \\ \vdots & \vdots & \ddots & \cdots & \vdots \\ -1 & -1 & \cdots & \cdots & N-1 \end{bmatrix}_{N \times N} = \frac{1}{N\sigma^{2}} \left(N\mathbf{I} - \mathbf{1}\right)$$

Finally, with the derivation of the above elements FIM for our problem can be represented by

$$[\mathbf{J}(Z_S)] = \begin{bmatrix} [J(Z_S)]_{x,x} & [J(Z_S)]_{x,y} \\ [J(Z_S)]_{y,x} & [J(Z_S)]_{y,y} \end{bmatrix}$$

Therefore, the CRLB can be found by

$$var(\hat{Z})_i \geq [\mathbf{J}(Z_S)]_{ii}$$



(b) Uncertainties in measurement is illustrated by the Covariance Ellipse (zoomed in)

Figure 4.7: Confidence assessment on localisation estimation. Uncertainty in estimation for arbitrary source location (450, 800) when $\sigma = 0.2$ is illustrated by the Covariance error Ellipse.

4.4 Chapter Summary

In this chapter, we described the approach on elephant localization using timing observation measurements in a sensor network. In addition, we derived and presented mathematical models for the elephant localization system framework with two different estimation models. We evaluated the performance of the presented mathematical model implementing a simulation framework. Finally, we illustrated the statistical bounds for the localization error estimate.

Chapter 5 Propagation Constraints in Elephant Localization

In this chapter, we expand elephant localization system framework to introduce real-world input model constraints. We analyze the effect of various meteorological parameters in elephant vocalization propagation and their consequential effects on elephant positioning accuracy. In addition, we present the effect of signal-to-noise ratio (SNR) on detection of vocalization signal which consequently affects the localization accuracy.

5.1 Introduction

PASSIVE acoustic localization systems depend on the signal transmission through the propagation medium. Therefore, changes in the properties of the medium could significantly affect the detection and the localization accuracy. Positioning estimation in these systems is computed based on speed and sound in the medium and the timing measurements. If the received signals are processed in a centralized location or if the sensors are synchronized, the timing measurement error can always be controlled within acceptable margins. However, the speed of sound in the medium is an external factor and could vary extensively with the properties of the medium. In our application of elephant localization utilizing acoustic sensor network, the signal for the sensors is elephant rumbles and propagation medium for the signal is air. It is important to examine the sound signal variation through the air and its effect on localization error. Therefore, in this chapter of the thesis, we quantitatively analyse the effects of various environmental parameters on the positioning accuracy of the system. In Section 5.2 we discuss about the sound propagation in air as the basis for understanding the environmental effects. Section 5.3 we revisit the localization algorithm that utilized in this framework. In Section 5.4 we explain the simulation setup and Section 5.5 outlines results of the analysis and discuss it, further.

5.2 Sound Propagation in Air

The accuracy of elephant localization and detection significantly depends on the characteristics of sound propagation in the air. The sound propagates through the air as longitudinal waves and the propagation is affected by the properties of the medium. For instance, the variations in the metrological parameters such as temperature, wind, relative humidity ect. results in the sound propagation variations in terms of intensity and speed. In this chapter we basically focus on temperature and wind effects on the speed of sound.

5.2.1 Sound Propagation Model

Elephant vocalization generation can be assumed as a point acoustic source and if the medium of air is assumed to be homogeneous, the sound propagation is considered to be spherical. Therefore, the sound intensity in a certain direction is inversely proportional to the expanding surface area of the sphere. The sound pressure level (SPL) at distance *r* is given by $SPL = SWL_{point} - 10log(4\pi r^2)$ dB where SWL_{point} represent the sound pressure level measured at one meter from the source. The model can be simplified to $SPL = SWL_{point} - 20log(r) - 11$ dB which is often referred to as the standard inverse square law for point source sound propagation [119]. In summary, the relative sound intensity is reduced by 6 dB per doubling of distance. Depletion in sound intensity reduces the SNR levels at the sensors. Consequently, signals of interest which concealed in the background noise become difficult to detect.

5.2.2 Temperature Effect

The speed of sound in air changes with respect to variations in several phenomena such as temperature, water vapour mole fraction, pressure and CO_2 content in the air. Cramer [120] has proposed a following closed form equation for the relationship between speed of sound and the above environmental factors.

$$v_{0}(t, p, x_{w}, x_{c}) = a_{0} + a_{1}t + a_{2}t^{2} + (a_{3} + a_{4}t + a_{5}t^{2})x_{w}$$

$$+ (a_{6} + a_{7}t + a_{8}t^{2})p + (a_{9} + a_{10}t + a_{11}t^{2})x_{c}$$

$$+ a_{12}x_{w}^{2} + a_{13}p^{2} + a_{14}x_{c}^{2} + a_{15}x_{w}px_{c},$$
(5.1)

where v_0 is zero-frequency speed of sound, t is the temperature of air in Celcius, x_w is the water vapor mole fraction, x_c is the CO₂ mole fraction, and p is the air pressure. The coefficients a_0 to a_{15} are given in Table 5.1.

However, the speed of sound mainly depends on temperature and effects of other variables are insignificant. Therefore, the speed of sound in air can be expressed in terms of temperature as $C_{Sound_T} = C_{Sound_0} \sqrt{(1 + \frac{T}{273.15})}$ where $C_{Sound_0} = 331.45 \text{ ms}^{-1}$ is sound speed at $0^{0}C$ [121]. In general, per degree Celsius rise in temperature, the speed of sound increases by 0.61 ms⁻¹.

Coefficient	Modified Cramer Values
<i>a</i> ₀	330.8860 m/s
<i>a</i> ₁	0.6324 m/s °C
<i>a</i> ₂	-0.000528 m/s $^{\circ}C^{2}$
<i>a</i> ₃	51.471935 m/s
a_4	0.1495874 m/s °C
<i>a</i> ₅	-0.000782 m/s °C ²
<i>a</i> ₆	-1.82x10-7 m/s Pa
<i>a</i> ₇	3.73x10-8 m/s °CPa
<i>a</i> ₈	-2.93x10-10 m/s °C ² Pa
<i>a</i> 9	-85.20931 m/s
<i>a</i> ₁₀	-0.228525 m/s °C
<i>a</i> ₁₁	5.91x10-5 m/s °C ²
<i>a</i> ₁₂	-2.835149 m/s
<i>a</i> ₁₃	-2.15x10-13 m/s Pa ²
<i>a</i> ₁₄	29.179762 m/s
<i>a</i> ₁₅	0.000486 m/s Pa

Table 5.1: Coefficients of Eq. (5.1) (Carmer [120] equation for speed of sound dependencies in air.)

5.2.3 Wind Effect

Sound propagation medium of air itself is in a motion due to the wind. The sound propagating through the air will be transported by the moving air mass. Therefore, when the sound propagates towards the direction of the wind, the sound speed will decrease. On the otherhand, if the sound propagates along wind direction, the speed of sound will increase. In a situation, wind makes an angle with the sound path it is necessary to take the vector component along the sound path. In our application, sound path is the elephant–sensor path and Fig. 5.1 depicts a situation where wind flows in an angle of $\beta(= |\theta_2 - \theta_1|)$ with respect to sound path. Let V_1 be the speed of sound without wind effect, and V_2 be the wind speed. Then, the resultant sound speed V in the direction of



Figure 5.1: Wind effect on sound speed in the direction of propagation. Effective component of V_2 along the elephant—sensor direction should be incorporated.

source-sensor is given by

$$V = V_1 + V_2 \cos(\beta).$$

In the follwoing section we present the localization algorithm utilized in our framework.

5.3 Localization Algorithm

Once an elephant (source) generates a vocalization, the sound signal propagates through the air following the model discussed in Section 5.2. The sensors at the vicinity of the source receive the attenuated and noise corrupted signal. In our analysis, we assume that all the sensors in our deployment receive the elephant vocalization generated at our interested observation area. Then, we obtain the time difference of arrival (TDOA) information by performing pair—wise cross-correlations between observation signals at each sensor pair.

Let the time of arrival measurements at sensor *i* to be t_i where $i = 0, 1, \dots, N - 1$. The infrasonic acoustic sensors are deployed at pre-determined coordinates (x_i, y_i) . It is necessary to locate the source position (x_s, y_s) . The arrival time measurements can be



Figure 5.2: *N* number of sensors are deployed at the vicinity of the expected elephant location. Here, the distance difference between source and the p^{th} and q^{th} sensor is defined as g_{pq} .

modelled by

$$t_i = T_0 + \frac{R_i}{c_i} + w_i$$
 $i = 0, 1, \cdots, N-1$ (5.2)

where T_0 is the signal emitted time by the source, R_i is the range difference between source and the i^{th} sensor, c_i is the sound propagation speed and w_i is the uncorrelated measurement noise with zero mean and σ^2 variance. It should be noted that the speed of sound between source and each sensor could vary due to the ecological effects discussed in Section 5.2. The distance from source to i^{th} sensor R_i is given by

$$R_i = \sqrt{(x_s - x_i)^2 + (y_s - y_i)^2}.$$
(5.3)

The signal emitted time T_0 in (5.2) is unknown. Therefore, now we incorporate TDOA measurements to the model. Let the TDOA measurement between p^{th} and q^{th} sensor denoted by Δt_{pq} which is given by

$$\Delta t_{pq} = t_p - t_q + w_{pq} \tag{5.4}$$
where $p = 0, 1, \dots, N-2$ and $q = p+1, p+2, \dots, N-1$. The equation (5.4) can be written in terms of (5.3) by

$$\Delta t_{pq} = \frac{R_p}{c_p} - \frac{R_q}{c_q} + w_{pq} \tag{5.5}$$

which can be explicitly characterized by the non linear model

$$\mathbf{Y} = \mathbf{h}(\mathbf{x}) + \mathbf{w} \tag{5.6}$$

where

$$\mathbf{Y} = \begin{bmatrix} \Delta t_{01} \\ \Delta t_{02} \\ \vdots \\ \Delta t_{N-2,N-1} \end{bmatrix}$$
$$\mathbf{h}(\mathbf{x}) = \begin{bmatrix} \frac{R_0}{c_0} - \frac{R_1}{c_1} \\ \frac{R_0}{c_0} - \frac{R_2}{c_2} \\ \vdots \\ \frac{R_{N-2}}{c_{N-2}} - \frac{R_{N-1}}{c_{N-1}} \end{bmatrix}$$
$$\mathbf{w} = \begin{bmatrix} w_{01} \\ w_{02} \\ \vdots \\ w_{N-2,N-1} \end{bmatrix}.$$

Solving non linear equation in (5.6) can be approached with diverse techniques [118] [122]. We appreciate the problem in a geometrical approach as depicted in Fig. 6.1. Lets assume that the $w_{pq} = 0$ and the $c_p = c_q = c$ for $\forall (p,q) = 0, 1, \dots, N-1$. Here, Δt_{pq} corresponds to the distance difference between source and the pair of sensors p and q. Let

$$h_{pq} = c \times \Delta t_{pq} \tag{5.7}$$

and

$$g_{pq} = R_p - R_q \tag{5.8}$$

where both (5.7), (5.8) represent the distance differences. However, g_{pq} consists of the unknown source location. Therefore, the source location can be determined by solving the function defined by

$$f(x_s, y_s) = \underset{x_s, y_s}{\operatorname{argmin}} \sum_{p, q} (h_{pq} - g_{pq})^2.$$
(5.9)

However, the location estimation found by (6.1) does not consider the measurement errors and the sound propagation velocity variances discussed previously. Therefore, by introducing those constraints externally to the localization routine, we can attain the effect of those parameters. The next section of the chapter presents the simulaions of the estimation accuracy that affected by different constraints.

5.4 Simulation Framework

The simulation is involved in two major steps. Former part of the simulation is carried out to implement elephant vocalization signal detection at the sensors. This routine also provides the time delay information between each pair of sensors. In this section, the SNR effects on signal detection and acquiring time delay information are also considered. The latter part of the simulation estimate the source position utilizing the time delay information provided by the previos section. The wind and temperature effect on source position estimation is considered in this section.

5.4.1 Vocalization Detection

In real world scenario, once an elephant made a rumble signal, each sensor in the vicinity of the source location hear time delayed and noise corrupted versions of the original signal. This is simulated by shifting and zero padding the original rumble vocalization data file by relevent time delays for each sensor. Then, the noise corruption is incorporated by adding white Gaussian noise to time shifted signal. Now at the receiver, if the signal is cross–correlated with signature rumble vocalization signal, it is possible to find the peak cross–correlation coefficient at number of samples which represent the original time delay information. We perform multiple cross–correlations between each pair of sensors' observation signals to find the time delay information represented by Δt_{pq} in (5.7).

5.4.2 Sensor Placement and Source Localization

In this part of the simulation several real world aspects are taken into the consideration. We chose 2000 m × 2000 m square as our analysis area which represent the woodland. We assume that coordinate axis y = 0 as our village boundary. Countermeasure can be taken only if the elephants are localized at least before they enter into the 500 m margin from the village boundary. Therefore, we choose interested source observation area as the rectangle bounded by coordinates {(0, 500), (0, 2000), (2000, 500), (2000, 2000)}.

Sensors might need occasional maintenance, for instance, to replace the batteries, to sensor health checks etc. Practically people could not frequently move into deep forest area which is not pleasing in terms of risk and the cost of the visits. Therefore, the sensors need to be placed close to the village boundary. We decide that the sensors should be placed within 200 m distance from the village boundary. Consequently, the sensor deployment area is a rectangle bounded by coordinates $\{(0, 0), (0, 200), (2000, 0), (2000, 200)\}$. However, sensors need to be placed optimally so that they avoid the redundancy and provide maximum accuracy. We, considered several options, such as placing sensors randomly, homogeneously, in a line and in a curvature. In the line deployment, we selected centre line of the chosen sensor deployment strip (y = 100 m) and we deployed sensors in equidistance separations. In curvature deployment, the proper arc is found as follows. First, we drew a circle centred at (1000, 2000) and radius 2000 m. Then we select the arc which is laid inside the sensor placement strip. Then the sensors are placed such that they are separated from each other by equal arc distances. In our simulations, we used four sensors which have identical capabilities. It is identified, by means of several simulation experiments that the curvature sensor placement outperforms other methods

in source localization. Therefore, it is chosen as the sensor placement criteria for the results presented in this work.

The effect of each constraint discussed in Section 5.2 is attained by placing the source at different positions in the interested observation area and calculating the estimation error. Initially, the rectangular observation area is divided into $100 \text{ m} \times 100 \text{ m}$ small girds. Then, the source is placed 100 times randomly inside the small grid and the source position is estimated according to the algorithm discussed in Section 5.3. However, each of position estimation obtained here is affected by different parameters such as wind, temperature and SNR and thus they are erroneous. We compute the error for each random source position by calculating the Euclidean distance between actual and the estimated source placements. The computed mean error is considered as the average inaccuracy in positioning when the source is located at the area represented by that particular grid. The above process is repeated for each grid in the observation area. The effects of different meteorological parameters and SNR are presented in the subsequent section.

5.5 Results and Discussions

The simulation setup discussed in the Section 5.4 is obtained using Matlab 7.9.0 (R2009b) development environment for the analysis. The effect of individual parameter is analyzed via simulations while keeping the other parameters constant. Figures 5.3-5.5 show the temperature effects on the source positioning. Initially, it is assumed that the temperature is uniform throughout the area. Results for the instances of temperature 22° C and 30° C are depicted in Figs. 5.3, 5.4, respectively. The maximum mean estimation errors in these incidents are less than one meter throughout the observation grid. Therefore, the increase in the temperature equally, does not affect the source positioning accuracy significantly. Then, we varied the temperatures at each node such that $T_{S1} = 25^{\circ}$ C, $T_{S2} = 23^{\circ}$ C, $T_{S3} = 22^{\circ}$ C, $T_{S4} = 24^{\circ}$ C, and execute the simulation again. Figure 5.5 shows that in this scenario the positioning error increases, drastically.

The wind has a very significant effect on localization accuracy as depicted in Fig.



Figure 5.3: The source localization error variation at different source positions within the observation area. Four sensors are deployed in curvature with equal arc length separations as discussed in Section 6.4. The $\mathbf{T} = \mathbf{22} \ ^{0}\mathbf{C}$ uniform throughout the area. $V_{2} = 0 \ \mathrm{ms}^{-1}$, $\theta_{2} = 270^{0}$, SNR = 10 dB



Figure 5.4: The source localization error variation at different source positions within the observation area. Four sensors are deployed in curvature with equal arc length separations as discussed in Section 6.4. The $\mathbf{T} = \mathbf{30} \ ^{0}\mathbf{C}$ uniform throughout the area. $V_{2} = 0 \ \mathrm{ms}^{-1}$, $\theta_{2} = 270^{0}$, SNR = 10 dB



Figure 5.5: The source localization error variation at different source positions within the observation area. The sensors are deployed in curvature with equal arc length separations as discussed in Section 6.4. The Temperature varies between each sensor path. $T_{S1} = 25 \ ^{0}C$, $T_{S2} = 23 \ ^{0}C$, $T_{S3} = 22 \ T_{S4} = 24 \ ^{0}C$. $V_{2} = 0 \ ms^{-1}$, $\theta_{2} = 270^{0}$, $SNR = 10 \ dB$



Figure 5.6: The source localization error variation at different source positions within the observation area for wind velocities $V_2 = 5 \text{ ms}^{-1}$, $V_2 = 10 \text{ ms}^{-1}$, $V_2 = 15 \text{ ms}^{-1}$, $V_2 = 20 \text{ ms}^{-1}$. Four sensors are deployed in curvature with equal arc length separations as discussed in Section 6.4. The $T = 22 \ ^{0}C$ uniform throughout the area. $\theta_2 = 270^{0}$, SNR = 10 dB.



Figure 5.7: The source localization error variation at different source positions within the observation area. Four sensors are deployed in curvature with equal arc length separations as discussed in Section 6.4. The T = 22 ^{0}C uniform throughout the area. $V_{2} = 0$ ms⁻¹, $\theta_{2} = 270^{0}$, **SNR = -15 dB**.

5.6. The rise in the wind velocity increases error in positioning especially towards the boundaries of *x* axis. In lower wind velocities, perhaps the error margin of \pm 20 m at 500 m might not exceed; however, inaccuracy at higher velocities cannot be tolerated. Faster wind velocities toward the sensors will increase the sound velocity while decreasing the time-of-flight (TOF) of the signal from the source to the sensor. The TOF is higher when the source is far away from the sensors. Therefore, the errors in time delay increase with the distance resulting in higher positioning errors at far away source locations from the sensors.

The SNR has a much more considerable effect on the source localization accuracy in our algorithm. Figure 5.7 depicts the distribution of mean estimation error throughout our observation grid, while the SNR is set to -15 dB. Note that in our analysis we add the noise from a uniform distribution, assuming the noise is equivalent even for far away distances from the sensors. However, this assumption does not follow the actual scenario. According to the discussion in Section 5.2.1 the sound intensity depletes with the distance, resulting in lower SNR levels at the sensors when the source is far away. Therefore, it can be expected higher SNR levels while the elephants are closer to the sensors. The SNR level has substantial effect when the source is placed in the XY corners of the observation area. This results from inaccurate time delay information provided by correlation routine, at lower SNR levels. The inaccuracy is significant once sensors are spatially in closer to the source. These erroneous time delays subsequently cause the minimization algorithm to estimate enormously inaccurate source positions as depicted in Fig. 5.7. The effect was further analyzed by increasing the SNR level in steps. The inaccuracy disappeared at the SNR level of -12 dB. We identified this as the marginal SNR level that is necessary for the system to provide proper localization estimation in this simulation setup.

5.6 Chapter Summary

In this chapter we have analysed major propagation constraints that could affect the elephant localization framework. Our results show that the impact of the temperature for the accuracy of source localization is insignificant if the temperature is uniform over the considered area. In contrast, the temperature variation between each source—sensor path significantly affects the source positioning error. Therefore, in real implementation integrating mechanism to compensate temperature variations will vastly improve the localization accuracy. In addition, it is identified that wind velocity has a very significant effect on positioning error. Therefore, incorporating real-time wind velocity information is also important in this localization system. The SNR levels at the sensors also play major role in localization error. The system provides enormously inaccurate position estimations once the SNR level drops under certain marginal level. More precisely, this level is -12dB with respect to our simulation framework discussed in this chapter. Therefore, in an actual implementation, incorporating signal processing techniques for noise reduction will increase the overall performance of the system.

Chapter 6 Localization Accuracy Enhancement Technique

In this chapter, we propose a novel "probe" technique to estimate the average speed of sound which is affected by the environmental parameters. The novel technique is incorporated to enhance the accuracy of elephant localization system which utilizes an acoustic sensor network. We explain the localization system framework and then we introduce the probe technique for wind and temperature affected sound speed correction. Incorporation of probes in our sensor network eliminates the requirement to explicitly measure temperature and wind velocity for accurate determination of sound velocity. Performance of the system is evaluated against a conventional sensor network equipped with wind and temperature sensors. The performance of the technique is evaluated under different real-world ecological scenarios and results are presented.

6.1 Introduction

E STIMATED position accuracy dependency on accurate signal propagation speed used in the computation algorithm in a passive acoustic sensor network is significant in some applications [123, 124]. In Chapter 5 the possibility of using an acoustic sensor network to detect and localize elephants has been extensively analysed. However, the speed of sound propagation in air, depends on ecological parameters such as temperature and wind velocity. Therefore, wind and temperature variations are identified as the main cause of performance deterioration of such system.

In this chapter, we propose a novel technique to correct the errors arising from variations in sound speeds due to environmental effects. We propose a system framework to implement the algorithms that have been developed and describe the results of experiments, under different environmental profiles.

In Section 6.2 we revisit the localization algorithm that will be utilized in the proposed system. In Section 6.3 the technique used to correct the speed of sound to mitigate temperature and wind velocity effects is described in detail. In Section 6.4 we explain the simulation framework and Section 6.5 outlines performance of the novel technique.

6.2 Localization Algorithm



Figure 6.1: *N* number of sensors are deployed at the vicinity of the expected elephant location. The distance difference between source and the p^{th} and q^{th} sensor is defined as g_{pq} .

We consider an acoustic sensor network with N sensors that listen to elephant vocalizations. In consideration of the long propagation range of infrasonic rumble signals [108], it is assumed that all sensors receive an attenuated and noise corrupted replica of a call generated by an elephant (source). Let the time of arrival measurements at sensor S_i be t_i where $i = 0, 1, \dots, N - 1$. The infrasonic acoustic sensors are deployed at predetermined coordinates (x_i, y_i) . It is necessary to locate the source position (x_s, y_s) . The arrival time measurements at sensor S_i can be modeled by $t_i = T_0 + R_i/c_i + \epsilon_i$ where T_0 is the signal emitted time by the source, R_i is the distance between source and the i^{th} sensor, c_i is the sound propagation speed and ϵ_i is the normally distributed measurement noise with zero mean and σ^2 variance. It should be noted that the c_i between the source and each sensor could vary because of environmental effects such as wind and the temperature. The distance from source to i^{th} sensor R_i is given by $R_i = \sqrt{(x_s - x_i)^2 + (y_s - y_i)^2}$. The time T_0 of signal emission is unknown, and so it is appropriate to perform a time-difference-of-arrival (TDOA) estimation. TDOA information is obtained by performing pair-wise cross-correlations between the observation sound signals at each sensor pair. TDOA measurement between the p^{th} and q^{th} sensors denoted by Δt_{pq} is given by $\Delta t_{pq} = t_p - t_q + \epsilon_{pq} = R_p/c_p - R_q/c_q + \epsilon_{pq}$, where $p = 0, 1, \dots, N-2$ and $q = p + 1, p + 2, \dots, N - 1, \epsilon_{pq} = \epsilon_p - \epsilon_q$.

Let the measured TDOA ranges be

$$\mathbf{m} = [m_{2,1} \quad m_{3,1} \quad \cdots \quad m_{n,1}]^T = \mathbf{d} + \boldsymbol{\epsilon}$$

where

$$\boldsymbol{d}(\boldsymbol{\theta}) = \begin{bmatrix} r_{2,1} & r_{3,1} & \cdots & r_{n,1} \end{bmatrix}$$

and $r_{p,q} = R_p - R_q$. The probability density function [125] of **m** given θ is

$$f(\boldsymbol{m}/\boldsymbol{\theta}) = 2\pi^{-n/2} (\det \mathbf{Q})^{-1/2} \exp\{-J/2\}$$

where

$$J = [\boldsymbol{m} - \boldsymbol{r}(\boldsymbol{\theta})]^T \mathbf{Q}^{-1} [\boldsymbol{m} - \boldsymbol{r}(\boldsymbol{\theta})].$$

The Maximum Likelihood Estimate (MLE) is the $\theta(x_s, y_s)$ that minimizes *J*. This can be reduced to minimizing the non-linear least square problem [126] of

$$f(x_s, y_s) = \arg\min_{x_s, y_s} \sum_{p, q} (h_{pq} - g_{pq})^2.$$
(6.1)

where $h_{pq} = c \times \Delta t_{pq}$ and $g_{pq} = R_p - R_q$. The solution to (6.1) gives the unknown source location. However, to locate the source accurately the above system needs the correct sound propagation speeds between the source and each sensor, which is affected by wind and temperature variations. Our approach to estimate sound speed with mitigated temperature and wind velocity effects is presented in the next section.

6.3 Average Sound Speed Estimation Methodology

Our goal is to estimate the approximate speed of sound that signals propagate with respect to each sensor. In order to build a solution that attains the above sound speeds in a real world environmental scenario we first develop a model assuming uniform environmental conditions. Then the model is extended to real world comparable, non uniform and totally unknown environmental conditions.

To locate the elephants accurately, acoustic sensors and some specific sound (chirp) signal generators are deployed at predetermined location coordinates. These sound generators are referred to as "probes" in the remainder of the chapter. The chirp signal generation time schedule is predefined and known to both devices. Therefore, by listening to the signal, each sensor can determine the time of flight (TOF) from probes to the sensor. Then, the approximate speed of sound between the source and the sensors is estimated in two different models as presented below.

Model A: The temperature over the area is assumed to be uniform but unknown. The wind speed and the direction over the area are assumed to be uniform but unknown.

We estimate the wind information and the uniform temperature over the area completely depending on the probe technique without incorporating additional sensors. Consider that the *i*th sensor (*S_i*) receives the signal from the probes *P_l*, *P_m*, *P_n* as depicted in Model A in the Figure 6.2. The temperature affected speed of sound between *S_i* and *P_z* is *V_{P_zS_i*(*T*) where z = l, m, n, and, so the measured speed of sound *V_{P_zS_i* between *S_i* and *P_z* is given by *V_{P_zS_i* = *V*(*T*) + *V_w* cos($\theta_w - \theta_{P_zS_i}$). Thus, by resolving the system of equations for each probe *z* and sensor *S_i*, it is possible to estimate the approximate speed of sound in the}}}



Figure 6.2: The *i*th sensor (*S*_{*i*}) receiving the signal from the probes *P*_{*l*}, *P*_{*m*}, *P*_{*n*}. The wind flows with speed of *V*_{*w*} with an angle of θ_w with respect to the horizontal axis. The temperature affected speed of sound between *P*_{*z*} is *V*_{*P*_{*z*}*S*_{*i*}(*T*) where *z* = *l*, *m*, *n*.}

area by evaluating three unknowns (wind speed V_w , wind direction θ_w , and temperature affected speed of sound V(T)). Now the wind direction can be derived as

$$\theta_{w} = \tan^{-1} \left[\frac{A \sin(\beta) - B \sin(\alpha)}{A \cos(\beta) - B \cos(\alpha)} \right]$$
(6.2)

where

$$A=(V_{P_lS_i}-V_{P_mS_i})\sin\left((heta_{P_nS_i}- heta_{P_lS_i})/2
ight)$$
 ,

$$B = (V_{P_lS_i} - V_{P_nS_i}) \sin\left(\left(\theta_{P_mS_i} - \theta_{P_lS_i}\right)/2\right),$$

 $\alpha = (\theta_{P_mS_i} + \theta_{P_lS_i})/2$, and $\beta = (\theta_{P_nS_i} + \theta_{P_lS_i})/2$. The wind speed V_w is now

$$V_w = \frac{V_{P_l S_i} - V_{P_m S_i}}{\cos(\theta_w - \theta_{P_l S_i}) - \cos(\theta_w - \theta_{P_m S_i})}.$$
(6.3)

Bu using (6.2) and (6.3) we obtain the temperature affected speed of sound V(T) from

$$V(T) = V_{P_l S_i} - V_w \cos(\theta_w - \theta_{P_l S_i}).$$
(6.4)

The temperature and the wind effects compensated sound velocity can then be determined.

Model B: The temperature over the area is non-uniform and unknown. The wind speed and the direction over the area are also non-uniform and unknown.

Under the condition where wind and temperature are non-uniform and information is unavailable, it is difficult to produce closed form equations to correct the actual sound propagation speed, because not enough information is available. In this case real world spatial properties of the temperature and wind variations were incorporated. The assumption of slow spatial variation of temperature permits temperatures in adjacent regions to be considered as correlated random variables. We model this variation, therefore, as a two-dimensional Gaussian function in the phase of data generation. The temperature at particular location coordinates (x, y) is given by

$$f(x,y) = T_{min} + \Delta T_{Vmax} \exp\left(-\left(\frac{(x-x_o)^2}{2\sigma_x^2} + \frac{(y-y_o)^2}{2\sigma_y^2}\right)\right),$$

where T_{min} is the minimum temperature, ΔT_{Vmax} is the maximum temperature variation, (x_o, y_o) is the centre coordinates of the interested area, σ_x , σ_y are the x and y spreads of the temperature blob over the area. A representation of this temperature variation is depicted Fig. S7 in Section 6.7.1 of the supplemental information. In addition, the wind flow is assumed to be a Markov Random Field (MRF). The wind vector at a particular position is determined by values drawn from a normal distribution $\mathcal{N}(\mathbf{V}_{\mathbf{w}\mathbf{Adj}}, \sigma_w^2)$ where $\mathbf{V}_{\mathbf{w}\mathbf{Adj}}$ is the wind vector in adjacent locations. Representations for wind speed and direction maps under this scenario are depicted Figures S8-S9 in Section 6.7.1 of the supplemental information. Further, the probe technique is incorporated to evaluate the approximated actual sound propagation by dividing the interested area into effective isothermal spatial regions. The validity of the above approximation depends upon the selection of the spatial region. The smallest region to uniquely estimate the environmental parameters is chosen as the isothermal region. This region is the area depicted as Model A in Fig. 6.2. The set of equations derived in the Model A can be used to estimate the average sound speed correction within the effective isothermal areas. Sound signals to distinct sensors will experience different speeds of propagation as affected by wind and temperature variations. By efficiently averaging them considering the spatial proximity, approximation to the effective sound speed can be obtained.

Comparison Model 1: The temperature over the area is non—uniform and unknown. The wind speed and the direction over the area are also non—uniform and unknown. Wind and temperature sensors are incorporated at various locations for the environmental affect modelling. To compensate the environmental effects, area partitioning and sound speed averaging technique was utilized.

Comparison Model 1 comprises a conventional acoustic sensor network. In this model we incorporate temperature and wind sensors in the system and place them in the locations where probes and sensors were placed in the previous models. These wind and temperature sensors provide point measurements. To obtain a better representation of the environmental profile over the area, we partition it into small regions which include at leaset one sound sensor location and three probe locations. Then we average the measurements from temperature and wind sensors incorporated with those partitioned region. Consequently, the speed of sound with respect to each acoustic sensor is determined taken into account the measurements.

Comparison Model 2: The temperature over the area is non-uniform and unknown. The wind speed and the direction over the area are also non-uniform and unknown. Wind and temperature sensors are incorporated at various locations for the environmental affect modelling. Two-dimensional Gaussian function and MRF are used to model the environmental effect.

In Comparison Model 2 we utilize temperature and wind sensors as in Comparison Model 1. In this model instead of averaging the measurements in the regions effectively, we utilize Gaussian temperature model to determine the temperature variation over the area. This Gaussian model is similar to the one that introduced in synthetic data generation above described in Model B. Model parameters for temperature variation can be produced through the measurements at sensor and probe coordinates and subsequently, temperature values at every coordinate can be determined. In addition, the wind velocity vector over the region is also represented as a MRF. The wind speed and direction measurements from the sensors are used to determine the initial mean and the variance of the field. Consequently, this information is incorporated to estimate the wind velocity field values over the area.

6.4 Simulation Framework

To test the proposed methodology in a real elephant localization scenario, a software simulation framework was first developed; and subsequently, a hardware system was implemented. Previously recorded elephant rumble signals were used to replicate the elephants in testing the hardware system along with a sensor network to capture these elephant calls. In the software framework, signal processing and localization mechanisms were implemented. Further, a separate simulation analysis was carried out taking into account real forest environmental scenario for localization error evaluation. Three major steps in the above software simulation are summarized below.

Step 1: Sensors and Probes Placement and Sound Speed Estimation

In this part of the simulation, several real world constraints are taken into consideration. We chose a 2000 m × 2000 m square as our analysis area which represents the wood-lands. The coordinate axis y = 0 is assumed as the village boundary. Once the elephants are detected at a sufficient distance (safety distance) from the village boundaries safety measures can be taken. As the average walking speed of elephant is around 3-6 kmh⁻¹, distance of 500 m was selected as the safety distance. Now the interested elephant observation area became the rectangle bounded by coordinates {(0,500), (0,2000), (2000,500), (2000,2000)}. Sensors need to be placed close to the village boundary for practical reasons, so that the rectangle bounded by coordinates {(0,0), (0,200), (2000,0), (2000,200)} was chosen as the sensor deployment area. In a similar way to [123] sensors are placed at equidistance locations on a segment of circle centered at (1000,2000) with a radius 2000 m. In the simulations 12 sensors which assumed to have identical capabilities were used.

The operation of the probes are limited to generate chirp signals at a predefined time schedule. These probes are assumed more robust than sensors and need less maintenance. Further, to acquire general environmental information in the observation area, probes need to be deployed in or close to the interested observation region. We chose to place them around the lower boundary (y = 500 m) of the observation area. Ten equally separated probes were deployed in a strip, such that every sensor had access to at least three probes at essentially the same range.

Sound signal experiences different sound speeds across the region it travels. For instance, if it is considered an area depicted in Figure 6.3, divided into nine squares, each square can be considered as a microclimate region with a particular wind velocity W_l and temperature T_l where $l = 1, 2, \dots, 9$. In each square, sound speed is affected by the local environmental parameters. For instance, the signal from source *E* to Sensor S_1 will expe-



Figure 6.3: Example observation area with variable sound speeds. Each grid can be considered as a microclimate region with a particular wind velocity W_l and temperature T_l where $l = 1, 2, \dots, 9$. The effective sound speed varies according to the environmental parameters at each grid.

rience the effects of temperatures T_2 , T_4 , T_5 , T_7 and wind velocities V_{W2} , V_{W4} , V_{W5} , V_{W7} along the propagation path. Apparently, the most complex sound speed variation can be observed in the analysis of Model B where temperatures as well as wind velocities vary between each square. A detailed discussion on effects of sound speed variations on the propagation time delays is presented in Section 6.7.2 of the supplemental information.

Step 2: Vocalization Detection and TDOA Estimation

Once an elephant makes a rumble call, every sensor in the vicinity of the source location receives time delayed and noise corrupted versions of the original signal. This scenario is simulated by shifting and zero padding the original rumble vocalization data files by relevant time delays. Then, the noise corruption is incorporated by adding white Gaussian noise to the time shifted signals. Now the signals are cross-correlated with each other at the centralized processing unit to obtain the time delay difference information. These time delays correspond to Δt_{pq} in Section 6.2.

Step 3: Source Localization and Error Validation

The performance of the probe technique is obtained by placing the source at different positions in the observation area and calculating the estimation error. Initially, the rectangular observation area is divided into 100 m \times 100 m small square. Then, the source is placed 100 times randomly inside the small grid and the source position is estimated according to the algorithm discussed in Section 6.2. We compute the error for each random source position by calculating the Euclidean distance between the actual and the estimated source locations. The mean estimation error is calculated for the above 100 source placements. The normalized mean estimation error is calculated as the ratio between the mean estimation error of the grid with respect to the distance to the grid from the village boundary (y = 0). The above process is repeated for each grid in the observation area. The performance of the probe technique under above simulation setup is presented in Section 6.5.

6.5 **Performance Evaluation on Probe Technique**

The performance of the proposed methodology discussed as in Model B was tested against comparison models described in Section 6.3. In all models identical and real world comparable environmental condition was maintained with synthetically generated temperature and wind data. The wind velocity was non-uniform and the relationship was given by the Markov-Random Field properties whilst the temperature distribution was attained through Gaussian distribution as described in Model B.

Figure 6.4 depicts the performance of Comparison Model 1. In this scenario it was assumed that all sensors and probes consist of additional temperature and wind sensors. Exact temperature and wind information is known to the centralized processor at those particular positions. Thus, the sound speed can be attained through effectively averaging these measurements. However, this averaging methodology does not represent exact environmental condition. As a result, actual effect on sound speed cannot be attained through this process and model shows higher normalized error distribution.

Analysis results for normalized error distribution over the area, for Comparison Model



Figure 6.4: Normalized source localization error variation at different source positions within the observation area. At every sensor and probe position, a **wind** and **temperature** sensors were incorporated as in Comparison Model 1 in Section 6.3. The sound speed estimation was done through partitioning the area and effectively averaging the point measurements with respect to each sensor. (i.e. Not through probe technique). Ten probes and 12 sensors are deployed as discussed in Section 6.4. The temperature and wind velocities are non-uniform over the area. (Simulation parameters were chosen as follows: $T \in \{22 - 32\}^{\circ}C$. Initial mean values for wind are $V_w = 20 \text{ ms}^{-1}$, $\theta_w = 120^{0}$. SNR = 10 dB.)

2 is shown in Figure 6.5. In this model, wind and temperature sensors were utilized and the measurements from these sensors were incorporated to determine the environmental model parameters as in synthetic data generation explained in Section 6.3. In the above condition, it is possible to find exact temperature values over the area as we incorporate identical Gaussian function that we utilized in data generation to represent the temperature variation attain through the measurements. However for the wind velocity variation, the measurements only provide initial mean V_{wlnit} and variation σ_w^2 of the field and the exact wind velocity at each grid could be different from the values attained through the model. Therefore, even this model represents the environmental data over the area with more reasonable manner, it does not provide superior performance over the previous comparison model.

Figure 6.6 shows the performance of positioning estimation whilst sound speed eval-



Figure 6.5: Normalized source localization error variation at different source positions within the observation area. At every sensor and probe position, a **wind** and **temperature** sensors were incorporated as in Comparison Model 2 in Section 6.3. Gaussian function was utilized to model the temperature and MRF was used to model the wind velocities over the observation field. The sound speed is then determined considering the above environmental model values. (i.e. Not through probe technique). Ten probes and 12 sensors are deployed as discussed in Section 6.4. The temperature and wind velocities are non-uniform over the area. (Simulation parameters were chosen as follows: $T \in \{22 - 32\}^{\circ}C$. Initial mean values for wind are $V_w = 20 \text{ ms}^{-1}$, $\theta_w = 120^0$. SNR = 10 dB.)

uation was solely done utilizing probe technique as described in Model B. Utilization of the technique has surpassed the positioning accuracy than in above scenarios despite the sound speed estimation is performed with limitations of the ecological data. The normalized mean error estimation is always below the 0.06 margin while the estimation error is just above 20 m in average around y = 500 m border. In an elephant localizing system, the positioning accuracy is highly important only when the elephants are close to the village. Therefore, the above accuracy attained through our inexpensive probe technique is well over the acceptable margin. In addition, this accuracy is achieved without explicitly measuring the wind and temperature over the area. A performance analysis of probe technique with respect to above sensor network scenarios under several different environmental conditions is presented in Section 6.7.4 of supplemental material.



Figure 6.6: Normalized source localization error variation at different source positions within the observation area. No temperature or wind sensors utilized in the system. The sound speed estimation was done solely depending on probe technique as described in Model B in Section 6.3. Ten probes and 12 sensors are deployed as in Section 6.4. Temperature and wind velocities are non-uniform and unknown over the area. (Simulation parameters were chosen as follows: $T \in \{22 - 32\}^\circ C$. Initial mean values for wind are $V_w = 20 \text{ ms}^{-1}$, $\theta_w = 120^0$. SNR = 10 dB.)

6.6 Chapter Summary

In this chapter, we have investigated the feasibility of enhancing the accuracy in elephant localization system by utilizing a sound generating probe technique. The performance of the technique is analysed under different ecological scenarios in a simulation framework. Comparison models have been introduced to evaluate the performance of the technique under different environmental conditions. The results are presented to illustrate the effectiveness in utilizing the probe technique under comparable real-world wind and temperature variations. The technique preserves the normalized mean error estimation always below the 0.06 margin throughout the interested observation area. The positioning accuracy of elephant localization system is more important when the elephants are closer to the village boundaries. The analysis confirms that probe technique can limit the positioning error just above 20 m in average around y = 500 m from the village border.

6.7 Supplementary Material

6.7.1 Environmental Parameter Variation Maps

In the simulation analysis for the data generation purposes temperature variation over the area was represented by two dimensional Gaussian function. The temperature at particular location coordinates (x, y) is given by

$$f(x,y) = T_{min} + \Delta T_{Vmax} \exp\left(-\left(\frac{(x-x_o)^2}{2\sigma_x^2} + \frac{(y-y_o)^2}{2\sigma_y^2}\right)\right),$$

where T_{min} is the minimum temperature, ΔT_{Vmax} is the maximum temperature variation, (x_o, y_o) is the centre coordinates of the interested area, σ_x , σ_y are the *x* and *y* spreads of the temperature blob over the area. Figure S7 depicts the temperature variation map utilized in the simulation analysis.

The wind direction and the speed variation over the area are represented assuming Markov random field property. Subsequently, wind vector at a particular position was determined by values drawn from a normal distribution $\mathcal{N}(\mathbf{V_{wAdj}}, \sigma_w^2)$ where $\mathbf{V_{wAdj}}$ is the wind vector at the adjacent location. Figures S8- S9 represent the wind velocity and the direction variation maps utilised in the simulation analysis.

6.7.2 Source Localization under Sound Speed Variation

The speed of sound in air depends on several environmental variables such as temperature, water vapour mole fraction, pressure and CO₂ content in the air. Cramer [120] has proposed a closed form equation for sound speed in air as a s function of above environmental factors. However, the sound speed mainly depends on temperature variations and effects of other variables are relatively insignificant. The speed of sound in air can be expressed in terms of temperature as

$$C(T) = C(T_0) \sqrt{\left(1 + \frac{T}{273.15}\right)},$$



Figure S7: Temperature variation map over the simulation area. Temperature variation over the area was represented by a two dimensional Gaussian function. Simulation parameters were chosen as follows: $T_{min} = 22^{\circ}$ C, $\Delta T_{Vmax} = 10^{\circ}$ C, $(x_o, y_o) = (1000, 1000)$, and $\sigma_x = 500$, $\sigma_y = 500$.

where $C_{Sound_0} = 331.45 \text{ ms}^{-1}$ is sound speed at 0 ^{0}C [121]. In general, the speed of sound increases by 0.61 ms⁻¹ per degree Celsius rise in temperature. When the sound propagates under wind the propagation medium itself is in a motion. The sound propagating through the air will be transported by the moving air mass. Therefore, resultant sound speed under wind condition *V* in the direction propagation is given by

$$C(T, V_w) = C(T) + C(W)cos(\psi)$$

where C(W) is the wind speed and ψ is the angle between sound path and the wind direction.

We consider an area which is divided into grids similar to the example region depicted in Fig. S10. The temperature and the wind velocities are constant in each grid; however, these two variables vary between grids. Therefore, the sound speeds between each grid vary due to temperature and wind effects. Probes and sensors are deployed in predefined coordinates and source elephant location is unknown. Let us consider that we have *M* number of probes and *N* number of sensors. The area is divided into Q number of grids.



Figure S8: Wind speed variation map over the simulation area. The wind direction and the speed variation over the area are represented assuming Markov random field property. Simulation parameters were chosen as follows: Initial mean values for wind speed is $V_w = 20 \text{ ms}^{-1}$ and $\sigma_{wSpeed} = 0.5$.

Time of arrival measurement from i^{th} sensor to j^{th} probe is given by

$$t_{S_i P_j} = \sum_{k=1}^{Q} \frac{d_{S_i P_j k}}{C_{S_i P_j k}} + \xi_{S_i P_j}$$
(6.5)

where, $d_{S_iP_jk}$ is the distance that sound signal travelled from probe P_j towards sensor S_i in the k^{th} grid which is given by

$$d_{S_i P_j k} = \sqrt{(x_{S_i P_j k_1} - x_{S_i P_j k_2})^2 + (y_{S_i P_j k_1} - y_{S_i P_j k_2})^2}$$

and $C_{S_iP_jk}$ is the sound speed in the k^{th} grid at source sensor path which is given by

$$C_{S_iP_ik}(T,W) = C_{S_iP_ik}(T) + C(W)_{S_iP_ik}cos(\psi_{S_iP_i})$$

and $\xi_{S_iP_j}$ is measurement noise which was assumed to be Gaussian. In the simulation frameworks we have implemented piecewise summation of time in (6.5) to calculate the actual TDOA measurements between probes and sensor pairs. The algorithm to calculate the total propagation time from k^{th} probe to i^{th} sensor is given in Algorithm 1.

```
Algorithm 1 Calculating total sound propagation time from k^{th} probe to i^{th} sensor
  Step 1: get wind velocity, temperature matrix over all grids
  Step 2: calculate sound speeds at each grids
  Step 3: find all grid cross points in probe sensor path
  Step 4: calculate probe-sensor path angle WRT x axis (\theta_{P_k S_i})
  if \theta_{P_kS_i} \leq \pi/2 then
     for thisPiece \in (piece1,..., piece(totalCrossPoints-1)) do
         if piece1 then
             thisPieceVelocity = probe belong grid sound speed
         else if piece crosses only Horizontal grid line then
             thisPieceVelocity = sound speed at the grid below previous grid
         else if piece crosses only Vertical grid line then
             thisPieceVelocity = sound speed at the grid left to previous grid
         else if piece crosses Horizontal grid line and Vertical grid line then
             thisPieceVelocity = sound speed at the grid south west to the previous grid
             thisPieceTime= pieceLength/thisPieceVelocity
         end if
         totalTime=\Sigma thisPieceTime
     end for
  else if \theta_{P_kS_i} > \pi/2 then
     for thisPiece \in (piece1,..., piece(totalCrossPoints-1)) do
         if piece1 then
             thisPieceVelocity = probe belong grid sound speed
         else if piece crosses only Horizontal grid line then
             thisPieceVelocity = sound speed at the grid below previous grid
         else if piece crosses only Vertical grid line then
             thisPieceVelocity = sound speed at the grid right to previous grid
         else if piece crosses Horizontal grid line and Vertical grid line then
             thisPieceVelocity = sound speed at the grid south east to the previous grid
             thisPieceTime= pieceLength/thisPieceVelocity
         end if
         totalTime=\sum thisPieceTime
     end for
  end if
```



Figure S9: Wind direction variation map over the simulation area. The wind direction and the speed variation over the area are represented assuming Markov random field property. Simulation parameters were chosen as follows: Initial mean values for wind direction is $\theta_w = 120^0$ and $\sigma_{wDirection} = 0.1$.

6.7.3 Algorithm for Source Localization with Probes

Overview of the algorithm used in simulations for source localization with probe technique is summarized in Fig. S11.

6.7.4 Performance of the Probe Technique under Different Environmental Conditions

In this section we present further performance comparison between our probe technique and the conventional sensor networks. The performance is presented under three different Sensor/ Probe models as in Section 6.5 of this chapter.

Sensor/ Probe setup 1: At every sensor and probe position, a **wind** and **temperature** sensors were incorporated. The sound speed estimation was done through effectively averaging the point measurements with respect to each sensor as explained in Comparison Model 1 of this chapter. (i.e. Not through probe technique).

Sensor/ Probe setup 2: At every sensor and probe position, a **wind** and **temperature** sensors were incorporated. Gaussion function and Markov random field properties



Figure S10: Sound source localization under variable sound speeds in observation area. Each grid consists of constant wind velocity and temperature. However, temperature and wind velocity varies between each grid. Therefore, sound signal experiences different sound speeds in diverse regions it is travelling in. For instance, the effective sound speed in the path of *E* and S₁ at grid 2 is given by $C_{S_1E2} = C(T_2) + V_{w2} \cos \xi_{S_1E2}$. Similarly, the sound propagation speed between *E* to Sensor S₁ will be affected by temperatures T₂, T₄, T₅, T₇ and wind velocities V_{W2}, V_{W4}, V_{W5}, V_{W7} in between the propagation path.

utilized to model the environmental effect as explained in Comparison Model 2 of this chapter. (i.e. Not through probe technique).

Sensor/ Probe setup 3: No temperature or wind sensors utilized in the system. The sound speed estimation was done solely depending on probe technique as described in Model B of this chapter.



Figure S11: Summarized algorithm used in simulations explained in Section 6.4 of this chapter. The above algorithm is used to estimate unknown source (Elephant) location incorporating probe technique.



sor set up as in Comparison
Model 1sor set up as in Comparison
Model 2sor / Probe set up as in Model
BFigure S12: Performance analysis of three different sensor / probe models under the given

temperature and wind velocity variation. Sensor/ probe setup models are described in Section 6.3. In this scenario temperature is kept constant over the area. Ten probes and 12 sensors are deployed as explained in Section 6.3-Model B of this chapter.



(d) Performance under Sen-
sor set up as in Comparison
Model 1(e) Performance under Sen-
sor set up as in Comparison
Model 2(f) Performance under Sen-
sor/Probe set up as in Model
B

Figure S13: Performance analysis of three different sensor/ probe models under the given temperature and wind velocity variation. Sensor/ probe setup models are described in Section 6.3. In this scenario wind velocity is kept constant over the area. Ten probes and 12 sensors are deployed as explained in Section 6.3-Model B of this chapter.



(d) Performance under Sen-
sor set up as in Comparison
Model 1(e) Performance under Sen-
sor set up as in Comparison
Model 2(f) Performance under Sen-
sor/ Probe set up as in Model
B

Figure S14: Performance analysis of three different sensor/ probe models under the given temperature and wind velocity variation. Sensor/ probe setup models are described in Section 6.3. In this scenario performance under low variability of temperature and wind velocity is depicted. Ten probes and 12 sensors are deployed as explained in Section 6.3-Model B of the this chapter.



varia- (b) Wind speed variation (c) Wind direction variation (a) Temperature 22°C, map. $V_w = 40 \text{ ms}^{-1}$ and map. tion map T_{min} 120⁰ and = θ_w = $\Delta T_{Vmax} = 15^{\circ}C$ $\sigma_{wSpeed} = 1$ $\sigma_{wDirection} = 2$







(d) Performance under Sen- (e) Performance under Sen- (f) Performance under Sensor set up as in Comparison sor set up as in Comparison sor/Probe set up as in Model Model 1

Model 2

В

Figure S15: Performance analysis of three different sensor/ probe models under the given temperature and wind velocity variation. Sensor/ probe setup models are described in Section 6.3. In this scenario performance under highly varying of temperature and wind velocities is depicted. Ten probes and 12 sensors are deployed as explained in Section 6.3-Model B of the this chapter.

Chapter 7

Sensor Hardware Implementation, Data Collection and Analysis

Mitigating human-elephant conflict is a real-world challenge and it requires practical implementations. This chapter presents the implementation of hardware systems related to our approach. A sensor network that employs a sound probe technique has been built for data collection. Initially, we performed practical experiments by replaying recorded elephant sounds under different environmental conditions. Finally, the ideal system set up is tested in a real-forest environment to localize wild elephants in Sri Lanka and results. The results overall show that the system is capable of providing remarkable accuracy under distinct wind and temperature conditions.

7.1 Introduction

THE amount of implemented wireless sensor network systems is extremely low compared to the theoretical publications presented in the area [14]. The problem discussed in this thesis requires actions in the ground to mitigate human-elephant conflict. This chapter explains system hardware that we have built related to the application and real experiments conducted in the field.

Implementing hardware system to confirm theoretical and simulation analysis is a challenging task. First, it is necessary to identify the requirements related to the application and limitations. Once the hardware is built, the testing needs to be performed in a controlled environment to identify the possible constraints. Therefore, our system hardware development and experiments were carried out keeping the above concepts in mind.

During the initial stage of the project, we have made attempts to implement a stan-

dalone hardware system to record and elephant calls. The Section 7.2 briefly explains the system hardware on this attempt. In Section 7.3 we explain the sensor network hardware system implemented employing the sound probe technique. Section 7.4 describes the practical experiential set up for the tests we conducted in the open field and forest environment. Section 7.5 outlines the experimental results and discusses them further.

7.2 Standalone Signal Recording System

During the initial stage of the project, attempts were made to implement a cheap solution for digital elephant sound recorder system. The hardware system was primarily comprised of a microphone, pre-amplifier, microcontroller, power supply unit and a memory device.

The microcontroller is the interface between input signal processing and the signal recording memory device. It was chosen PIC16F877A as the microcontroller for this implementation which was a mid-range PIC at the time of implementation. MultiMediaC-ard (MMC) was chosen as the recording memory device. The implementation challenge was to access the MMC card using limited random-access memory (RAM) (256 bytes) of PIC16F877A. It was used in-built serial peripheral interface (SPI) of the PIC to read and write signal data. The microcontroller was programmed using PICkit 2 Development Programmer.

MMC works at max voltage of 3.6 V and PIC16F877A operates at 5.0V. Therefore, it was required a voltage level convert to interface PIC with the memory device. A simple resistor-based voltage divider is introduced to the interfacing pins of MMC (CS, SCK and CLK) [127]. As a result, 4.3 V maximum output of the PIC pins appears as 2.8 V to above three pins at the memory device. The chosen PIC also contains an inbuilt 10-bit analogue-to-digital converter (ADC) module where we used to convert amplified analogue sound signals to digital format before writing into the SD card. Pin RA0 of the PIC is configured as ADC input channel where microphone pre-amplifier output was connected. 16 x 2 LCD was interfaced to the microcontroller to recognise the status of the system. Circuit diagram for the microcontroller interfaces is shown in 7.1


Figure 7.1: Microcontroller interface overview. RA0 PIN of the PIC is configured as ADC input channel. RC2 – RC5 PINs are interfaced with MMC with simple resister-based voltage converter. PortD is interfaced to 16 x 2 LCD.

It was chosen condenser microphone as the input device, and it was tested for frequency response using a signal generator and oscilloscope. Through experiments, it was confirmed that the microphone provides a linear response over the low frequencies required for the application. The pre-amplifier also needed to be performed in the low frequencies as the elephant rumbles were having signal components in sub-sonic range. The amplifiers were then designed and simulated using Proteus 7.7 to match the above requirement. Circuit diagram of the amplifier is shown in 7.2.

The power supply was designed to be dual supply with mains AC powered for the testing purposes and battery powered for the recording in the field. Battery charger unit was also implemented such that the system could charge the battery unit while the device is connected to the mains power. Completed system set-up is depicted in 7.3.

The above hardware system was only used in the initial elephant sound recording and was abandoned from further development due to project duration time constrains.



Figure 7.2: Circuit diagram of the signal pre-amplifier. The amplifier was designed with Proteus 7.7 and simulated before implementation. The amplifier performs well in the frequency range of 10 - 4000 Hz range.

7.3 Central Processing Wireless Sensor Network

Following the extensive simulation analysis discussed in Chapter 6, analogous hardware system is designed to test the performance of the proposed probe technique and system framework.

The hardware system consisted of sensor units which contain microphones (Audio-Technica AT897) and FM transmitters (Fmuser SDA-01A) and sound probes which include chirping devices (see Figs. 7.4 and 7.5). A separate laboratory experiment was conducted to investigate the frequency response of the signal sensor microphones detection system. It was confirmed that the system performs well in the range of approximately 10-400 Hz and can capture main signal energy components of low frequency elephant calls as shown in [7].

The sensors listen to the sound signals and transmit them in a dedicated FM channel to the centralized receiver in real time. Each sensor was assigned a distinctive FM channel to prevent the interference at the receiver. These channels were carefully selected to avoid interference with the other FM transmissions such as local radio stations and communication within police forces and emergency services. The centralized receiver consisted of a set of FM receivers (Sony CRSW08) which are synchronized to each sensor



(a) Inside of the standalone recorder hardware unit



(b) Integrated 16 x 2 LCD unit displaying initialization of the recorder system.



(c) Packaged hardware system unit

Figure 7.3: Completed standalone recorder hardware system. The system consists of mic interface, power supply unit, integrated display interface and memory (MMC) interface.

unit FM frequency. The outputs of these receivers were then streamed into a laptop computer via a USB sound card hub which enabled simultaneous signal recording. Sound signal acquisition is attained through a commercial software program named Cool Edit Pro Version 2.0. The signals from each sensor channel was recorded into separate tracks in the software program and subsequently saved and exported into wave files for further analysis. The main feature of the software program that facilitated in our experiment is that it enables synchronized data recording from separate channels. Table 7.1 provides a cost analysis of the hardware system that utilized in this implementation. The overall cost of the system including a cheap processor is less than \$2750.



Figure 7.4: Snap of complete sensor unit deployed in the field. The microphone and the transmitter units are mounted on a tripod and units are protected with weather shields. Four sets of similar sensor units are deployed for the experiment. The transmitter units were powered by 12 V DC battery.

Our ultimate goal is to utilize the system to localize and track the forest elephants through their rumble signals. In our experiment, we decided to utilize a moving sound source which produces elephant rumble signals such that we can maintain the analogy up to some extent. First, we included rumble signals into a sound player (Sansa®



Figure 7.5: Snap of Probe unit deployed in the field. The unit consists of sound chirp generator and a controller unit which determines the timing of chirp generation. The unit was powered by a 12 V DC battery.

ExpressTM) which consisted of a storage device and connected it to a large speaker through an amplifier (TOATM A-50M). The speakers had frequency response of $\pm 3 dB$ over the range 16–250 Hz and nearly all the elephant calls used for the simulation had acoustic energy within the above range. Then the system is mounted on a wheel cart so that it can easily be moved in the field. We called this system setup as "Surrogate Elephant" (SE).

7.4 Experimental Setup

Experiments were conducted in open field and forest area using the hardware system discussed in 7.3 to analyze and validate the elephant localization system framework.

Item	Brand	Model	Unit Price (\$)	No. Units	Cost (\$)
Acoustic	Audio-Technica	AT897	212.00	4	848.00
Sensor					
Long	Fmuser	SDA-01A	112.00	4	448.00
range FM					
Transmit-					
ter					
Receiver	Sony	CRSW08	48.00	4	192.00
Probe	Custom Design	Custom Design	40.00	4	160.00
Battery	B.U.E	3FM-4.0	25.00	8	200.00
Recorder	Cool Edit	Pro V2.0	300.00	1	300.00
Software					
Processor	Toshiba	SATELLITE C50D	499.00	1	900.00
Unit (Lap-					
top)					
Interfacing	N/A	N/A	100.00	1	100.00
units and					
Connec-					
tors					
Total					

Table 7.1: Cost analysis of the Hardware System Utilized in the Real World Implementation

7.4.1 Open Field Experiment

The experiment site was a playground named Weeguluwathte at the city of Gampola which is located in the Central Province of Sri Lanka. The experiment was performed in three sets during different time periods of the day such that natural variation of temperature and wind velocities can be taken into account. The experiment was carried out in a sunny day and humidity was 64%. The average temperature and wind speed with respect to north was obtained using an anemometer (Bentech GM816) during each set and is summarized in Table 7.2.

Table 7.2: Average Wind and Temperature Measurements during the Experiment Trials

Experiment Set	Average Temperature	Average Wind Speed	
	(°C)	(ms^{-1})	
Data Set 1	27.4	6.9	
Data Set 2	29.3	8.7	
Data Set 3	28.1	10.1	

92

The sensors and the probes were deployed in the field as per the reasoning explained in the simulations in Chapter 6. The exact locations of the probes and sensors were acquired using eTrexTM GPS device and recorded. In addition, the distance between each sensor and probe was accurately measured via Leica DISTOTM A8 Laser distance meter. Once the probe sensor system was deployed and activated, the centralized data recorder system was positioned in a convenient location to start the recording. Then a GPS data recorder was attached to our Surrogate Elephant (SE) to obtain the actual path where it was travelling. Subsequently, SE is moved at a constant speed towards the deployed probe sensor direction whilst recording sound signals into the centralized data recording system.

7.4.2 Real Forest Experiment

Following the open field implementation and identifying the possible challenges, such as battery power limitations, general environmental parameter measurement constraints and hardware mounting requirements, we proceeded to test our methodology in real forest environment. The experimental site was a village called Swaranapaligama which was situated in North central province, Sri Lanka. The village is located adjacent to the border of the Wasgamuwa National Park and a branch of Amban river flows close to the village. The primary occupation of the villages is farming and they mainly cultivate rice and vegetables. As a result of close proximity to the forest and the continual production of crops, wild elephants frequently enter into the village especially during the dry season. Considering the above facts, we conducted our experiment in the end of September 2013 which is the latter period of the dry season.

A similar experimental setup as in the open field experiment described in Section 7.4.1 was employed in this implementation as well.

A set of sensors and probes were deployed at predetermined locations. The probes were kept in the forest area to obtain a better representation of the sound characteristics through the woodland (See Fig. 7.7). The sensors are kept close to the village bound-ary depending on practical considerations (See Fig. 7.8). A simplified block diagram of the hardware setup is depicted in Fig. 7.6. The probe and the sensor coordinates were



Figure 7.6: Simplified block diagram of the system hardware implementation. Each sensor consists of microphone and a FM transmitter. Vocalization data and the probe signals acquired through each microphone is transmitted through a dedicated FM channels to the centralized receiver. Probes are placed in the forest area to get a better representation of the real sound speeds of the observation area.



Figure 7.7: Snap of a Probe unit deployed in the forest implementation. The unit was deployed in a predetermined location in the forest area to obtain a better representation of the sound characteristics through the woodland.

recorded using eTrexTM GPS devise as in the previous experiment and those coordinates are summarized in Table 7.3.

	GPS coordinates		
	Longitude	Latitude	
Sensor 1	80.954352	7.863963	
Sensor 2	80.953244	7.864375	
Sensor 3	80.952465	7.864156	
Sensor 4	80.951085	7.864182	
Probe 1	80.953736	7.863446	
Probe 2	80.952939	7.863341	
Probe 3	80.952146	7.863349	
Probe 4	80.951457	7.863455	

Table 7.3: Sensor and Probes Location Coordinates at the Real Forest Experiment

During the background information gathering the villagers informed us that in many occasions, elephants come closer to the village shortly after the nightfall. Therefore, we set up equipment targeting to start the data collection during the above period. Owing to constrains in battery life and available resources, successful data collection was done in three consecutive days and typical weather information during the experiments is summarized in Table 7.4.



Figure 7.8: Snap of a Sensor unit deployed in the forest implementation. The unit was deployed in a predetermined location close to the village boundary.

Experiment Trial	Weather	Average	Average	Humidity
		Temperature	Wind Speed	(%)
		(°C)	(ms^{-1})	
Day 1	Sunny	29.5	4.2	64
Day 2	Sunny	29.3	8.4	72
Day 3	Sunny	28.1	7.1	68

Table 7.4: Weather Details during the Real Forest Experiment

Once the experiment was conducted, in the following day we walked through the elephant tracks to verify the presence of elephants during the previous day. The verification process was practically impossible to conduct in real time during the experiment due to safety issues. However, through the evidence such as fresh piles of elephant dung, broken tree branches, signs of browsing grass and footprints embossed in the ground, we were able to determine the presence of the elephants during the days we conducted the experiment (see Figures. 7.9-7.10). We obtained the assistance of few villages who were



Figure 7.9: A snap shot of recording information on a location where elephants were present in a day that we conducted the experiment. GPS coordinates and information on the observation were recorded manually for further reference.

experts in identifying elephant tracks and familiar with the forest area closer to the village to accomplish the above task. We recorded the coordinates of the elephant tracks using eTrexTM GPS device. In addition, GPS coordinates and the description of the signs were manually recorded at the specific locations where we identified as elephants were present. Through these field visits we identified that, out of three days we conducted the experiment, in two days elephants have travelled through the tracks we observed.

7.5 **Results and Discussions**

7.5.1 Open Field Experiment Results

The results of the practical implementation of our proposed methodology in an outdoor environment are depicted in the Figures 7.14–7.16 for three different data collection sets. It can be identified that our proposed technique clearly improves the localization accuracy of the sound source under different environmental conditions. The accuracy improves whilst the source is close to the probes. The reason for this is that, when probes



Figure 7.10: Picture of footprint embossed on the ground of an elephant that visited the area on Day 1 when we conducted the experiment. Observation locations was recorded for further reference.

and SE are in close proximity, the probes technique represents the actual sound speeds between source—sensor paths. Due to limited space and narrow variation of temperature and wind velocities in the experimental area, the localizations without probe technique correction also follow the path to a certain extent. However, this fact is invalid in a real forest application where environmental condition variation would be much more complicated.

The elephants have a behavioral characteristic of walking in the same particular paths which they have followed throughout the generations. Therefore, it is possible to incorporate these path data as a prior information whilst implementing tracking algorithms. We have implemented a particle filter considering prior information of the path data of the surrogate elephant whilst incorporating probe correction in our posterior analysis. As shown in Figures 7.14–7.16 the outcome of the implementation clearly indicates an accuracy improvement.



Figure 7.11: Snap of fresh pile of elephant dung on the ground in the elephant track. This observation was made on Day 1 of the forest experiment. Observation location was recorded for further reference.



Figure 7.12: Snap of earth marks made due to elephat bashings on the ground by an elephant on Day 3 of the foreset experiment. Observation location was recorded for further reference.

7.5.2 Real Forest Experiment Results

Figure 7.17 depicts the localization results from gathered data during the real forest experiment. The analysis revealed that elephant calls were recorded during only two days of the experiment. This is in fact verified through the observations made during the post experimental visits through elephant tracks. The probe technique has performed efficiently in this heterogeneous environment as well. Evidently, it surpasses the observation made without considering the wind and temperature variations. Due to large observation area and variations in the environmental facts, the remarkable performance of the



Figure 7.13: Snap of marks made on a tree by an elephnat on Day 3 of the foreset experiment. Observation location was recorded for further reference.

probe technique can be clearly identified. Our technique has located elephants within 30 m to the elephant tracks whilst error in conventional methodology extends over 70 m. The evidence gathered indicates the real localization accuracy attained through our technique. In addition, the results indicate that under the forest environmental condition, the performance of a conventional localization system can be significantly impacted even under a modest temperature and wind velocity variations.

Our system was able to localize elephants over 500 m margin from the village boundary. This implies that, with some modifications to the hardware system such as integrating additional amplifiers the system may perform well above this distance. This fact suggests that similar system can be utilized to monitor elephants noninvasively for behavioural aspects. In a broad sense, our system which costs around \$2750 can be utilized to protect a village area of around 2 km². However, the system needs to be modified for continuous power requirement in an alerting system. The challenge of battery power constraints can be overcome incorporating hybrid solar and battery power system. Similar methodology use in natural disaster notification which uses Short Message Ser-



(a) The average temperature during the experiment was 27.4 °C and average wind velocity 6.9 ms⁻¹ during the experiment. The estimated localization positions of the Surrogate Elephant closely follow the actual path whilst the probe technique correction was made.



(b) Surrogate Elephant localization error estimation for Data Set 1. The error is calculated deviation in Y direction from the actual path. Note that the ES is moved towards the sensors and 1st data point represent the furthest point from the sensors.

Figure 7.14: Localization of Surrogate Elephant; The analysis results from the Data Set 1 acquired from the real experiment.



(a) The average temperature during the experiment was 29.3 $^{\circ}$ C and average wind velocity 8.7 ms⁻¹ during the experiment. The estimated localization positions of the Surrogate Elephant closely follow the actual path whilst the probe technique correction was made.



(b) Surrogate Elephant localization error estimation for Data Set 2. The error is calculated deviation in Y direction from the actual path. Note that the ES is moved towards the sensors and 1st data point represent the furthest point from the sensors.

Figure 7.15: Localization of Surrogate Elephant; The analysis results from the Data Set 2 acquired from the real experiment.



(a) The average temperature during the experiment was 28.1 $^{\circ}$ C and average wind velocity 10.1 ms⁻¹ during the experiment. The estimated localization positions of the Surrogate Elephant closely follow the actual path whilst the probe technique correction was made.



(b) Surrogate Elephant localization error estimation for Data Set 3. The error is calculated deviation in Y direction from the actual path. Note that the ES is moved towards the sensors and 1st data point represent the furthest point from the sensors.

Figure 7.16: Localization of Surrogate Elephant; The analysis results from the Data Set 3 acquired from the real experiment.



(a) The analysis result of the data collection from the Day 1 in real forest experiment. The average temperature was 29.5° C and average wind velocity was 4.2 ms^{-1} during the experiment.



(b) The analysis result of the data collection from the Day 3 in real forest experiment. The average temperature was 28.1° C and average wind velocity was 7.1 ms^{-1} during the experiment.

Figure 7.17: Analysis results of the real forest experiment. Observations were associated to different tracks in Day 1 and Day 3. The estimated elephant localization positions closely follow the elephant tracks. Evidently, the probe technique has significantly improved the localization accuracy despite the complex terrain.

vice (SMS) alerting system [128] can be utilized for warning villagers regarding elephant threats.

In [124], acoustic monitoring of elephants is mentioned as a valuable noninvasive research tool. However, the predicted area of detection depends upon the atmospheric acoustic state. This acoustic state is mainly affected by meteorological effects such as temperature and wind velocity. It is necessary to have real-time knowledge of atmospheric condition to accurately locate the calling elephant [124]. Our proposed methodology addresses exact same problem and adjust the sound speed for atmospheric bias which results greatly improved localization accuracy.

Mayilvaganan et al. have proposed direction of arrival estimation methodology with partial hyperbolic circular array to localize elephants [129]. They also identified that wind velocity and temperature variation significantly affects the localization accuracy. In fact, their analysis shows that localization error could be more than 28 m under narrow wind condition of 2.2 ms⁻¹ whilst four sensors and the source is deployed in a 100 m × 100 m small-scale area. In comparison, analysis results indicate that our methodology performs far better than the above approach in locating elephants even under much stronger wind conditions.

The implemented hardware system is an economical approach for actual elephant localization system. By means of a centralized data collection system, we were able to eliminate the clock synchronization problem which is a main concern in distributed data collection schemes in similar applications. In addition, the inexpensive and effective methodology of utilizing probe technique remarkably increases the localization accuracy.

The sound signal at the receiver itself models the propagation channel characteristics. In our proposed technique this phenomenon is effectively utilized to estimate the sound propagation speed which is affected by temperature and wind variations. Further, information from sensors and probes which represent different spatial regions evidently assists for better sound speed estimation. Besides, the remarkable advantage of this approach is that it does not need both wind and temperature sensors in the network. In addition, we make use of the existing acoustic sensors for sensing probe signals and the elephant calls.

7.6 Chapter Summary

In this chapter we have presented hardware implementation on sensor network that employs a sound probe technique. We have further investigated the feasibility of enhancing the accuracy in elephant localization system by utilizing the sound generating probe technique in practical environment. The inexpensive hardware platform implemented is utilized to evaluate the proposed methodology in an outdoor environment under equivalent elephant localization scenario. Finally, the system was deployed in a real forest environment to localize wild elephants in Sri Lanka. The outdoor experimental implementation verified the significant improvement in accuracy whilst using the probe technique, in a surrogate elephant localization system. Further, the real forest implementation successfully localized elephants over 500 distance. The results confirmed the success of our methodology by enabling to localize wild elephants within 30 m accuracy to their natural walking paths in the real forest implementation.

Chapter 8 Conclusion

8.1 Summary of Contributions

THIS thesis focused on providing an engineering solution to mitigate human-elephant conflict by proposing a methodology to accurately localize the elephants before they move into the village areas. The findings are important not only to ascertain the importance of animal behavioural studies in conservation projects, but also to identify the challenges in animal localization using acoustic sensor networks. The contributions of this thesis are presented in Chapters 4, 5, 6 and 7. The thesis provide details evaluation of useful elephant behavioural patterns in mitigating HEC, the feasibility of using passive acoustic sensor network to localize elephants, mathematical modelling and comparative study of different source localization algorithms, challenges affecting elephant localization accuracy due to environmental effects and novel methodology to improve localization accuracy.

Chapter 2 briefly outlined the earlier researches carried out on habitat monitoring using sensor networks. It also discussed current methodologies used in mitigating HEC and the concerns related to these approaches in terms of effectiveness and availability of resources.

In Chapter 3, we discussed the structure of the elephant society and the behavioural patterns of these highly social animals. The behavioural patterns that ensue HEC and reasons concerning the impossibility of entirely illuminating HEC are emphasized in this chapter. The ability of elephants to adapt in a broad range of ecological conditions always reassure that they will move into village areas for the need food requirements. However,

their behavioural tendency to use same corridors allows our approach to locate them in certain known paths which then can be passed into the authorities and villages to take preventive measures. This chapter also discusses the long-range elephant communication call pattern called "rumbles" which is the key in locating elephants in far away distance and presented analysis of rumble signals in frequency and time domain.

Chapter 4 describes detail approach in localizing elephants using timing observation measurements. We outlined two mathematical modeling approaches for elephant localization system framework and derived elephant positioning estimators using BLUE and MLE. The performances of the models are analysed defining a simulation framework and resulted are illustrated in the chapter. The analysis suggested that there is no significant advantage in increasing the number of sensors above certain number to attain successive convergence of the algorithm for elephant location. Similar framework can be defined, and analysis can be performed in future research to identify the required scale of the sensor network with respect to the application. Finally, the chapter presents the statistical bounds of localization error on above mathematical models using FIM and covariance ellipses.

In Chapter 5, we investigated environmental constraints and results of SNR variations on elephant localization accuracy. The elephant rumble calls are the signal for our passive acoustic sensor network and signal travel through the air. Sound propagation in air is affected by various meteorological parameters such as temperature, humidity, water vapour mole fraction and CO_2 content. However, the speed of sound mainly depends on temperature and effect of other parameters are negligible. Consequently, in this chapter we mainly focused on the effect on temperature variations on elephant localization accuracy. In addition, sound signals are longitudinal waves and wind carry the medium that sound signal travels. Therefore, we investigated the effect on changes in wind velocity on our localization accuracy. Moreover, the sound signal intensity is inversely proportional to distance of travel and therefore, SNR varies based on the distance. In this chapter we further investigated the SNR dependency on the elephant localization accuracy. We defined a simulation framework and performed extensive simulation analysis to examine significance on above parameters on elephant localization system that we propose. The findings illustrated that SNR is an important fact that affects the localization accuracy. The system to converge into a source position the SNR need to be above certain marginal level. This has been given as -12 dB with respect our simulation framework discussed in the chapter. In addition, analysis has illustrated uniform temperature variation has insignificant impact on localization accuracy. In contrast, the variation in temperature between source–sensor path significantly impact localization accuracy. Wind velocity over the observation area also negatively impact on localization accuracy even the variation is uniform. For instance, a uniform wind velocity of 20 ms⁻¹ results in about a 100 m (20%) error at a distance of 500 m. An additional temperature variation of 4 °C can result in an increase in the error to over 175 m (35%). Therefore, chapter concluded mentioning the importance of implementing wind and temperature compensation mechanism in a real implementation of elephant localization and similar system.

Following the extensive analysis on the feasibility of using acoustic sensor network for elephant localization in Chapter 4 and Chapter 5 and the impact on wind and temperature variations on positioning accuracy of such system, in Chapter 6 we propose novel technique to improve the elephant localization accuracy. Sound generating probes are introduced into the passive acoustic sensor network which interns act as a probe that provide a means to estimate accurate sound speed over the observation area. In Chapter 6, the effectiveness of incorporating sound probes into the system in terms of localization accuracy is widely analysed using simulation framework. Comparison models with temperature and wind velocity sensors also introduced for comparative assessment of the probe methodology. The findings suggested that the technique can limit the mean estimation error under 0.06 throughout the interested observation area regardless the non-uniformity of temperature and wind speed of the area. Sound signal travel through the medium itself models the propagation channel characteristics. We introduced probes and an algorithm to effectively utilize this phenomenon to estimate sound propagation speed which is affected by temperature and wind variations. The remarkable aspect of this approach is that it does not need both temperature and wind velocity sensors in the network and we make use of the existing acoustic sensors for sensing probe signals and the elephant calls.

HEC is a real-world problem and it requires practical implementations to reduce the occurrences of conflict. Chapter 7 presents the hardware systems constructed, and the real-world implementation deployed to prove the concepts proposed in previous chapters. Sensor network incorporating the sound probes was built, and experiments were conducted under different ecological conditions in an outdoor environment using surrogate elephant which replicates elephant rumbles. Then the identical experiment setup is deployed in a forest environment to localize wild elephants in Sri Lanka. The analysis results suggest that probe technique has performed efficiently in this outdoor environment as well as heterogeneous forest environment. Our inexpensive methodology proved its effectiveness by enabling to localize wild elephants within 30m accuracy to their natural walking paths in the real forest implementation.

8.2 Future Research

In this thesis, we investigated the feasibility of using acoustic sensor network to localize the elephants. The hardware system is presented for sensor network and data was collected for very limited period. The extended behaviour of the sensor network has not been studied due to project time limitations. Considering the sensitivity of the issue, if the network is deployed as the primary source of information for alerting villages, the extended behaviour of our system framework need to be studied and potential concerns should be identified. This will require further research on areas such as sensor power optimisation and designing standalone sensors and probe units that capable of long-term operation.

This thesis limited the consideration of implementation for a single sensor network that operates in an observation area about $2 \ km^2$. This assumption of observation area is made based on the signal transmission distance of the elephant rumbles and the application requirement of identifying elephants in a location closer to the villages. In practical consideration, this could probably be covering single village area, and each affected village will need a separate elephant localization framework. This leads sensors to have cross-talk effect especially at the boundaries of sensor networks. However, this could be eliminated by incorporating sensor scheduling algorithms and cell planning methodologies yet to be confirmed through careful investigations with respect to the application.

The sensors introduced in this research acts as listening and continues transmission devices whilst the signal processing is conducted at the central processing unit. This approach increases sensor power consumption which interns a major concern in long-run in terms of sensor maintenance. If the sensors made smart to transmit the signals only when the rumble signals are identified, the sensors could run in standby mode for a longer period while increasing the sensor life cycle. This approach requires the development of standalone smart sensors which has signal processing capabilities. With the development of present hardware processing units such as graphics processing units (GPU) this should be feasible, however rigorous design research procedures need to be followed.

The thesis concerns only on identifying the presence of elephants. It has been identified that some of the elephants are more problematic than others. On the other hand, similar to individual human voices, elephants may have distinctive spectral features in their calls which could lead to identify individual elephants. Through extensive data collections and by incorporating speech recognition techniques such as classification and feature extraction or principle component analysis (PCA), this should be feasible. If the elephants can be distinctively identified, this information can be utilised by the authorise to make informed decisions based on prior characteristics of the individual elephant.

In addition, thesis does not consider to distinguish the number of elephants present in the observation area. If it is assumed individual elephant calls have distinctive features it may be possible to estimate the number of elephants moving towards the village area. This problem is similar to the well-known "cocktail party problem" however in a much more challenging environment. Techniques such as independent component analysis (ICA) and blind source separation may be useful in resolving above problem.

Other future research directions would be to define algorithms for optimal placement of probes and sensors. This will be a complex research problem considering it needs to incorporate terrain information as well as signal transmission models and distance constraints. The problem can be further extended by amalgamating elephant track information into the system and assessing the optimal placement of both sensors and probes, knowing the elephant tracks.

Appendix A Propagation Constraints in Elephant Localization Using an Acoustic Sensor Network

A.1 Introduction

The content of the thesis has been partially published in an international conference and a journal. Appendix A presents the published material in international conference:

Chinthaka M. Dissanayake, K. Ramamohanarao, M. N. Halgamuge, B. Moran, and P. Farrell "Propagation Constraints in Elephant Localization Using an Acoustic Sensor Network", IEEE 6th International Conference on Information and Automation for Sustainability (ICIAfS'12), pp 101-105, Beijing, China, 27-29 September 2012

A.2 Published manuscript

ICIAfS'12 1569626399

Propagation Constraints in Elephant Localization Using an Acoustic Sensor Network

Chinthaka M. Dissanayake*[†], Student Member, IEEE, Ramamohanarao Kotagiri[‡],

Malka N. Halgamuge[†], Member, IEEE, Bill Moran[†], Member, IEEE, and Peter Farrell[†],

Department of Infrastructure Engineering*, Department of Computer Science and Software Engineering[‡],

Department of Electrical and Electronic Engineering[†],

The University of Melbourne, Parkville, VIC 3010, Australia.

 $\label{eq:email: c.dissanayake@pgrad.unimelb.edu.au, \{malka.nisha, kotagiri, wmoran, pfarrell\}@unimelb.edu.au, \{malka.nisha, kotagiri, wmoran, pfarrell\}@unimelb.edu.au, \{malka.nisha, kotagiri, wmoran, pfarrell}\}@unimelb.edu.au, kotagiri, wmoran, pfarrell}\}@unimelb.edu.au, kotagiri, wmoran, pfarrell}\\$

Abstract—In this paper we discuss elephant localization framework using acoustic sensor network. We further analyze the effect of various meteorological parameters in elephant vocalization propagation and their consequential effects on elephant positioning accuracy. In addition, we present the effect of signalto-noise ratio (SNR) on detection of vocalization signal which consequently affects the localization accuracy. Extensive simulation results reviled that the uniform distribution of the temperature throughout the localization area does not defect the localization accuracy. Our results show that temperature variations between sensors and wind velocity and direction have significant effect on positioning accuracy of the system. Regardless of this, it is revealed that the localization accuracy decreases enormously once the SNR level fall under certain marginal level.

I. INTRODUCTION

Elephants and human were built with a long tradition of interdependence up until contemporary times. However, today human-elephant conflict (HEC) has become one of the most significant socio-ecological problems in some parts of the world. The conflict primarily is a consequence of frequent attacks from crop riding elephants on the rural agricultural communities. The consequences of HEC made loss of human lives or permanent disability, damage to the properties and cultivation while making negative attitude towards the elephants [1]. Subsequently, elephant deaths resulted from retaliatory killing by people has become ever increasing number during past few years, making Asian elephants (*Elephas maximus*) categorised as Endangered in the International Union for Conservation of Nature (IUCN) Red List [2].

Traditionally, several approaches have been taken to mitigate HEC such as lighting fires and fire crackers and using electric fencing systems. Recently, technological progression leads to use radio and Global Positioning Systems (GPS) collars to detect the presence of elephants [3]. However, these techniques suffer from various drawbacks including nonavailability of resources and operational difficulties. Therefore, interest has been growing in detecting elephant through their low frequency elephant calls called rumbles [4]. Previous experiments carried out have shown that these elephant calls travel over a range of at least two kilometres [5]. In this work, we focus on utilizing acoustic senor network to detect and localize elephants through rumble signals. Consequently, we quantitatively analyze the effects of various environmental parameters on the positioning accuracy of the system.

The remainder of the paper is organized as follows. In section II we discuss about the sound propagation in air as the basis for understanding the environmental effects. Section III we analyze the localization algorithm that utilized in our framework. In Section IV we explain the simulation setup and Section V outlines numerical results and discuss it, further. Section VI concludes the paper.

II. SOUND PROPAGATION IN AIR

The accuracy of elephant localization and detection significantly depends on the characteristics of sound propagation in the air. The sound propagates through the air as longitudinal waves and the propagation is affected by the properties of the medium. For instance, the variations in the metrological parameters such as temperature, wind, relative humidity ect. results in the sound propagation variations in terms of intensity and speed. In this paper we basically focus on temperature and wind effects on the speed of sound.

A. Sound Propagation Model

Elephant vocalization generation can be assumed as a point acoustic source and if the medium of air is assumed to be homogeneous, the sound propagation is considered to be spherical. Therefore, the sound intensity in certain direction is inversely proportional to the expanding surface area of the sphere. The sound pressure level (SPL) at distance r is given by $SPL = SWL_{point} - 10log(4\pi r^2)$ dB where SWL_{point} represent the sound pressure level measured at one meter from the source. The model can be simplified to $SPL = SWL_{point} - 20log(r) - 11$ dB which is often referred as the standard inverse square law for point source sound propagation [6]. In summary, relative sound intensity is reduced by 6 dB per doubling of distance. Depletion in sound intensity reduces the SNR levels at the sensors. Consequently, signals of interest which concealed in the background noise become difficult to detect.

B. Temperature Effect

The speed of sound in air changes with respect to variations in several phenomena such as temperature, water vapour mole

978-1-4673-1975-1/12/\$31.00 ©2012 IEEE

ICIAfS'12

fraction, pressure and CO_2 content in the air. Cramer [7] has proposed a closed form equation for the relationship between speed of sound and the above environmental factors. However, the speed of sound mainly depends on temperature and effects of other variables are insignificant. Therefore, the speed of sound in air can be expressed in terms of temperature as $C_{Sound_T} = C_{Sound_0} \sqrt{(1+\frac{T}{273.15})}$ where $C_{Sound_0} = 331.45$ ${\rm ms}^{-1}$ is sound speed at 0^0C [8]. In general, per degree Celsius rise in temperature, the speed of sound increases by 0.61 ${\rm ms}^{-1}$.

C. Wind Effect



Fig. 1. Wind effect on sound speed in the direction of propagation. Effective component of V_2 along the elephant—sensor direction should be incorporated.

Sound propagation medium of air itself is in a motion due to the wind. The sound propagating through the air will be transported by the moving air mass. Therefore, when the sound propagates towards the direction of the wind, the sound speed will decrease. On the otherhand, if the sound propagates along wind direction, the speed of sound will increase. In a situation, wind makes an angle with the sound path it is necessary to take the vector component along the sound path. In our application, sound path is the elephant–sensor path and Fig. 1 depicts a situation where wind flows in an angle of $\beta(=|\theta_2 - \theta_1|)$ with respect to sound path. Let V_1 be the speed of sound without wind effect, and V_2 be the wind speed. Then, the resultant sound speed V in the direction of source–sensor is given by

$$V = V_1 + V_2 \cos(\beta).$$

In the following section we present the localization algorithm utilized in our framework.

III. LOCALIZATION ALGORITHM

Once an elephant (source) generates a vocalization, the sound signal propagates through the air following the model discussed in Section II. The sensors at the vicinity of the source receive the attenuated and noise corrupted signal. In our analysis, we assume that all the sensors in our deployment receive the elephant vocalization generated at our interested observation area. Then, we obtain the time difference of arrival (TDOA) information by performing pair—wise crosscorrelations between observation signals at each sensor pair.

Let the time of arrival measurements at sensor i to be t_i where $i = 0, 1, \dots, N-1$. The infrasonic acoustic sensors are deployed at pre-determined coordinates (x_i, y_i) . It is



Fig. 2. N number of sensors are deployed at the vicinity of the expected elephant location. Here, the distance difference between source and the p^{th} and q^{th} sensor is defined as g_{pq} .

necessary to locate the source position (x_s, y_s) . The arrival time measurements can be modelled by

$$t_i = T_0 + \frac{R_i}{c_i} + w_i \qquad i = 0, 1, \cdots, N - 1 \qquad (1)$$

where T_0 is the signal emitted time by the source, R_i is the range difference between source and the i^{th} sensor, c_i is the sound propagation speed and w_i is the uncorrelated measurement noise with zero mean and σ^2 variance. It should be noted that the speed of sound between source and each sensor could vary due to the ecological effects discussed in Section II. The distance from source to i^{th} sensor R_i is given by

$$R_i = \sqrt{(x_s - x_i)^2 + (y_s - y_i)^2}.$$
 (2)

The signal emitted time T_0 in (1) is unknown. Therefore, now we incorporate TDOA measurements to the model. Let the TDOA measurement between p^{th} and q^{th} sensor denoted by Δt_{pq} which is given by

$$\Delta t_{pq} = t_p - t_q + w_{pq} \qquad (3)$$

where $p = 0, 1, \dots, N-2$ and $q = p + 1, p + 2, \dots, N-1$. The equation (3) can be written in terms of (2) by

$$\Delta t_{pq} = \frac{R_p}{c_p} - \frac{R_q}{c_q} + w_{pq} \tag{4}$$

which can be explicitly characterized by the non linear model

$$\mathbf{Y} = \mathbf{h}(\mathbf{x}) + \mathbf{w} \tag{5}$$

where

$$\mathbf{Y} = \begin{bmatrix} \Delta t_{01} \\ \Delta t_{02} \\ \vdots \\ \Delta t_{N-2,N-1} \end{bmatrix}$$
$$\mathbf{h}(\mathbf{x}) = \begin{bmatrix} \frac{R_0}{c_0} - \frac{R_1}{c_1} \\ \frac{R_0}{c_0} - \frac{R_2}{c_2} \\ \vdots \\ \frac{R_{N-2}}{c_{N-2}} - \frac{R_{N-1}}{c_{N-1}} \end{bmatrix}$$



Solving non linear equation in (5) can be approached with diverse techniques [9] [10]. We appreciate the problem in a geometrical approach as depicted in Fig. 2. Lets assume that the $w_{pq} = 0$ and the $c_p = c_q = c$ for $\forall (p,q) = 0, 1, \cdots, N-1$. Here, Δt_{pq} corresponds to the distance difference between source and the pair of sensors p and q. Let

$$h_{pq} = c \times \Delta t_{pq} \tag{6}$$

and

$$g_{pq} = R_p - R_q \tag{7}$$

where both (6), (7) represent the distance differences. However, g_{pq} consists of the unknown source location. Therefore, the source location can be determined by solving the function defined by

$$f(x_s, y_s) = \underset{x_s, y_s}{\operatorname{argmin}} \sum_{p, q} (h_{pq} - g_{pq})^2.$$
(8)

However, the location estimation found by (8) does not consider the measurement errors and the sound propagation velocity variances discussed previously. Therefore, by introducing those constraints externally to the localization routine, we can attain the effect of those parameters. The next section of the paper presents the simulaions of the estimation accuracy that affected by different constraints.

IV. SIMULATION FRAMEWORK

The simulation is involved in two major steps. Former part of the simulation is carried out to implement elephant vocalization signal detection at the sensors. This routine also provides the time delay information between each pair of sensors. In this section, the SNR effects on signal detection and acquiring time delay information are also considered. The latter part of the simulation estimate the source position utilizing the time delay information provided by the previos section. The wind and temperature effect on source position estimation is considered in this section.

A. Vocalization Detection

In real world scenario, once an elephant made a rumble signal, each sensor in the vicinity of the source location hear time delayed and noise corrupted versions of the original signal. This is simulated by shifting and zero padding the original rumble vocalization data file by relevent time delays for each sensor. Then, the noise corruption is incorporated by adding white Gaussian noise to time shifted signal. Now at the receiver, if the signal is cross–correlated with signature rumble vocalization signal, it is possible to find the peak cross–correlation coefficient at number of samples which represent the original time delay information. We perform multiple cross–correlations between each pair of sensors' observation signals to find the time delay information represented by Δt_{pq} in (6).

B. Sensor placement and Source localization

In this part of the simulation several real world aspects are taken into the consideration. We chose 2000 m \times 2000 m square as our analysis area which represent the woodland. We assume that coordinate axis y = 0 as our village boundary. Countermeasure can be taken only if the elephants are localized at least before they enter into the 500 m margin from the village boundary. Therefore, we choose interested source observation area as the rectangle bounded by coordinates $\{(0, 500), (0, 2000), (2000, 500), (2000, 2000)\}$.

Sensors might need occasional maintenance for instance, to replace the batteries, to sensor health checks etc. Practically people could not frequently move into deep forest area which is not pleasing in terms of risk and the cost of visit. Therefore, the sensors need to be placed close to the village boundary. We decide that the sensors should be placed within 200 m distance from the village boundary. Consequently, the sensor deployment area is a rectangle bounded by coordinates {(0,0), (0,200), (2000,0), (2000,200)}. However, sensors need to be placed optimally so that they avoid the redundancy and provide maximum accuracy. We, considered several options, such as placing sensors randomly, homogeneously, in a line and in a curvature. In the line deployment, we selected centre line of the chosen sensor deployment strip (y = 100m) and we deployed sensors in equidistance separations. In curvature deployment, proper arc is found as follows. First, we drew a circle centred at (1000, 2000) and radius 2000 m. Then we select the arc which is laid inside the sensor placement strip. Then the sensors are placed such that they are separated each other by equal arc distances. In our simulations we used four sensors which have identical capabilities. It is identified, by means of several simulation experiments that the curvature sensor placement outperforms other methods in source localization. Therefore, it is chosen as the sensor placement criteria for the results presented in this work.

The effect of each constraint discussed in Section II is attained by placing the source at different positions in the interested observation area and calculating the estimation error. Initially, the rectangular observation area is divided into 100 $m \times 100 m$ small girds. Then, the source is placed 100 times randomly inside the small grid and the source position is estimated according to the algorithm discussed in Section III. However, each of position estimation obtained here is affected by different parameters such as wind, temperature and SNR and thus they are erroneous. We compute the error for each random source position by calculating the Euclidean distance between actual and the estimated source locations. Then the mean estimation error is calculated for the above 100 source placements. The computed mean error is considered as the average inaccuracy in positioning when the source is located at the area represented by that particular grid. The above process is repeated for each grid in the observation area. The effects of different meteorological parameters and SNR are presented in the subsequent section.



Fig. 3. The source localization error variation at different source positions within the observation area. Four sensors are deployed in curvature with equal arc length separations as discussed in Section IV-B. The $\mathbf{T}=\mathbf{22}~^{0}\mathbf{C}$ uniform throughout the area. $V_{2}=0~\mathrm{ms}^{-1},~\theta_{2}=270^{0},~\mathrm{SNR}=10~\mathrm{dB}$



Fig. 4. The source localization error variation at different source positions within the observation area. Four sensors are deployed in curvature with equal arc length separations as discussed in Section IV-B. The T = **30** °C uniform throughout the area. $V_2 = 0 \ {\rm ms}^{-1}, \ \theta_2 = 270^0, \ {\rm SNR} = 10 \ {\rm dB}$

V. RESULTS AND DISCUSSIONS

The simulation setup discussed in the Section IV is obtained using Matlab 7.9.0 (R2009b) development environment for the analysis. The effect of individual parameter is analyzed via simulations while keeping the other parameters constant. Figures 3-5 show the temperature effects on the source positioning. Initially, it is assumed that the temperature is uniform throughout the area. Results for the instances of temperature 22ºC and 30ºC are depicted in Figs. 3, 4, respectively. The maximum mean estimation errors in these incidents are less than one meter throughout the observation grid. Therefore, the increase in the temperature equally, does not affect the source positioning accuracy significantly. Then, we varied the temperatures at each node such that $T_{S1} = 25^{\circ}C$, $T_{S2} = 23^{\circ}C$, $T_{S3} = 22^{0}C$, $T_{S4} = 24^{0}C$, and execute the simulation again. Figure 5 shows that in this scenario the positioning error increases, drastically,

The wind has very significant effect on localization accuracy as depicted in Fig. 6. Rise in the wind velocity increases error in positioning especially towards the boundaries of x axis. In



Fig. 5. The source localization error variation at different source positions within the observation area. The sensors are deployed in curvature with equal arc length separations as discussed in Section IV-B. The Temperature varies between each sensor path. Ts1 = 25 °°C, Ts2 = 23 °°C, Ts3 = 22 Ts4 = 24 °°C, V_2 = 0 ms^{-1}, \theta_2 = 270°, SNR= 10 dB



Fig. 6. The source localization error variation at different source positions within the observation area for wind velocities $V_2 = 5 \text{ ms}^{-1}$, $V_2 = 10 \text{ ms}^{-1}$, $V_2 = 20 \text{ ms}^{-1}$. Four sensors are deployed in curvature with equal are length separations as discussed in Section IV-B. The $T = 22 \ ^0C$ uniform throughout the area. $\theta_2 = 270^0$, SNR = 10 dB.

lower wind velocities, perhaps the error margin of ± 20 m at 500 m might not exceed; however, inaccuracy at higher velocities cannot be tolerated. Faster wind velocities toward the sensors, will increase the sound velocity while decreasing the time-of-flight (TOF) of signal from source to the sensor. The TOF is higher when the source is far away from the sensors. Therefore, the error in time delays increase with the distance resulting higher positioning errors at far away source locations from the sensors.

The SNR has much more considerable effect on the source localization accuracy in our algorithm. Figure 7 depicts the distribution of mean estimation error throughout our observation grid, while the SNR is set to -15 dB. Note that in our analysis we add the noise from uniform distribution, assuming noise is equivalent even for far away distances from the sensors. However, this assumption does not follow the actual scenario. Accoding to the discussion in Section II-A the sound intensity depletes with the distance, resulting lower SNR levels at the sensors when source is far away. Therefore, it can

118Propagation Constraints in Elephant Localization Using an Acoustic Sensor Network



Fig. 7. The source localization error variation at different source positions within the observation area. Four sensors are deployed in curvature with equal are length separations as discussed in Section IV-B. The T = 22 ^{0}C uniform throughout the area. $V_{2} = 0 \ \mathrm{ms}^{-1}, \ \theta_{2} = 270^{0}, \ \mathrm{SNR}$ = -15 dB.

be expected higher SNR levels while the elephants are closer to the sensors. The SNR level has substantial effect when the source is placed in the XY corners of the observation area. This results from inaccurate time delay information provided by correlation routine, at lower SNR levels. The inaccuracy is significant once sensors are spatially in closer to source. These erroneous time delays subsequently cause the minimization algorithm to estimate enormously inaccurate source positions as depicted in Fig. 7. The effect was further analyzed by increasing the SNR level in steps. The inaccuracy disappeared at the SNR level of -12 dB. We identified this as the marginal SNR level that is necessary for the system to provide proper localization estimation in this simulation setup.

VI. CONCLUSION

In this paper we have analyzed major propagation constraints that could affect the elephant localization framework. Our results show that the impact of the temperature for the accuracy of source localization is insignificant if the temperature is uniform over the considered area. In contrast, the temperature variation between each source-sensor path significantly affects the source positioning error. Therefore, in real implementation integrating temperature measurements to the system will vastly improve the localization accuracy. In addition to this, our results show the influence of the real time wind velocity data in to the sensor network. The SNR levels at the sensors also play major role in localization error. The system provides enormously inaccurate position estimations once the SNR level drops under certain marginal level. More precisely, this level is -12 dB with respect ur simulation framework discussed in this paper. Therefore, in actual implementation, incorporating signal processing techniques for noise reduction will increase the overall performance of the system.

ACKNOWLEDGMENT

This work was supported by the Institute for a Broadband-Enabled Society (IBES) and the Melbourne Sustainable Society Institute (MSSI), The University of Melbourne, Parkville, VIC 3010, Australia.

REFERENCES

- R. Bandara, C. Tisdell, "Comparison of rural and urban attitudes to the conservation of Asian elephants in Sri Lanka: Empirical evidence", Biological Conservation, 110 (3), pp. 327-342, 2003.
 IUCN Red List of Threatened Species, IUCN, Gland, Switzerland, 2007.
 "Tracking elephants with GPS collar", Centre for Conservation and Re-search, www.cersl.org/CCR/Programs/ElephantTracking/TrackingGPS.htm
 J. V. Wijayakulasooriya, "Automatic Recognition of Elephant Infrasound Calls Using Formant Analysis and Hidden Markov Model", 6th Inter-national Conference on Industrial and Information Systems, Sri Lanka, Ava. 2011

- national Conference on Industrial and Information Systems, Sri Lanka, Aug. 2011.
 [5] W. R. Langbauer, K. B. Payne, R. A. Charif, L. Rapaport, F. Osborn, "African elephants respond to distant playbacks of low-frequency conspecific calls", J. Exp. Biol. 157: 35-46, 1991.
 [6] B. Truax, Handbook for Acoustic Ecology, World Soundscape Project, Simon Fraser University, and ARC Publications ARC Publications, Second edition, 1999.
 [7] O. Cramer, "The variation of the specific heat ratio and the speed of sound in air with temperature, pressure, humidity, and CO₂ concentration", Journal of the Acoustical Society of America, vol. 93, pp. 2510-2516, 1993. 2516, 1993.
- Y. S. Huang, Y. P. Huang, K. N. Huang, and M. S. Young "An accurate [8] Y. S. Huang, Y. P. Huang, K. N. Huang, and M. S. Young "An accurate air temperature measurement system based on an envelope pulsed ultrasonic time-of-flight technique", Rev. Sci. Instrum. 78, 115102, 2007.
 S. M. Kay, Fundamentals of Statistical Signal Processing:Estimation Theory. Englewood Cliffs, NJ: Prentice-Hall, vol. 1, 1993.
 N. Patwari, A. O. Hero , M. Perkins, N. S. Correal, "Relative Location Estimation in Wireless Sensor Networks", IEEE transactions on signal processing, vol. 51, No. 8, Aug 2003.

Appendix B

Improving Accuracy of Elephant Localization using Sound Probes

B.1 Introduction

The content of the thesis has been partially published in an international conference and a journal. Appendix B presents the published material in international journal:

Chinthaka M. Dissanayake, K. Ramamohanarao, M. N. Halgamuge, and B. Moran, "Improving Accuracy of Elephant Localization using Sound Probes", Applied Acoustics, Volume 129, pp 92–103, 2018

B.2 Published manuscript

Applied Acoustics 129 (2018) 92–103 Contents lists available at ScienceDirect



Applied Acoustics

journal homepage: www.elsevier.com/locate/apacoust



Improving accuracy of elephant localization using sound probes

ABSTRACT

CrossMark

Chinthaka M. Dissanayake^{a,*}, Ramamohanarao Kotagiri^b, Malka N. Halgamuge^c, Bill Moran^d

^a Department of Infrastructure Engineering. The University of Melbourne, VIC 3010, Australia ^b Department of Computer Science and Software Engineering, The University of Melbourne, VIC 3010, Australia ^c Department of Electrical and Electronic Engineering, The University of Melbourne, VIC 3010, Australia ^d School of Electrical and Computer Engineering, RMIT University, VIC 3000, Australia

ARTICLE INFO

Article history: Received 5 June 2017 Accepted 11 July 2017

Kevwords: TDOA estimation Elephant localization Sound speed estimation Vocalization detection Localization of elephants in the vicinity of villages is an important issue in mitigating human-elephant conflict. This paper proposes an inexpensive, effective and non-invasive framework that employs a sound probe technique with an acoustic sensor network to localize elephants. Incorporation of probes in our sensor network eliminates the requirement to explicitly measure temperature and wind veloc-ity for accurate determination of sound velocity. A sensor network has been built and experiments performed by replaying recorded elephant sounds under three different environmental conditions. The results overall show that the system is capable of providing remarkable accuracy under distinct wind and temperature conditions. An identical experimental set up was used to localize wild elephants in Sri Lanka. Our approach enabled localization of wild elephants at a distance of over 500 m from the sensors to within 30 m, providing adequate time for the villages to take appropriate safety measures.

© 2017 Elsevier Ltd. All rights reserved.

1. Introduction

Changes in social and ecological conditions precipitated by human needs are resulting in serious depletions in certain animal populations. Human-elephant conflict (HEC) is one such problem resulting in deaths of members of both species. Humans are clear-ing large areas of land for food production, increasing pressure on traditional elephant habitats. This results in migration of elephants to villages to fulfil their food requirements and ensuing HEC. In Sri Lanka, HEC has is causing more than 60 human deaths and over 200 killings of elephants, annually [1]

Various strategies have been implemented to mitigate HEC. These include implementing electric fencing systems and using satellite imaging and radio and Global Positioning Systems (GPS) collars to detect the presence of elephants [2]

Electrified wire fences are used to restrain elephants into the forest areas. This is an expensive option due to high installation and maintenance cost. In practice, the long-term success with anti-elephant fences has often fallen well below expectation. This is basically because of deficiencies in meeting the considerable demands of meticulous routine maintenance. In addition, wild elephants have low resistance to these barriers as they have learned

http://dx.doi.org/10.1016/j.apacoust.2017.07.007 0003-682X/© 2017 Elsevier Ltd. All rights reserved

to demolish the fences using tree branches and thereby enter into the protected areas [3].

Tagging elephants with radio and GPS collars for understanding the behavioural aspects is presently conducted in many parts of the world. However, this methodology is impractical for realtime monitoring of wild elephants due to the battery power limi-tations in collar systems and the cost of GPS data retrieval. In addition, capturing elephants and fitting GPS collars to these animals is a complex and dangerous procedure which risks lives of both elephants and personals involved. Therefore, it is in practice even impossible to collar elephants which are deemed problematic.

With the advancements in technology, satellite imaging is also proposed as a methodology for tracking wildlife [4]. However, thick vegetation and environmental factors hinder this sophisticated and expensive technology for real-time monitoring of wild elephants

Recently, interest has grown in the detection of elephants through their low-frequency calls, commonly referred to as "rumbles" [5]. This technique is a safe, practical and non-intrusive methodology to detect these highly social animals that use rumbles for long distance communication. Once the elephants are detected at a sufficient distance from the village boundaries countermeasures such as the use of firecrackers, broadcasting of noise through loudspeakers, and warnings to the local population can be deployed.

^{*} Corresponding author. E-mail address: c.dissanayake@pgrad.unimelb.edu.au (C.M. Dissanayake)

C.M. Dissanayake et al. / Applied Acoustics 129 (2018) 92-103

However, the variations in environmental conditions seriously affect the performance of such systems [6,7]. This is due to the sound speed dependency on several environmental phenomena such as temperature, water vapour mole fraction, atmospheric pressure and CO_2 content in the air. Cramer [8] has proposed a closed form equation for the relationship between speed of sound and the above environmental factors through a laboratory experiment. However, the speed of sound mainly depends on temperature and effects of other variables are insignificant. The speed of sound in air can be expressed in terms of temperature as $C_{Sound_T} = C_{Sound_0} \sqrt{\left(1 + \frac{T}{273.15}\right)}$ where $C_{Sound_0} = 331.45 \text{ ms}^{-1}$ is sound speed at 0 °C [9]. In addition, wind causes to movement of the propagation medium which effectively alters the propagation sound speed between sound source and the receivers [10].

In [6] the possibility of using an acoustic sensor network to detect and localize elephants has been extensively analysed. Wind and temperature variations are identified as the main cause for the deterioration of the performance of the above approach [6,7]. For instance, a uniform wind velocity of 20 ms^{-1} results in about a 100 m (20%) error at a distance of 500 m. An additional temperature variation of 4 °C can result in an increase in the error to over 175 m (35%) [6].

In this paper we propose a technique to correct the errors arising from variations in sound speeds due to environmental effects. We propose a system to implement the algorithms that have been developed and describe the results of experiments, under three different environmental profiles, using a sensor network system that we designed. Then, we implement our system in a village area in Sri Lanka and test it for the real application of wild elephant monitoring. Our contribution is to present an effective solution that significantly improves localization accuracy over a conventional acoustic sensor network, and which does not require explicit wind and temperature sensors to compensate for the environmental effects on sound speed.

The remainder of the paper is organized as follows. In Section 2 we summarize the localization algorithm that will be utilized in our proposed system. In Section 3 the technique used to correct the speed of sound to mitigate temperature and wind velocity effects is described in detail. In Section 4 we describe the corresponding experimental setup for tests we conducted in open field and forest environment. Section 5 outlines the experimental results and discusses them further. Section 6 concludes the paper.

2. Localization algorithm

We consider an acoustic sensor network with N sensors that listen to elephant vocalizations. In consideration of the long propagation range of infrasonic rumble signals [11], it is assumed that all sensors receive an attenuated and noise corrupted replica of a call generated by an elephant (source). Let the time of arrival measurements at sensor S_i be t_i where $i = 0, 1, \dots, N - 1$. The infrasonic acoustic sensors are deployed at predetermined coordinates (x_i, y_i) . It is necessary to locate the source position (x_s, y_s) . The arrival time measurements at sensor S_i can be modeled by $t_i = T_0 + R_i/c_i + \epsilon_i$ where T_0 is the signal emitted time by the source, R_i is the distance between source and the *i*th sensor, c_i is the sound propagation speed and ϵ_i is the normally distributed measurement noise with zero mean and σ^2 variance. It should be noted that the c_i between the source and each sensor could vary because of environmental effects such as wind and the temperature. The distance from source to *i*th sensor R_i is given by $R_i = \sqrt{(x_s - x_i)^2 + (y_s - y_i)^2}$. The time T_0 of signal emission is

unknown, and so it is appropriate to perform a time-differenceof-arrival (TDOA) estimation. TDOA information is obtained by performing pair-wise cross-correlations between the observation sound signals at each sensor pair. TDOA measurement between the *p*th and *q*th sensors denoted by Δt_{pq} is given by $\Delta t_{pq} = t_p - t_q + \epsilon_{pq} = R_p/c_p - R_q/c_q + \epsilon_{pq}$, where $p = 0, 1, \dots, N-2$ and $q = p + 1, p + 2, \dots, N-1$, $\epsilon_{pq} = \epsilon_p - \epsilon_q$ (see Fig. 1).

$$\mathbf{m} = [m_{2,1} \quad m_{3,1} \quad \cdots \quad m_{n,1}]^T = \mathbf{d} + \boldsymbol{\epsilon}$$

$$\mathbf{d}(\theta) = \begin{bmatrix} r_{2,1} & r_{3,1} & \cdots & r_{n,1} \end{bmatrix}$$

and $r_{p,q} = R_p - R_q$. The probability density function [12] of **m** given θ is

$$f(\mathbf{m}/\theta) = 2\pi^{-n/2} (\det \mathbf{Q})^{-1/2} \exp\{-J/2\}$$

where

$$J = [\mathbf{m} - \mathbf{r}(\theta)]^T \mathbf{Q}^{-1} [\mathbf{m} - \mathbf{r}(\theta)].$$

The Maximum Likelihood Estimate (MLE) is the $\theta(x_s, y_s)$ that minimizes J. This can be reduced to minimizing the non-linear least square problem [13] of

$$f(x_{s}, y_{s}) = \arg\min_{x_{s}, y_{s}} \sum_{p,q} (h_{pq} - g_{pq})^{2}.$$
 (1)

where $h_{pq} = c \times \Delta t_{pq}$ and $g_{pq} = R_p - R_q$. The solution to (1) gives the unknown source location. However, to locate the source accurately the above system needs the correct sound propagation speeds between the source and each sensor, which is affected by wind and temperature variations. Our approach to estimate sound speed with mitigated temperature and wind velocity effects is presented in the next section.

3. Average sound speed estimation methodology

Our goal is to estimate the approximate speed of sound that signals propagate with respect to each sensor. In order to build a solution that attains the above sound speeds in a real world $% \left({\left[{{{\mathbf{x}}_{i}} \right]}_{i}} \right)$ environmental scenario we first develop a model assuming uniform environmental conditions. Then the model is extended to real world comparable, non uniform and totally unknown environmental conditions

To locate the elephants accurately, acoustic sensors and some specific sound (chirp) signal generators are deployed at predetermined location coordinates. These sound generators are referred to as "probes" in the remainder of the paper. The chirp signal generation time schedule is predefined and known to both devices. Therefore, by listening to the signal, each sen-sor can determine the time of flight (TOF) from probes to the sensor. Then, the approximate speed of sound between the source and the sensors is estimated in two different models as presented below.

Model A: The temperature over the area is assumed to be uniform but unknown. The wind speed and the direction over the area are assumed to be uniform but unknown.

Here we estimate the wind information and the uniform temperature over the area completely depending on the probe tech-nique without incorporating additional sensors. Consider that the *i*th sensor (S_i) receives the signal from the probes P_i , P_m , P_n as depicted in Model A in the Fig. 2. The temperature affected speed of sound between S_i and P_z is $V_{P_z S_i}(T)$ where z = l, m, n, and, so the measured speed of sound $V_{P_z S_i}$ between S_i and P_z is given by

93



Fig. 1. N number of sensors are deployed at the vicinity of the expected elephant location. The distance difference between source and the pth and qth sensor is defined as g_{pq}.



Fig. 2. The ith sensor (S_i) receiving the signal from the probes P_i, P_m, P_n . The wind flows with speed of V_w with an angle of θ_w with respect to the horizontal axis. The temperature affected speed of sound between P_z is $V_{P,S_i}(T)$ where z = l, m, n.

 $V_{P_iS_i} = V(T) + V_w \cos(\theta_w - \theta_{P_iS_i})$. Thus, by resolving the system of equations for each probe *z* and sensor S_i , it is possible to estimate the approximate speed of sound in the area by evaluating three unknowns (wind speed V_w , wind direction θ_w , and temperature affected speed of sound V(T)). Now the wind direction can be derived as

$$\theta_{\rm w} = \tan^{-1} \left[\frac{A \sin(\beta) - B \sin(\alpha)}{A \cos(\beta) - B \cos(\alpha)} \right]$$
(2)

where

$$A = (V_{P_l S_i} - V_{P_m S_i}) \sin \left((\theta_{P_n S_i} - \theta_{P_l S_i})/2 \right),$$

$$B = \left(V_{P_l S_i} - V_{P_n S_i}\right) \sin\left(\left(\theta_{P_m S_i} - \theta_{P_l S_i}\right)/2\right),$$

 $\alpha = (\theta_{P_m S_i} + \theta_{P_i S_i})/2$, and $\beta = (\theta_{P_n S_i} + \theta_{P_i S_i})/2$. The wind speed V_w is now

$$V_{\rm w} = \frac{V_{P_i S_i} - V_{P_{\rm m} S_i}}{\cos(\theta_{\rm w} - \theta_{P_i S_i}) - \cos(\theta_{\rm w} - \theta_{P_{\rm m} S_i})}.$$
(3)

By using (2) and (3) we obtain the temperature affected speed of sound V(T) from

$$V(T) = V_{P_i S_i} - V_w \cos(\theta_w - \theta_{P_i S_i}). \qquad (4)$$

The temperature and the wind effects compensated sound velocity can then be determined.

Model B: The temperature over the area is non-uniform and unknown. The wind speed and the direction over the area are also non-uniform and unknown.

Under the condition where wind and temperature are nonuniform and information is unavailable, it is difficult to produce closed form equations to correct the actual sound propagation speed, because not enough information is available. In this case real world spatial properties of the temperature and wind variations were incorporated. The assumption of slow spatial variation of temperature permits temperatures in adjacent regions to be considered as correlated random variables. We model this variation, therefore, as a two-dimensional Gaussian function. The temperature at particular location coordinates (x, y) is given by $f(x, y) = T_{min} + \Delta T_{Vmax} \exp\left(-\left(\frac{(x-x_y)^2}{2\sigma_x^2} + \frac{(y-y_y)^2}{2\sigma_y^2}\right)\right)$, where T_{min} is the minimum temperature, ΔT_{Vmax} is the maximum temperature variation, (x_0, y_0) is the centre coordinates of the interested area, σ_x, σ_y are the x and y spreads of the temperature blob over the area. In addition, the wind flow is assumed to be a Markov Random Field (MRF). The wind vector at a particular position is determined by values drawn from a normal distribution $\mathcal{N}(\mathbf{Vwndj}, \sigma_w^2)$ where

94
\mathbf{V}_{wAdj} is the wind vector in adjacent locations. Now, the probe technique is incorporated to evaluate the approximated actual sound propagation by dividing the interested area into effective isothermal spatial regions. Validity of the above approximation depends upon selection of the spatial region. The smallest region to uniquely estimate the environmental parameters is chosen as the isothermal region. This region is the area depicted as Model A in Fig. 2. The set of equations derived in the Model A can be used to estimate the average sound speed correction within the effective isothermal areas. Sound signals to distinct sensors will experience different speeds of propagation as affected by wind and temperature variations. By efficiently averaging them considering the spatial proximity, approximation to the effective sound speed can be obtained.

4. Experimental setup

4.1. Open field experiment

The hardware system consisted of sensor units which contain microphones (Audio-Technica AT897) and FM transmitters (Fmuser SDA-01A) and sound probes which include chirping devices (see Figs. 3 and 4). Separate laboratory experiment was conducted to investigate the frequency response of the signal sensor microphones detection system. It was confirmed that system performs well in the range of approximately 10–400 Hz and can capture main signal energy components of low frequency elephant calls as shown in [5].

The sensors listen to the sound signals and transmit them in a dedicated FM channel to the centralized receiver in real time. Each sensor was assigned a distinctive FM channel to prevent the interference at the receiver. These channels were carefully selected to avoid interference with the other FM transmissions such as local radio stations and communication within police



Fig. 3. Snap of complete sensor unit deployed in the field. The microphone and the transmitter units are mounted on a tripod and units are protected with weather shields. Four sets of similar sensor units are deployed for the experiment. The transmitter units were powered by 12 V DC battery.



Fig. 4. Snap of Probe unit deployed in the field. The unit consists of sound chirp generator and a controller unit which determines the timing of chirp generation. The unit was powered by a 12 V DC battery.

forces and emergency services. The centralized receiver consisted of a set of FM receivers (Sony CRSW08) which are synchronized to each sensor unit FM frequency. The outputs of these receivers were then streamed into a laptop computer via a USB sound card hub which enabled simultaneous signal recording. Sound signal acquisition is attained through a commercial software program named Cool Edit pro Version 2.0. The signals from each sensor channel was recorded into separate tracks in the software program and subsequently saved and exported into wave files for further analysis. The main feature of the software program that facilitated in our experiment is that it enables synchronized data recording from separate channels. Table 1 provides a cost analysis of the hardware system that utilized in this implementation. The overall cost of the system including a cheap processor is less than \$2750.

Our ultimate goal is to utilize the system to localize and track the forest elephants through their rumble signals. In our experiment, we decided to utilize a moving sound source which produces elephant rumble signals such that we can maintain the analogy up to some extent. First, we included rumble signals into a sound player (Sansa® Express[™]) which consisted of a storage device and connected it to a large speaker through an amplifier (TOA[™] A-50 M). The speakers had frequency response of $\pm 3d$ B over the range 16–250 Hz and nearly all the elephant calls used for the simulation had acoustic energy within the above range. Then the system is mounted on a wheel cart so that it can easily be moved in the field. We called this system setup as "Surrogate Elephant" (SE) (see Fig. 5).

The experiment site was a playground named Weeguluwathte at the city of Gampola which is located in the Central Province of Sri Lanka. The experiment was performed in three sets during dif-

 Table 1

 Cost analysis of the hardware system utilized in the real world implementation

Item	Brand	Model	Unit price	No. units	Cost
Acoustic sensor	Audio-Technica	AT897	\$212.00	4	\$848.00
Long range FM Transmitter	Fmuser	SDA-01A	\$112.00	4	\$448.00
Receiver	Sony	CRSW08	\$48.00	4	\$192.00
Probe	Custom design	Custom design	\$40.00	4	\$160.00
Battery	B.U.E	3FM-4.0	\$25.00	8	\$200.00
Recorder software	Cool Edit	Pro V2.0	\$300.00	1	\$300.00
Processor unit (Laptop)	Toshiba	SATELLITE C50D	\$499.00	1	\$900.00
Interfacing units and connectors	N/A	N/A	\$100.00	1	\$100.00
Total					\$2747.00



Fig. 5. Picture of the surrogate elephant which is being moved by one of the assistants. The unit consists of sound player connected to a speaker through an amplifier. The system was designed to play previously recorded elephant rumble signals during the experiment.

Table 2

Average wind and temperature measurements during the experiment trials.	

Experiment set	Average temperature (°C)) Average wind speed (ms ⁻¹)	
Data Set 1	27.4	6.9	
Data Set 2	29.3	8.7	
Data Set 3	28.1	10.1	

ferent time periods of the day such that natural variation of temperature and wind velocities can be taken into account. The experiment was carried out in a sunny day and humidity was 64%. The average temperature and wind speed with respect to north was obtained using an anemometer (Bentech GM816) during each set and is summarized in Table 2.

The sensors and the probes were deployed in the field as per the reasoning explained in the simulations. The exact locations of the probes and sensors were acquired using eTrex^{IM} GPS device and recorded. In addition, the distance between each sensor and probe was accurately measured via Leica DISTO^{IM} A8 Laser distance meter. Once the probe sensor system was deployed and

activated, the centralized data recorder system was positioned in a convenient location to start the recording. Then a GPS data recorder was attached to our Surrogate Elephant (SE) to obtain the actual path where it was travelling. Subsequently, SE is moved at a constant speed towards the deployed probe sensor direction whilst recording sound signals into the centralized data recording system.

4.2. Real forest experiment

Following the open field implementation and identifying the possible challenges, we proceeded to test our methodology in real forest environment. The experimental site was a village called Swaranapaligama which was situated in North central province, Sri Lanka. The village is located adjacent to the border of the Wasgamuwa National Park and a branch of Amban River flows close to the village. The primary occupation of the villages is farming and they mainly cultivate rice and vegetables. As a result of close proximity to the forest and the continual production of crops, wild elephants frequently enter into the village especially during the dry season. Considering the above facts, we conducted our experiment in the end of September 2013 which is the latter period of the dry season.

A similar experimental setup as in the open field experiment described in Section 4.1 was employed in this implementation as well. A set of sensors and probes were deployed at predetermined locations. The probes were kept in the forest area to obtain a better representation of the sound characteristics through the woodland. The sensors are kept close to the village boundary depending on practical considerations. A simplified block diagram of the hardware setup is depicted in Fig. 6. The probe and the sensor coordinates were recorded using eTrexTM GPS devise as in the previous experiment and those coordinates are summarized in Table 3.

During the background information gathering the villagers informed us that in many occasions, elephants come closer to the village shortly after the nightfall. Therefore, we set up equipment targeting to start the data collection during the above period. Owing to constrains in battery life and available resources, successful data collection was done in three consecutive days and typical weather information during the experiments is summarized in Table 4.

Once the experiment was conducted, in the following day we walked through the elephant tracks to verify the presence of elephants during the previous day. The verification process was practically impossible to conduct in real time during the experiment due to safety issues. However, through the evidence such as fresh piles of elephant dung, broken tree branches, signs of browsing grass and foot prints embossed in the ground, we were able to determine the presence of the elephants during the days we conducted the experiment (see Figs. 7 and 8).



Fig. 6. Simplified block diagram of the system hardware implementation. Each sensor consists of microphone and a FM transmitter. Vocalization data and the probe signals acquired through each microphone is transmitted through a dedicated FM channels to the centralized receiver. Probes are placed in the forest area to get a better representation of the real sound speeds of the observation area.

Table	3

Sensor and probes location coordinates at the real forest experiment	ıt.
--	-----

	GPS coordinates	
	Longitude	Latitude
Sensor 1	80.954352	7.863963
Sensor 2	80.953244	7.864375
Sensor 3	80.952465	7.864156
Sensor 4	80.951085	7.864182
Probe 1	80.953736	7.863446
Probe 2	80.952939	7.863341
Probe 3	80.952146	7.863349
Probe 4	80.951457	7.863455

We obtained the assistance of few villages who were experts in identifying elephant tracks and familiar with the forest area closer to the village to accomplish the above task. We recorded the coordinates of the elephant tracks using eTrexTM GPS device. In addition, GPS coordinates and the description of the signs were manually recorded at the specific locations where we identified as elephants were present. Through these field visits we identified that, out of three days we conducted the experiment, in two days elephants have travelled through the tracks we observed.

Table 4

Weather details during the real forest experiment

5. Results and discussions

5.1. Open field experiment results

The results of the practical implementation of our proposed methodology in an outdoor environment are depicted in the Figs. 9–11 for three different data collection sets. It can be identified that our proposed technique clearly improves the localization accuracy of the sound source under different environmental conditions. The accuracy improves whilst the source is close to the probes. The reason for this is that, when probes and SE are in close proximity, the probes technique represents the actual sound speeds between source-sensor paths. Due to limited space and narrow variation of temperature and wind velocities in the experimental area, the localizations without probe technique correction also follow the path to a certain extent. However, this fact is invalid in a real forest application where environmental condition variation of use much more complicated.

The elephants have a behavioral characteristic of walking in the same particular paths which they have followed throughout the generations. Therefore, it is possible to incorporate these path data as a prior information whilst implementing tracking algorithms. We have implemented a particle filter considering prior information of the path data of the surrogate elephant whilst incorporating probe correction in our posterior analysis. As shown in Figs. 9–11

Experiment trial	Weather	Average temperature (°C)	Average wind speed (ms ⁻¹)	Humidity (%)
Day 1	Sunny	29.5	4.2	64
Day 2	Sunny	29.3	8.4	72
Day 3	Sunny	28.1	7.1	68



Fig. 7. A snap shot of recording information on a location where elephants were present in a day that we conducted the experiment. GPS coordinates and information on the observation were recorded manually for further reference.



Fig. 8. Picture of foot print embossed in the ground of an elephant that visited the area in the day which we conducted the experiment. Similar observations were made on fresh pile of elephant dung, broken tree branches, and signs of browsing grass and the observation locations were recorded for further reference.

the outcome of the implementation clearly indicates an accuracy improvement.

5.2. Real forest experiment results

Fig. 12 depicts the localization results from gathered data during the real forest experiment. The analysis revealed that elephant calls were recorded during only two days of the experiment. This is in fact verified through the observations made during the post experimental visits through elephant tracks. The probe technique has performed efficiently in this heterogeneous environment as well. Evidently, it surpasses the observation made without considering the wind and temperature variations. Due to large observation area and variations in the environmental facts, the remarkable performance of the probe technique can be clearly identified. Our technique has located elephants within 30 m to the elephant tracks whilst error in conventional methodology extends over 70 m. The evidence gathered indicates the real localization accuracy attained through our technique. In addition, the results indicate that under the forest environmental condition, the performance of a conventional localization system can be significantly impacted even under a modest temperature and wind velocity variations.

Our system was able to localize elephants over 500 m margin from the village boundary. This implies that, with some modifications to the hardware system such as integrating additional amplifiers the system may perform well above this distance. This fact suggests that similar system can be utilized to monitor elephants noninvasively for behavioural aspects. In a broad sense, our system which costs around \$2750 can be utilized to protect a village area of around 2 km². However, the system needs to be modified for continuous power requirement in an alerting system. The challenge of battery power constrains can be overcome incorporating hybrid solar and battery power system. Similar methodology use in natural disaster notification which uses Short Message Service (SMS) alerting system [14] can be utilized for warning villagers regarding elephant threats.

In [7], acoustic monitoring of elephants is mentioned as a valuable noninvasive research tool. However, the predicted area of detection depends upon the atmospheric acoustic state. This acoustic state is mainly affected by meteorological effects such as temperature and wind velocity. It is necessary to have real-time knowledge of atmospheric condition to accurately locate the calling elephant [7]. Our proposed methodology addresses exact same problem and adjust the sound speed for atmospheric bias which results greatly improved localization accuracy.

Mayilvaganan et al. have proposed direction of arrival estimation methodology with partial hyperbolic circular array to localize elephants [15]. They also identified that wind velocity and temperature variation significantly affects the localization accuracy. In fact, their analysis shows that localization error could be more than 28 m under narrow wind condition of 2.2 ms⁻¹ whilst four sensors and the source is deployed in a 100 m × 100 m small-scale area. In comparison, analysis results indicate that our methodology performs far better than the above approach in locating elephants even under much stronger wind conditions.

The implemented hardware system is an economical approach for actual elephant localization system. By means of a centralized data collection system, we were able to eliminate the clock synchronization problem which is a main concern in distributed data collection schemes in similar applications. In addition, the inexpensive and effective methodology of utilizing probe technique remarkably increases the localization accuracy. The sound signal at the receiver itself models the propagation

The sound signal at the receiver itself models the propagation channel characteristics. In our proposed technique this phenomenon is effectively utilized to estimate the sound propagation speed which is affected by temperature and wind variations. Further, information from sensors and probes which represent different spatial regions evidently assists for better sound speed estimation. Besides, the remarkable advantage of this approach is that it does not need both wind and temperature sensors in the network. In addition, we make use of the existing acoustic sensors for sensing probe signals and the elephant calls.



(a) The average temperature during the experiment was 27.4 $^{\circ}$ C and average wind velocity 6.9 ms⁻¹ during the experiment. The estimated localization positions of the Surrogate Elephant closely follow the actual path whilst the probe technique correction was made.



(b) Surrogate Elephant localization error estimation for Data Set 1. The error is calculated deviation in Y direction from the actual path. Note that the ES is moved towards the sensors and 1st data point represent the furthest point from the sensors.

Fig. 9. Localization of Surrogate Elephant; the analysis results from the Data Set 1 acquired from the real experiment.





(a) The average temperature during the experiment was 29.3 $^{\circ}$ C and average wind velocity 8.7 ms⁻¹ during the experiment. The estimated localization positions of the Surrogate Elephant closely follow the actual path whilst the probe technique correction was made.



(b) Surrogate Elephant localization error estimation for Data Set 2. The error is calculated deviation in Y direction from the actual path. Note that the ES is moved towards the sensors and 1st data point represent the furthest point from the sensors.

Fig. 10. Localization of Surrogate Elephant; the analysis results from the Data Set 2 acquired from the real experiment.



(a) The average temperature during the experiment was 28.1 $^{\circ}$ C and average wind velocity 10.1 ms⁻¹ during the experiment. The estimated localization positions of the Surrogate Elephant closely follow the actual path whilst the probe technique correction was made.



(b) Surrogate Elephant localization error estimation for Data Set 3. The error is calculated deviation in Y direction from the actual path. Note that the ES is moved towards the sensors and 1st data point represent the furthest point from the sensors.

Fig. 11. Localization of Surrogate Elephant; the analysis results from the Data Set 3 acquired from the real experiment.

102





(a) The analysis result of the data collection from the Day 1 in real forest experiment. The average temperature was 29.5° C and average wind velocity was 4.2 ms^{-1} during the experiment.



(b) The analysis result of the data collection from the Day 3 in real forest experiment. The average temperature was 28.1°C and average wind velocity was 7.1 ms⁻¹ during the experiment.

Fig. 12. Analysis results of the real forest experiment. Observations were associated to different tracks in Day 1 and Day 3. The estimated elephant localization positions closely follow the elephant tracks. Evidently, the probe technique has significantly improved the localization accuracy despite the complex terrain.

6. Conclusion

In this paper we have investigated the feasibility of enhancing the accuracy in elephant localization system by utilizing a sound generating probe technique. An inexpensive hardware platform is implemented to evaluate the proposed methodology in an outdoor environment under equivalent elephant localization scenario. Finally, the system was deployed in a real forest environment to localize wild elephants in Sri Lanka. The outdoor experimental implementation verified the significant improvement in accuracy whilst using the probe technique, in a surrogate elephant localization system. Further, the real forest implementation successfully

localized elephants over 500 distance. Moreover, our methodology enabled to localize wild elephants within 30 m accuracy to their natural walking paths in the real forest implementation.

Acknowledgment

This work was supported by the Institute for a Broadband-Enabled Society (IBES) and the Melbourne Sustainable Society Institute (MSSI), The University of Melbourne, Parkville, VIC 3010, Australia. Authors would like to express their gratitude to Dr. M. I. Bandara, Mr. K. Dissanayake, Mr. S. Wijerathne, Mr. P. Wijerathne, Mr. B. M. Dissanayake, the supporting team and the villagers from the Swarnapali Gama for their support during the data collection procedure.

References

- [1] Elephant conservation <www.ccrsl.org/CCR/Programs/ElephantConservation.
- htm>.
 [2] Galanti V, Tosi G, Rossi R, Foley C. The use of GPS radio-collars to track elephants (loxodonta africana) in the tarangire national park (tanzania). Hystrix, Ital J Mammal 2000;11(2).
- Hystrix, Ital J Mammal 2000;11(2).
 Hoare R. Fencing and other barriers against problem elephants; 2003 <www.
- iucn.org/afesg>. [4] Gutro R. Satellite data to track wildlife: elephants in space, 17 January 2005 <http://www.nasa.gov/vision/earth/lookingatearth/lelephants_space.html>.

- [5] Wijajakulasooriya JV. Automatic recognition of elephant infrasound calls using formant analysis and hidden markov model. In: 6th IEEE International Conference on Industrial and Information Systems (ICIIS). IEEE; 2011. p. 244–8.
 [6] Dissanayake CM, Kotagiri R, Halgamuge MN, Moran B, Farrell P. Propagation constraints in elephant localization using an acoustic sensor network. In: IEEE 6th international conference on information and automation for sustainability (ICIARS). p. 101–5. Intre/Ick.doi.org/10.1109/ICIAFS.2012.6419889.
 [7] Carstang M, Fitzjarrald DR, Fristrup K, Brain C. The daily cycle of low-frequency elephant calls and near-surface atmospheric conditions. Farth Interact 9(1).
 [8] Cramer O. The variation of the specific-heat ratio and the speed of sound in air with temperature, pressure, humidity, and Co₂ concentration. J Acoust Soc Am 93(5):2510–6.
 [9] Huang YS, Huang YP, Huang KN, Young MS. An accurate air temperature measurement system based on an envelope pulsed ultrasonic time-of-flight technique. Rev Scient Instrum 78 (115102).
 [10] Chen Zhixim M R. KImospheric sound propagation considerations for the birdstrike project; 2004.
 [11] Langbauer WR, Payne KB, Charif RA, Rapaport L, Osborn F. African elephants report to distant playbacks of low-frequency conspecific calls. J Exp Biol

- respond to distant playbacks of low-frequency conspecific calls. J Exp Biol 157:35-46.
- [12] Trees HLV. Detection, estimation, and modulation theory: radar-sonar signal
- Irees HLV. Detection, estimation, and modulation theory: radar-sonar signal processing and gaussian signals in noise. Krieger Publishing Co., Inc.; 1992.
 Shen G, Zetik R, Thoma RS. Performance comparison of ToA and TDOA based location estimation algorithms in LOS environment. In: 5th Workshop on Positioning, Navigation and Communication, 2008, WPNC 2008. IEEE, p. 71–8.
 Jayasinghe G, Fahmy F, Gajawera N, Dias D, A CSM alarm device for disaster early warning. In: First international conference on industrial and information
- Systems, IEE, p. 383–7. Mayilvaganan M, Devaki M. Elephant vocalization direction of arrival estimation and external factors affecting wave propagation using acoustic sensor network. Int J Innov Res Comp Commun Eng 2013;1:789–94. [15]

Bibliography

- G. Schwammer, "Austrian Sri Lankan elephant research and conservation (aserc) project", ZOOs' Print, vol. 21, no. 8, p. 21, 2006.
- [2] SLWCS, "Sri Lanka wildlife conservation society", 2018. [Online]. Available: http://www.slwcs.org/
- [3] R. Bandara and C. Tisdell, "Comparison of rural and urban attitudes to the conservation of Asian elephants in Sri Lanka: empirical evidence", *Biological Conservation*, no. 3, p. 327, 2003.
- [4] "Annual performance report of the wildlife conservation department 2016", Department of Wildlife Conservation Sri Lanka, Report, 2016.
- [5] IUCN, "IUCN red list of threatened species", International Union for Conservation of Nature, UK, 2007.
- [6] V. Galanti, G. Tosi, R. Rossi, and C. Foley, "The use of GPS radio-collars to track elephants (Loxodonta Africana) in the Tarangire national park (Tanzania)", *Hystrix*, *the Italian Journal of Mammalogy*, vol. 11, no. 2, 2000.
- [7] J. V. Wijayakulasooriya, "Automatic recognition of elephant infrasound calls using formant analysis and hidden markov model", in 2011 6th IEEE International Conference on Industrial and Information Systems (ICIIS),. IEEE, Conference Proceedings, pp. 244–248, 2011.

- [8] C. Santiapillai, S. Wijeyamohan, G. Bandara, R. Athurupana, N. Dissanayake, and B. Read, "An assessment of the human-elephant conflict in Sri Lanka", *Ceylon Journal of Science (Biological Sciences)*, vol. 39, no. 1, 2010.
- [9] "Tracking elephants with GPS collar", 2018. [Online]. Available: http://www. ccrsl.org/Programs/tracking-elephants/with-GPS-collars
- [10] P. J. Clemins, "Automatic classification of animal vocalizations", Faculty of the Graduate School, Marquette University, Thesis, 2005.
- [11] "Habitat research", 2002. [Online]. Available: http://www.habitresearch.com/
- [12] NASA, "NASA satellite tracking of threatened species", 2017. [Online]. Available: http://sdcd.gsfc.nasa.gov/ISTO/satellite_tracking/
- [13] "Carribean conservation corporation", 2018. [Online]. Available: http://www. cccturtle.org/satwelc.htm
- [14] A. Mainwaring, D. Culler, J. Polastre, R. Szewczyk, and J. Anderson, "Wireless sensor networks for habitat monitoring", in 1st ACM international workshop on Wireless sensor networks and applications, Conference Proceedings, pp. 88–97, 2002.
- [15] P. Juang, H. Oki, Y. Wang, M. Martonosi, L. S. Peh, and D. Rubenstein, "Energyefficient computing for wildlife tracking: design tradeoffs and early experiences with Zebranet", pp. 96–107, 2002.
- [16] A. Cerpa, J. Elson, D. Estrin, L. Girod, M. Hamilton, and J. Zhao, "Habitat monitoring: application driver for wireless communications technology", SIGCOMM Comput. Commun. Rev., vol. 31, no. 2 supplement, pp. 20–41, 2001.
- [17] W. Hu, N. Bulusu, C. T. Chou, S. Jha, A. Taylor, and V. N. Tran, "Design and evaluation of a hybrid sensor network for cane toad monitoring", ACM Trans. Sen. Netw., vol. 5, no. 1, pp. 1–28, 2009.
- [18] T. Ko, S. Ahmadian, J. Hicks, M. Rahimi, D. Estrin, S. Soatto, S. Coe, and M. P. Hamilton, "Heartbeat of a nest: Using imagers as biological sensors", ACM Trans. Sen. Netw., vol. 6, no. 3, pp. 1–31, 2010.

- [19] C. Santiapillai, "Mitigation of human-elephant conflicts in Sri Lanka", Gajah, vol. 15, pp. 1–8, 1996.
- [20] D. Lahiri-Choudhury, "Keeping wild elephants at bay", The illustrated encyclopedia of elephants. Salamander Books, London, pp. 166–173, 1991.
- [21] R. Sukumar, "Elephant–man conflict in karnataka", Karnataka—State of Environment Report, vol. 85, 1984.
- [22] R. C. Morris, "Note on the use of bamboo gun rocket for scaring wild animals out of cultivation", J. Bombay Nat. Ilist.Soc, vol. 55, pp. 344–345, 1958.
- [23] C. Punchihewa, "Souvenirs of a forgotten heritage", Colombo: Department of Information of Sri Lanka, 1989.
- [24] R. Piesse, "Wildlife-proof barriers in India", Report, 1982.
- [25] F. V. Osborn and L. Rasmussen, "Evidence for the effectiveness of an oleo-resin capsicum aerosol as a repellent against wild elephants in Zimbabwe", *Pachyderm*, vol. 20, pp. 55–64, 1995.
- [26] S. Hedges and D. Gunaryadi, "Reducing human–elephant conflict: do chillies help deter elephants from entering crop fields?" *Oryx*, vol. 44, no. 1, pp. 139–146, 2010.
- [27] S. R. Noga, O. D. Kolawole, O. Thakadu, and G. Masunga, "Small farmers' adoption behaviour: Uptake of elephant crop-raiding deterrent innovations in the Okavango delta, Botswana", *African Journal of Science, Technology, Innovation and Development*, vol. 7, no. 6, pp. 408–419, 2015.
- [28] J. A. Blair and N. M. Noor, Conflict between the Malaysian elephant and agriculture. Faculty of Science Universiti Kebangsaan Malaysia, 1981.
- [29] P. W. Coopers, "Wildlife electric fencing projects in communal areas of Zimbabwe: current efficacy and future role", Zimbabwe: WWF Programme Office, 1998.

- [30] R. Hoare, "Fencing and other barriers against problem elephants", AFESG website-HEC section Technical brief series, IUCN African elephant Specialist Group, 2003. [Online]. Available: www.iucn.org/afesg
- [31] V. Perera, *Living with Wild Elephants*, vol. 01, June 2015.
- [32] R. Taylor, "A review of problem elephant policies and management options in Southern Africa", IUCN African Elephant Specialist Group Report, 1999.
- [33] F. Vollrath and I. Douglas-Hamilton, "African bees to control African elephants", *Naturwissenschaften*, vol. 89, no. 11, pp. 508–511, 2002.
- [34] L. E. King, I. Douglas-Hamilton, and F. Vollrath, "African elephants run from the sound of disturbed bees", *Current Biology*, vol. 17, no. 19, pp. R832–R833, 2007.
- [35] L. E. King, "Beehive fences as effective deterrents for crop-raiding elephants: field trials in northern Kenya", *African Journal of Ecology*, no. 4, p. 431, 2011.
- [36] L. King, M. Pardo, S. Weerathunga, T. Kumara, N. Jayasena, J. Soltis, and S. de Silva, "Wild Sri Lankan elephants retreat from the sound of disturbed Asian honey bees", *Current Biology*, vol. 28, no. 2, pp. R64–R65, 2018.
- [37] P. Fernando, "Managing elephants in Sri Lanka: where we are and where we need to be", *Ceylon Journal of Science (Biological Sciences)*, vol. 44, no. 1, 2015.
- [38] B. I. Annasiwaththa, R. Munasinghe, P. Fernando, and P. Leimgruber, "Design and development of power optimized satellite elephant collar with over the air programmability", in 2012 IEEE 6th International Conference on Information and Automation for Sustainability (ICIAfS),. IEEE, Conference Proceedings, pp. 13–18, 2012.
- [39] G. Haynes, Mammoths, mastodonts, and elephants : biology, behavior, and the fossil record. Cambridge : New York : Cambridge University Press, 1991.
- [40] A. Choudhury, D. Lahiri Choudhury, A. Desai, J. Duckworth, P. Easa, A. Johnsingh,
 P. Fernando, S. Hedges, M. Gunawardena, and F. Kurt, "Elephas maximus. the
 IUCN red list of threatened species.'e. t7140a12828813. 10.2305/iucn. uk. 2008", *RLTS. T7140A12828813. en*, 2008.

- [41] S. Blake and S. Hedges, "Sinking the flagship: the case of forest elephants in Asia and Africa", *Conservation Biology*, vol. 18, no. 5, pp. 1191–1202, 2004.
- [42] P. Leimgruber, J. Gagnon, C. Wemmer, D. Kelly, M. Songer, and E. Selig, "Fragmentation of Asia's remaining wildlands: implications for Asian elephant conservation", in *Animal Conservation forum*, vol. 6. Cambridge University Press, Conference Proceedings, pp. 347–359, 2003.
- [43] R. Sukumar, "Ecology of the Asian elephant in Southern India. i. movement and habitat utilization patterns", *Journal of tropical Ecology*, vol. 5, no. 1, pp. 1–18, 1989.
- [44] R. Sukumar, *The Asian elephant: ecology and management*. Cambridge University Press, 1992.
- [45] F. Prithiviraj, W. Eric, W. Devaka, J. L. K.A, G. Manori, and J. H. K, "Perceptions and patterns of human–elephant conflict in old and new settlements in Sri Lanka: Insights for mitigation and management", *Biodiversity & Conservation*, no. 10, p. 2465, 2005.
- [46] A. Campos-Arceiz, "Shit happens (to be useful)! use of elephant dung as habitat by amphibians", *Biotropica*, vol. 41, no. 4, pp. 406–407, 2009.
- [47] A. Campos-Arceiz and S. Blake, "Megagardeners of the forest-the role of elephants in seed dispersal", *Acta Oecologica*, vol. 37, no. 6, pp. 542–553, 2011.
- [48] J. P. Wright and C. G. Jones, "The concept of organisms as ecosystem engineers ten years on: progress, limitations, and challenges", *AIBS Bulletin*, vol. 56, no. 3, pp. 203–209, 2006.
- [49] R. M. Pringle, "Elephants as agents of habitat creation for small vertebrates at the patch scale", *Ecology*, vol. 89, no. 1, pp. 26–33, 2008.
- [50] D. Beaune, B. Fruth, L. Bollache, G. Hohmann, and F. Bretagnolle, "Doom of the elephant-dependent trees in a Congo tropical forest", *Forest Ecology and Management*, vol. 295, pp. 109–117, 2013.

- [51] P. Fernando and R. Lande, "Molecular genetic and behavioral analysis of social organization in the Asian elephant (elephas maximus)", *Behavioral Ecology and Sociobiology*, vol. 48, no. 1, pp. 84–91, 2000. [Online]. Available: https://doi.org/10.1007/s002650000218
- [52] S. de Silva, A. D. Ranjeewa, and S. Kryazhimskiy, "The dynamics of social networks among female Asian elephants", *BMC Ecology*, vol. 11, no. 1, p. 17, 2011.
- [53] J. M. Samy, "Social structure of Asian elephants (elephas maximus) in Sri Lanka", University of Maryland, College Park, Thesis, 2015.
- [54] R. Sukumar, "A brief review of the status, distribution and biology of wild Asian elephants elephas maximus", *International Zoo Yearbook*, vol. 40, no. 1, pp. 1–8, 2006.
- [55] H. Kummer, Social organization of hamadryas baboons. University of Chicago Press Chicago, vol. 89, 1968.
- [56] G. Wittemyer, I. Douglas-Hamilton, and W. M. Getz, "The socioecology of elephants: analysis of the processes creating multitiered social structures", *Animal Behaviour*, vol. 69, no. 6, pp. 1357–1371, 2005. [Online]. Available: http://www.sciencedirect.com/science/article/pii/S0003347205000667
- [57] I. D. Couzin and M. E. Laidre, "Fission–fusion populations", *Current biology*, vol. 19, no. 15, pp. R633–R635, 2009.
- [58] F. Kurt, "Remarks on the social structure and ecology of the Ceylon elephant in the Yala national park", *The Behavior of Ungulates and Its Relation to Management*, pp. 618–634, 1974.
- [59] G. M. McKay, Behavior and ecology of the Asiatic elephant in southeastern Ceylon [electronic resource] / George M. McKay, ser. Smithsonian contributions to zoology: no. 125. Washington, D.C. : Smithsonian Institution Press, 1973. [Online]. Available: http://purl.fdlp.gov/GPO/gpo29050
- [60] J. Seidensticker and B. Mundial, Managing elephant depredation in agricultural and forestry projects. World Bank, 1984.

- [61] C. Santiapillai and P. Jackson, *The Asian elephant: an action plan for its conservation*. IUCN, 1990.
- [62] R. N. Owen-Smith, Megaherbivores: the influence of very large body size on ecology. Cambridge university press, 1992.
- [63] P. Jarman, "The social organisation of antelope in relation to their ecology", Behaviour, vol. 48, no. 1, pp. 215–267, 1974.
- [64] K. McCullagh, "The growth and nutrition of the African elephant ii. the chemical nature of the diet", African Journal of Ecology, vol. 7, no. 1, pp. 91–97, 1969.
- [65] R. Sukumar, "Ecology of the Asian elephant in Southern India. ii. feeding habits and crop raiding patterns", *Journal of Tropical Ecology*, vol. 6, no. 1, pp. 33–53, 1990.
- [66] S. McNaughton, , and N. J. Georgiadis, "Ecology of African grazing and browsing mammals", Annual review of ecology and systematics, vol. 17, no. 1, pp. 39–66, 1986.
- [67] J. P. Bryant, F. D. Provenza, J. Pastor, P. B. Reichardt, T. P. Clausen, and J. T. du Toit, "Interactions between woody plants and browsing mammals mediated by secondary metabolites", *Annual Review of Ecology and Systematics*, vol. 22, no. 1, pp. 431–446, 1991.
- [68] F. Osborn, "Seasonal variation of feeding patterns and food selection by cropraiding elephants in Zimbabwe", *African Journal of Ecology*, vol. 42, no. 4, pp. 322– 327, 2004.
- [69] R. Joshi and R. Singh, "Feeding behaviour of wild Asian elephants (elephas maximus) in the Rajaji national park", *The Journal of American Science*, vol. 4, no. 2, pp. 34–48, 2008.
- [70] R. Joshi, "Asian elephant's (elephas maximus) behaviour in the Rajaji national park, North-West India: eight years with Asian elephants", *Nat. Sci*, vol. 7, no. 1, pp. 49– 77, 2009.

- [71] L. Conradt and T. J. Roper, "Activity synchrony and social cohesion: a fissionfusion model", *Proceedings of the Royal Society of London B: Biological Sciences*, vol. 267, no. 1458, pp. 2213–2218, 2000.
- [72] S. Chamaillé-Jammes, M. Valeix, and H. Fritz, "Managing heterogeneity in elephant distribution: interactions between elephant population density and surfacewater availability", *Journal of Applied Ecology*, vol. 44, no. 3, pp. 625–633, 2007.
- [73] V. Fishlock and P. C. Lee, "Forest elephants: fission–fusion and social arenas", Animal Behaviour, vol. 85, no. 2, pp. 357–363, 2013.
- [74] A. K. Turkalo, P. H. Wrege, and G. Wittemyer, "Long-term monitoring of Dzanga Bai forest elephants: forest clearing use patterns", *PloS one*, vol. 8, no. 12, p. e85154, 2013.
- [75] J. S. Brashares and P. Arcese, "Role of forage, habitat and predation in the behavioural plasticity of a small African antelope", *Journal of Animal Ecology*, vol. 71, no. 4, pp. 626–638, 2002.
- [76] A. Frid and L. Dill, "Human-caused disturbance stimuli as a form of predation risk", *Conservation Ecology*, vol. 6, no. 1, 2002.
- [77] C. M. Beale and P. Monaghan, "Human disturbance: people as predation-free predators?" *Journal of applied ecology*, vol. 41, no. 2, pp. 335–343, 2004.
- [78] J. Shoshani and P. Tassy, The Proboscidea: evolution and palaeoecology of elephants and their relatives. Oxford: Oxford University Press, 1996.
- [79] R. Nadin, M. Anna-Sapfo, L. P. Joshua, S. Montgomery, M. Paul, and H. Michael, "Proboscidean mitogenomics: chronology and mode of elephant evolution using mastodon as outgroup", *PLoS Biology*, Vol 5, Iss 8,, p. e207, 2007.
- [80] I. Douglas-Hamilton, "On the ecology and behaviour of the African elephant: the elephants of Lake Manyara", Thesis, 1972, 391727281 eng.
- [81] C. Moss and J. Poole, *Relationships and social structure in African elephants*, Primate social relationships: An integrated approach, pp. 315–325, 1983.

- [82] S. K. Sikes, "Natural history of the African elephant", 1971.
- [83] P. Lee, "Allomothering among African elephants", Animal Behaviour, vol. 35, no. 1, pp. 278–291, 1987.
- [84] P. C. Lee and C. J. Moss, "The social context for learning and behavioural development among wild African elephants", *Mammalian social learning: Comparative and ecological perspectives*, pp. 102–125, 1999.
- [85] R. M. Laws, "Elephants as agents of habitat and landscape change in east africa", *Oikos*, pp. 1–15, 1970.
- [86] R. M. Laws, I. S. Parker, and R. C. Johnstone, *Elephants and their habitats*. Clarendon Press, 1975.
- [87] E. A. Archie, J. E. Maldonado, J. A. HOLLISTER-SMITH, J. H. Poole, C. J. Moss, R. C. Fleischer, and S. C. Alberts, "Fine-scale population genetic structure in a fission–fusion society", *Molecular Ecology*, vol. 17, no. 11, pp. 2666–2679, 2008.
- [88] J. A. Hollister-Smith, J. H. Poole, E. A. Archie, E. A. Vance, N. J. Georgiadis, C. J. Moss, and S. C. Alberts, "Age, musth and paternity success in wild male African elephants, Loxodonta Africana", *Animal Behaviour*, vol. 74, no. 2, pp. 287–296, 2007.
- [89] P. Lee, J. Poole, N. Njiraini, C. Sayialel, and C. Moss, "Male social dynamics: independence and beyond", *The Amboseli Elephants: A longterm perspective on a long-lived mammal*, pp. 260–271, 2011.
- [90] J. F. Eisenberg and M. Lockhart, "An ecological reconnaissance of Wilpattu national park, ceylon", 1972.
- [91] J. L. Brown, "Reproductive endocrine monitoring of elephants: an essential tool for assisting captive management", Zoo Biology: Published in affiliation with the American Zoo and Aquarium Association, vol. 19, no. 5, pp. 347–367, 2000.
- [92] P. Fernando, E. D. Wikramanayake, H. Janaka, L. Jayasinghe, M. Gunawardena,S. W. Kotagama, D. Weerakoon, and J. Pastorini, "Ranging behavior of the Asian

elephant in Sri Lanka", *Mammalian Biology-Zeitschrift für Säugetierkunde*, vol. 73, no. 1, pp. 2–13, 2008.

- [93] J. H. Poole, "Musth and male-male competition in the African elephant", Thesis, 1982.
- [94] J. H. Poole, "Mate guarding, reproductive success and female choice in African elephants", Animal Behaviour, vol. 37, pp. 842–849, 1989.
- [95] J. F. Eisenberg, G. M. McKay, and M. Jainudeen, "Reproductive behavior of the Asiatic elephant", *Behaviour*, vol. 38, no. 3, pp. 193–224, 1971.
- [96] K. E. Evans and S. Harris, "Adolescence in male African elephants, Loxodonta Africana, and the importance of sociality", *Animal Behaviour*, vol. 76, no. 3, pp. 779–787, 2008.
- [97] E. Goodale, G. Beauchamp, R. D. Magrath, J. C. Nieh, and G. D. Ruxton, "Interspecific information transfer influences animal community structure", *Trends in ecology* & evolution, vol. 25, no. 6, pp. 354–361, 2010.
- [98] M. T. Hansen, "The search-transfer problem: The role of weak ties in sharing knowledge across organization subunits", *Administrative science quarterly*, vol. 44, no. 1, pp. 82–111, 1999.
- [99] K. McComb, C. Moss, S. M. Durant, L. Baker, and S. Sayialel, "Matriarchs as repositories of social knowledge in African elephants", *Science*, vol. 292, no. 5516, pp. 491–494, 2001.
- [100] M. L. Riedman, "The evolution of alloparental care and adoption in mammals and birds", *The Quarterly Review of Biology*, vol. 57, no. 4, pp. 405–435, 1982.
- [101] A. Roulin, "Why do lactating females nurse alien offspring? a review of hypotheses and empirical evidence", *Animal Behaviour*, vol. 63, no. 2, pp. 201–208, 2002.
- [102] L. A. Fairbanks, "Reciprocal benefits of allomothering for female vervet monkeys", *Animal Behaviour*, vol. 40, no. 3, pp. 553–562, 1990.

- [103] H. T. Dublin, "Cooperation and reproductive competition among female African elephants", Social behavior of female vertebrates, pp. 291–313, 1983.
- [104] MAIL FOREIGN SERVICE, "Elephants march through hotel lobby after it was built on their migration trail", 2 October 2008. [Online]. Available: http://www.dailymail.co.uk/news/article-1065865/ Pictured-Elephants-march-hotel-lobby-built-migration-trail.html
- [105] "Return of elehpant passes back to elephants", 2016. [Online]. Available: http://www.bbc.com/sinhala/sri-lanka-36426952
- [106] J. H. Poole, K. Payne, W. R. Langbauer, and C. J. Moss, "The social contexts of some very low frequency calls of African elephants", *Behavioral Ecology and Sociobiology*, vol. 22, no. 6, pp. 385–392, 1988. [Online]. Available: https://doi.org/10.1007/BF00294975
- [107] J. H. Poole, "Signals and assessment in African elephants: evidence from playback experiments", Animal Behaviour, vol. 58, no. 1, pp. 185–193, 1999. [Online]. Available: http://www.sciencedirect.com/science/article/pii/S000334729991117X
- [108] W. R. Langbauer, K. B. Payne, R. A. Charif, L. Rapaport, and F. Osborn, "African elephants respond to distant playbacks of low-frequency conspecific calls", *Journal of Experimental Biology*, vol. 157, pp. 35–46, 1991. [Online]. Available: (GotoISI)://WOS:A1991FM39300003
- [109] W. R. Langbauer, "Elephant communication", Zoo Biology, vol. 19, no. 5, pp. 425– 445, 2000.
- [110] M. Garstang, "Long-distance, low-frequency elephant communication", Journal of Comparative Physiology A, vol. 190, no. 10, pp. 791–805, 2004. [Online]. Available: https://doi.org/10.1007/s00359-004-0553-0
- [111] K. B. Payne, M. Thompson, and L. Kramer, "Elephant calling patterns as indicators of group size and composition: the basis for an acoustic monitoring system", *AFRICAN JOURNAL OF ECOLOGY*, no. 1, p. 99, 2003.

- [112] C. E. O'Connell, B. T. Arnason, and L. A. Hart, "Seismic transmission of elephant vocalizations and movement", *The Journal of the Acoustical Society* of America, vol. 102, no. 5, pp. 3124–3124, 1997. [Online]. Available: https: //doi.org/10.1121/1.420597
- [113] C. O'Connell-Rodwell, B. Arnason, and L. Hart, "Seismic properties of Asian elephant (elephas maximus) vocalizations and locomotion", *The Journal of the Acoustical Society of America*, vol. 108, no. 6, pp. 3066–3072, 2000.
- [114] S. Nair, R. Balakrishnan, C. S. Seelamantula, and R. Sukumar, "Vocalizations of wild Asian elephants (elephas maximus): Structural classification and social context", *The Journal of the Acoustical Society of America*, vol. 126, no. 5, pp. 2768–2778, 2009.
- [115] T. D. Rossing, *Springer handbook of acoustics.* [electronic resource]. New York: Springer, 2007.
- [116] F. Gustafsson, Statistical sensor fusion. Lund : Studentlitteratur, 2010.
- [117] H. B. Lee, "A novel procedure for assessing the accuracy of hyperbolic multilateration systems", *IEEE Transactions on Aerospace and Electronic Systems*, vol. AES-11, no. 1, pp. 2–15, 1975.
- [118] S. M. Kay, Fundamentals of statistical signal processing. 1. Estimation theory, ser. Prentice-Hall signal processing series. PTR Prentice-Hall, 2010.
- [119] B. Truax and P. World Soundscape, "Handbook for acoustic ecology", 1999.
- [120] O. Cramer, "The variation of the specific-heat ratio and the speed of sound in air with temperature, pressure, humidity, and co2 concentration", *Journal of the Acoustical Society of America*, vol. 93, no. 5, pp. 2510–2516, 1993. [Online]. Available: (GotoISI)://WOS:A1993LC57200008
- [121] Y. S. Huang, Y. P. Huang, K. N. Huang, and M. S. Young, "An accurate air temperature measurement system based on an envelope pulsed ultrasonic time-of-flight technique", *Review of Scientific Instruments*, vol. 78, no. 115102, 2007.

- [122] N. Patwari, A. O. Hero, M. Perkins, N. S. Correal, and R. J. O'Dea, "Relative location estimation in wireless sensor networks", *Signal Processing*, *IEEE Transactions on*, vol. 51, no. 8, pp. 2137–2148, 2003.
- [123] C. M. Dissanayake, R. Kotagiri, M. N. Halgamuge, B. Moran, and P. Farrell, "Propagation constraints in elephant localization using an acoustic sensor network", in 2012 IEEE 6th International Conference on Information and Automation for Sustainability (ICIAfS), Conference Proceedings, pp. 101–105, 2012.
- [124] M. Garstang, D. R. Fitzjarrald, K. Fristrup, and C. Brain, "The daily cycle of lowfrequency elephant calls and near-surface atmospheric conditions", *Earth Interactions*, vol. 9, no. 1, 2005.
- [125] H. L. V. Trees, Detection, Estimation, and Modulation Theory: Radar-Sonar Signal Processing and Gaussian Signals in Noise. Krieger Publishing Co., Inc., 1992.
- [126] G. Shen, R. Zetik, and R. S. Thoma, "Performance comparison of TOA and TDOA based location estimation algorithms in LOS environment", in 5th Workshop on Positioning, Navigation and Communication, WPNC 2008. IEEE, Conference Proceedings, pp. 71–78, 2008.
- [127] "Hitachi multimediacard", Report ADE-203-1294C (Z), 2002.
- [128] G. Jayasinghe, F. Fahmy, N. Gajaweera, and D. Dias, "A GSM alarm device for disaster early warning", in *First International Conference on Industrial and Information Systems*. IEEE, Conference Proceedings, pp. 383–387, 2006.
- [129] M. Mayilvaganan and M. Devaki, "Elephant vocalization direction of arrival estimation and external factors affecting wave propagation using acoustic sensor network", *target*, vol. 6, p. 18, 2013.

University Library



A gateway to Melbourne's research publications

Minerva Access is the Institutional Repository of The University of Melbourne

Author/s: Dissanayake, Chinthaka M.

Title: Mitigating human-elephant conflict through an engineering approach

Date: 2018

Persistent Link: http://hdl.handle.net/11343/220263

File Description:

Thesis_ Chinthaka M Dissanayake_Mitigating Human-Elephant Conflict through an Engineering Approach

Terms and Conditions:

Terms and Conditions: Copyright in works deposited in Minerva Access is retained by the copyright owner. The work may not be altered without permission from the copyright owner. Readers may only download, print and save electronic copies of whole works for their own personal non-commercial use. Any use that exceeds these limits requires permission from the copyright owner. Attribution is essential when quoting or paraphrasing from these works.